## Model-based Control of Large-scale Baggage Handling Systems

1

1

Leveraging the Theory of Linear Positive Systems for Robust Scalable Control Design

\_ I

\_ I

## Model-based Control of Large-scale Baggage Handling Systems

Leveraging the Theory of Linear Positive Systems for Robust Scalable Control Design

### Proefschrift

ter verkrijging van de graad van doctor aan de Technische Universiteit Delft, op gezag van de Rector Magnificus prof. dr. ir. T. H. J. J. van der Hagen, voorzitter van het College voor Promoties, in het openbaar te verdedigen op vrijdag 24 juni 2022 om 15:00 uur

door

#### Yashar ZEINALY

Master of Science in Systems, Control, and Mechatronics, Chalmers Tekniska Högskola, Göteborg, Zweden, geboren te Urmia, Iran. Dit proefschrift is goedgekeurd door de

promotoren: prof. dr. ir. B. De Schutter, prof. dr. ir. H. Hellendoorn

Samenstelling promotiecommissie:

Rector Magnificus, Prof. dr. ir. B. De Schutter, Prof. dr. ir. H. Hellendoorn, voorzitter Technische Universiteit Delft Technische Universiteit Delft

*Onafhankelijke leden:* Prof. dr. A. Alessandri, Prof. dr. ir. R. de Koster, Prof. dr. R. R. Negenborn, Prof. dr. ir. J. W. van Wingerden,

Università di Genova Erasmus University Technische Universiteit Delft Technische Universiteit Delft

*Overige leden:* Prof. dr. ir. J. H. van Schuppen,

Technische Universiteit Delft





Delft Center for Systems and Control

This dissertation has been completed in partial fulfillment of the requirements of the Dutch Institute of Systems and Control (DISC) for graduate studies.

TRAIL Thesis Series, T2022/8, The Netherlands TRAIL Research School P.O. Box 5017 2600 GA Delft, The Netherlands T: +31 (0) 15 278 6046 E: info@rstrail.nl

Published and distributed by: Y. Zeinaly E-mail: yashar.ze@gmail.com

ISBN 978-90-5584-311-4

Keywords: Baggage Handling Systems, Model Predictive Control, Linear Positive Systems, Reachable Subsets, Polyhedral Cones

Copyright © 2022 by Y. Zeinaly

An electronic version of this dissertation is available at https://repository.tudelft.nl/.

To my dearest Vida

Ι\_\_

\_ I

\_ I

# Contents

Ι\_\_

| Pr | reface   | xi   |  |  |  |
|----|--|--|--|--|--|
| Ac | cknowledgements  | xiii   |  |  |  |
| 1  | Introduction         1.1 Model-Based Control of BHSs         1.2 Robust MPC for Linear Positive Systems         1.3 Reachability of Linear Positive Systems         1.4 Organization         References  |  |  |  |  |
| 2  | An Integrated MPC Approach for Baggage Handling Systems         2.1 Introduction.         2.2 Dynamical Model.         2.2.1 Notation and Assumptions         2.2.2 Model Description         2.3 MPC Problem Formulation         2.3.1 Objective Function         2.3.2 Linear Programming Approach         2.3.3 Iterative Linear Programming Approach         2.3.4 Nonlinear Programming Approach         2.4 Case Study         2.5 Conclusions and Future Work | <b>11</b><br>. 12<br>. 15<br>. 15<br>. 22<br>. 22<br>. 22<br>. 25<br>. 26<br>. 27<br>. 27<br>. 32  |  |  |  |
| 3  | References.         Towards Robust Multi-level Control of Baggage Handling Systems         3.1 Introduction.         3.2 Dynamical Model.         3.3 MPC Problem Formulation         3.4 Feedback Control Problem Formulation         3.4.1 Problem Setup         3.4.2 Hard State Constraints         3.4.3 Soft State Constraints         3.5 Integration of MPC and Feedback Controllers         3.6 Case Study         3.7 Conclusions and Future Work          | . 34<br>37<br>. 37<br>. 38<br>. 41<br>. 44<br>. 44<br>. 45<br>. 46<br>. 48<br>. 49<br>. 51<br>. 53 |  |  |  |
| Ap | <b>ppendices</b><br><b>3.A</b> Derivation of Linear Program  | <b>54</b><br>. 54  |  |  |  |

| 4               | ustly Positively Invariant Sets for Linear Positive Systems   | 56  |   |  |  |  |  |
|-----------------|---|---|---|--|--|--|--|
|                 | 4.1 Introduction  |   |   |  |  |  |  |
|                 | 4.2   | <b>4.2</b> Notation, terminology and definition   |   |  |  |  |  |
|                 | 4.3   | output $L_{\infty}$ -norm-optimial recuback control for Positive Systems  | 60  |  |  |  |  |
|                 |   | 4 3 1 Preliminaries   | 60  |  |  |  |  |
|                 |   | 4.3.2 Upper Bound on Lnorm of Output  | 60  |  |  |  |  |
|                 |   | 4.3.3 Derivation of the Feedback Gain.  | 61  |  |  |  |  |
|                 |   | 4.3.4 Derivation of the Feedback Gain With Soft Control   | 01  |  |  |  |  |
|                 |   | Constraints   | 62  |  |  |  |  |
|                 | 4.4   | Model Predictive Control  | 63  |  |  |  |  |
|                 |   | 4.4.1 Baseline MPC  | 64  |  |  |  |  |
|                 |   | 4.4.2 Tube-based MPC  | 64  |  |  |  |  |
|                 |   | 4.4.3 Recursive Feasibility and Asymptotic Stability  | 67  |  |  |  |  |
|                 | 4.5   | Model Description of BHS  | 70  |  |  |  |  |
|                 |   | 4.5.1 Evolution of Link Density   | 70  |  |  |  |  |
|                 |   | 4.5.2 Inputs, Outputs and Constraints   | 74  |  |  |  |  |
|                 | 4.6   | Case Study  | 74  |  |  |  |  |
|                 |   | 4.6.1 Configuring the MPC Controller  | 76  |  |  |  |  |
|                 | 4 7   | 4.6.2 Simulation Results  | 77  |  |  |  |  |
|                 | 4.1   |   | 79<br>01  |  |  |  |  |
|                 | Rele  | erences   | 01  |  |  |  |  |
|                 |   |   |   |  |  |  |  |
| Ap              | pen   | dices   | 83  |  |  |  |  |
| Ap              | <b>pen</b><br>4.A   | <b>dices</b><br>Proofs  | <b>83</b><br>83   |  |  |  |  |
| Ap<br>5         | 4.A   | dices<br>Proofs<br>chable Subsets of Linear Positive Systems from the Origin  | <b>83</b><br>83<br><b>86</b>  |  |  |  |  |
| <b>А</b> р<br>5 | <b>9 pen</b><br>4.A<br><b>Rea</b><br>5.1  | dices Proofs chable Subsets of Linear Positive Systems from the Origin Introduction   | <ul> <li>83</li> <li>83</li> <li>86</li> <li>87</li> <li>97</li> </ul>  |  |  |  |  |
| <b>А</b> р<br>5 | 4.A<br><b>Rea</b><br>5.1  | dices         Proofs.         chable Subsets of Linear Positive Systems from the Origin         Introduction.         5.1.1 Motivation and Scope         5.1.2 Dravisus Week  | <ul> <li>83</li> <li>83</li> <li>86</li> <li>87</li> <li>87</li> <li>87</li> <li>87</li> </ul>  |  |  |  |  |
| Ap              | 4.A<br><b>Rea</b><br>5.1  | dices         Proofs.         chable Subsets of Linear Positive Systems from the Origin         Introduction.         5.1.1 Motivation and Scope         5.1.2 Previous Work.         5.1.2 Contribution of This Work   | <ul> <li>83</li> <li>83</li> <li>86</li> <li>87</li> <li>87</li> <li>87</li> <li>87</li> <li>87</li> <li>80</li> </ul>  |  |  |  |  |
| Ap              | 4.A<br><b>Rea</b><br>5.1  | dices         Proofs.         chable Subsets of Linear Positive Systems from the Origin         Introduction.         5.1.1 Motivation and Scope         5.1.2 Previous Work.         5.1.3 Contribution of This Work         Preliminaries   | <ul> <li>83</li> <li>83</li> <li>86</li> <li>87</li> <li>87</li> <li>87</li> <li>88</li> <li>90</li> </ul>  |  |  |  |  |
| <b>A</b> p<br>5 | 4.A<br><b>Rea</b><br>5.1  | dices         Proofs.         chable Subsets of Linear Positive Systems from the Origin         Introduction.         5.1.1 Motivation and Scope         5.1.2 Previous Work.         5.1.3 Contribution of This Work         Preliminaries         5.2.1 Positive Real Numbers   | <ul> <li>83</li> <li>83</li> <li>86</li> <li>87</li> <li>87</li> <li>87</li> <li>87</li> <li>88</li> <li>90</li> <li>90</li> </ul>  |  |  |  |  |
| <b>A</b> p<br>5 | 4.A<br><b>Rea</b><br>5.1  | dices<br>Proofs   | <ul> <li>83</li> <li>83</li> <li>86</li> <li>87</li> <li>87</li> <li>87</li> <li>88</li> <li>90</li> <li>90</li> <li>92</li> </ul>  |  |  |  |  |
| <b>A</b> p<br>5 | 4.A<br>Rea<br>5.1   | dices<br>Proofs   | <ul> <li>83</li> <li>83</li> <li>86</li> <li>87</li> <li>87</li> <li>87</li> <li>88</li> <li>90</li> <li>90</li> <li>92</li> <li>92</li> <li>92</li> </ul>  |  |  |  |  |
| Ap<br>5         | <b>Rea</b> 5.1  | dices<br>Proofs   | <ul> <li>83</li> <li>83</li> <li>86</li> <li>87</li> <li>87</li> <li>87</li> <li>88</li> <li>90</li> <li>90</li> <li>92</li> <li>92</li> <li>92</li> </ul>  |  |  |  |  |
| <b>A</b> p      | 4.A<br>Rea<br>5.1   | dices<br>Proofs   | <ul> <li>83</li> <li>83</li> <li>86</li> <li>87</li> <li>87</li> <li>87</li> <li>88</li> <li>90</li> <li>90</li> <li>92</li> <li>92</li> <li>92</li> <li>94</li> </ul>  |  |  |  |  |
| Ap<br>5         | <ul> <li>4.A</li> <li>Rea</li> <li>5.1</li> <li>5.2</li> <li>5.3</li> </ul>   | dices<br>Proofs   | <b>83</b><br>83<br>86<br>87<br>87<br>87<br>87<br>88<br>90<br>90<br>92<br>92<br>92<br>92   |  |  |  |  |
| Ap              | <ul> <li>4.A</li> <li>Rea</li> <li>5.1</li> <li>5.2</li> <li>5.3</li> </ul>   | dices<br>Proofs   | <b>83</b><br>83<br><b>86</b><br>87<br>87<br>87<br>87<br>87<br>87<br>87<br>87<br>87<br>90<br>90<br>92<br>92<br>92<br>92<br>92  |  |  |  |  |
| Ap              | <ul> <li>4.A</li> <li>Rea</li> <li>5.1</li> <li>5.2</li> <li>5.3</li> </ul>   | dices<br>Proofs   | <ul> <li>83</li> <li>83</li> <li>86</li> <li>87</li> <li>87</li> <li>87</li> <li>87</li> <li>88</li> <li>90</li> <li>90</li> <li>92</li> <li>92</li> <li>92</li> <li>92</li> <li>94</li> <li>96</li> <li>96</li> <li>98</li> </ul>                          |  |  |  |  |
| Ap<br>5         | <ul> <li><b>Pend</b></li> <li>4.A</li> <li><b>Rea</b></li> <li>5.1</li> <li>5.2</li> <li>5.3</li> <li>5.4</li> </ul>              | dices<br>Proofs   | <ul> <li>83</li> <li>83</li> <li>86</li> <li>87</li> <li>87</li> <li>87</li> <li>87</li> <li>88</li> <li>90</li> <li>92</li> <li>92</li> <li>92</li> <li>92</li> <li>92</li> <li>94</li> <li>96</li> <li>98</li> <li>98</li> </ul>                          |  |  |  |  |
| Ap              | <ul> <li><b>pend</b></li> <li>4.A</li> <li><b>Rea</b></li> <li>5.1</li> <li>5.2</li> <li>5.3</li> <li>5.4</li> <li>5.5</li> </ul> | dices<br>Proofs   | <ul> <li>83</li> <li>83</li> <li>86</li> <li>87</li> <li>87</li> <li>87</li> <li>87</li> <li>88</li> <li>90</li> <li>90</li> <li>92</li> <li>92</li> <li>92</li> <li>92</li> <li>94</li> <li>96</li> <li>96</li> <li>98</li> <li>98</li> <li>104</li> </ul> |  |  |  |  |
| Ap              | <ul> <li>4.A</li> <li>Rea</li> <li>5.1</li> <li>5.2</li> <li>5.3</li> <li>5.4</li> <li>5.5</li> </ul>                             | dices<br>Proofs   | <b>83</b><br>83<br>86<br>87<br>87<br>87<br>87<br>87<br>87<br>87<br>87<br>87<br>90<br>90<br>92<br>92<br>92<br>92<br>92<br>94<br>96<br>96<br>98<br>98<br>104<br>108   |  |  |  |  |
| <b>A</b> p<br>5 | <ul> <li>4.A</li> <li>Rea</li> <li>5.1</li> <li>5.2</li> <li>5.3</li> <li>5.4</li> <li>5.5</li> <li>5.6</li> </ul>                | dices Proofs. chable Subsets of Linear Positive Systems from the Origin Introduction. 5.1.1 Motivation and Scope 5.1.2 Previous Work. 5.1.3 Contribution of This Work Preliminaries 5.2.1 Positive Real Numbers, Positive Matrices, and Cones 5.2.2 Linear Positive Systems 5.2.3 Terminology of Controllability and Reachability. 5.2.4 Existing Results on Reachability and Controllability of Linear Positive Systems 5.3.1 Concepts. 5.3.2 Problem Formulation When Is the Infinite-Time Reachable Set a Polyhedral Set?. 5.5.1 Special Case. Does the Reachable Set Contain a Pre-specified Set?                             | <b>83</b><br>83<br>86<br>87<br>87<br>87<br>87<br>87<br>87<br>87<br>87<br>87<br>87<br>90<br>90<br>92<br>92<br>92<br>92<br>92<br>92<br>94<br>96<br>96<br>96<br>98<br>98<br>104  |  |  |  |  |
| <b>A</b> p<br>5 | <ul> <li>4.A</li> <li>Rea</li> <li>5.1</li> <li>5.2</li> <li>5.3</li> <li>5.4</li> <li>5.5</li> <li>5.6</li> <li>5.7</li> </ul>   | dices Proofs. Chable Subsets of Linear Positive Systems from the Origin Introduction. 5.1.1 Motivation and Scope 5.1.2 Previous Work. 5.1.3 Contribution of This Work Preliminaries 5.2.1 Positive Real Numbers, Positive Matrices, and Cones 5.2.2 Linear Positive Systems 5.2.3 Terminology of Controllability and Reachability. 5.2.4 Existing Results on Reachability and Controllability of Linear Positive Systems 5.3.1 Concepts. 5.3.2 Problem Formulation When Is the Infinite-Time Reachable Set a Polyhedral Set? 5.5.1 Special Case. Does the Reachable Set Contain a Pre-specified Set? Conclusions and Future Work. | <b>83</b><br>83<br>86<br>87<br>87<br>87<br>87<br>87<br>87<br>87<br>87<br>87<br>87<br>90<br>90<br>92<br>92<br>92<br>92<br>92<br>92<br>92<br>92<br>94<br>96<br>96<br>98<br>98<br>104<br>109<br>112  |  |  |  |  |

\_ |

-I

| Appendices 1         |                     |         |   |            |  |  |  |
|----------------------|---------------------|---------|---|------------|--|--|--|
|                      | 5.A                 | Positiv | ve Matrices   | 117<br>122 |  |  |  |
| 6                    | Con                 | clusio  | ns and Further Research                             | 122        |  |  |  |
| Ū                    | 6.1                 | Concl   | uding Remarks.                                      | 124        |  |  |  |
|                      |                     | 6.1.1   | Robust Model-based Control of Baggage Handling      |            |  |  |  |
|                      |                     |         | Systems   | 125        |  |  |  |
|                      |                     | 6.1.2   | Reachability Problem for Discrete-time Linear       |            |  |  |  |
|                      |                     |         | Positive Systems                                    | 127        |  |  |  |
|                      | 6.2                 | Recon   | nmendations for Further Research                    | 127        |  |  |  |
|                      |                     | 6.2.1   | Extensions of Models and Methods Developed for      | 107        |  |  |  |
|                      |                     | 600     | BHSs  | 127        |  |  |  |
|                      |                     | 6.2.2   | Extensions of the Control Theory of Linear Positive | 100        |  |  |  |
|                      |                     | 623     | Other Application Areas of Control Theory of Linear | 129        |  |  |  |
|                      |                     | 0.2.5   | Positive Systems                                    | 129        |  |  |  |
|                      | Refe                | rences  |   | 132        |  |  |  |
| 0                    |                     |         |   | 100        |  |  |  |
| Su                   | Summary             |         |   |            |  |  |  |
| List of Publications |                     |         |   |            |  |  |  |
| Curriculum Vitæ      |                     |         |   |            |  |  |  |
| ТF                   | TRAIL Thesis Series |         |   |            |  |  |  |

۱\_\_\_

\_ I

\_ I

# Preface

This work was driven by the quest to develop model-based control algorithms for control of large-scale systems with the focus on large-scale baggage handling system. The main problem addressed in this thesis has overlaps with the problem of routing and scheduling of vehicles in automated warehousing, which is a vast field of research at the intersection of control theory & optimization of dynamical systems and operations research. Researchers from both fields have studied similar problems, albeit with a different scope and objective, for applications such as route planning of Automated Guided Vehicles (AGV).

What makes this work different is the especial attention paid to three key design features of the developed control solution, namely scalability, optimal performance, and robustness, which are necessary to apply the developed control algorithm in practice. In fact, these three features of the control algorithm guided me in making key decisions with respect to the choice of modeling framework and the control approach, and, hence, had significant impact on the composition of this thesis. For example, in Chapter 2, the choice for modeling the movement of vehicle as continuous flows rather than considering individual movement of vehicles was made to arrive at a tractable optimization problem that scales linearly with the size of problem. In addition, in the same chapter, model-predictive control was chosen as the main control approach as it enables one to incorporate the model in the control design with the aim of achieving a defined optimal performance. The robustness requirement led me to first write Chapter 3 to develop a model-based approach to deal with unmodeled disturbances in an optimal manner. My concerns with scalability of the solution developed in Chapter 3 subsequently led to the work of Chapter 4, where specific tools were developed for linear positive systems, of which baggage handling system in an example. Using the tools developed in Chapter 4, the original problem of Chapter 3 was solved in a scalable manner. In addition, the work of Chapter 4 incorporates the model-based design of Chapter 2, guaranteeing the optimal performance of the controlled system. Therefore, Chapter 4 marks the culmination of efforts made in its preceding chapters with the aim of achieving a robust scalable model-based control design for baggage handling systems.

Even though the subject of Chapter 5 was not initially among the main research questions driving my work, the work done in Chapter 4 on *robustly positively invariant sets* for discrete-time linear positive systems and the interesting properties of positive system intrigued me to study the theory of positive systems in more depth. In doing so, I gravitated towards the controllability problem for linear positive systems as it was surprisingly very different from the same problem for linear system. This led to the work of Chapter 5 on studying the geometry of reachable subsets from the origin for discrete-time linear positive systems.

This thesis may be of interest to those working on advanced model-based control

design approaches for large-scale logistic system, and to those who see scalability as an inseparable part of their approach. In addition, enthusiasts of the theory of positive system and those curious about how certain properties of linear positive systems render certain control problems of linear system much simpler may find reading this work useful.

> Yashar Zeinaly Utrecht, June 2022

# Acknowledgements

I am finally here, writing the last piece of my PhD dissertation, the piece I was quite frankly looking forward to all along. On my first day of PhD, I had to call the police to get me out of the 3ME building because I had confidently answered "yes" to my officemate, Dang Doan, when he asked me if I was sure I had access to the 3ME building after the business hours. Little did I know that there was a big difference between obtaining my access card and having the proper access rights installed on it. This marked the beginning of the excitement, drama, struggle, and joy that ensued during the next 4 years. Now, looking back, I would like to briefly reflect upon my time in DCSC, and thank the ones who supported, encouraged, and guided me throughout this journey.

First and foremost, I would like to express my deepest gratitude to my promoter Prof. Bart De Schutter for his endless support and his supervision. This thesis would not have been possible without his constant support, his valuable input on my research, and his insightful comments. I am especially thankful to him for showing extreme patience with me, for challenging my ideas and playing the devil's advocate, and for nudging me in the right direction whenever things didn't seem to work well. Moreover, I highly appreciate him making himself truly available for his students in the times of need.

I would also like to sincerely thank Prof. Jan van Schuppen for his support, encouragement, dedication, and his immense knowledge. It was truly an honor and a great pleasure working with him. I have very much enjoyed our discussions on the topic of positive systems, and I have learned a lot from his profound knowledge on this subject. I highly appreciate Jan's major contribution to Chapter 5, without which this thesis would not have been possible. Collaboration with Jan will definitely be missed.

Special thanks go to my co-promoter Hans Hellendoorn for encouraging me to see the "bigger picture", for his positive attitude, and for always lending an ear when needed. I also appreciate his useful comments on the thesis and the propositions.

I highly appreciate all committee members for their time and suggestions. Particularly, I would like to thank Prof. René de Koster and Prof. Rudy Negenborn for their comments on Chapter 1 and Chapter 2.

I have greatly enjoyed DISC lectures of Prof. Harry Trentelman of the University of Groningen and Prof. Maarten Steinbuch of Eindhoven University of Technology on systems theory, LMIs in control, and mechatronics. Both gentlemen were excellent enthusiastic teachers.

I very much enjoyed working together with Amir and Anna Sadowska as the teacher assistances of the *Control Theory* course. Thanks for your constructive collaboration. In addition, my appreciation goes to the students of the *Control Theory* course, who kept me very motivated by asking questions and triggering

interesting discussions. Without any doubt, my experience as the teacher assistant for this course was one of the best memories, if not the best memory, of my time as a PhD candidate in DCSC.

Huge thanks go to my dear officemate in my first two years: Amol, Dang, and Alfredo, who assisted me with practical matters in my first week, Ismini, Nicolas, Alfredo, Bart Kersbergen for making a fun work place, for intriguing discussions on society, politics and economy, and for DCSC game nights. I would also like to thank my DCSC colleagues Marco, Renshi, Le, Majid, Dieky, Elisabeth, Edwin, Farid, Hans, Iliya, and Ruxandra for intense foosball matches and for creating nice work atmosphere. I want to express my gratitude to Sadegh, Farzaneh, and Pirooz for their friendship and hospitality. My dear friends Amir, Sasan, Patricio and Mojtaba, thanks for your friendship. I consider myself lucky to have known you guys. I would like to express my sincere gratitude to my officemate-turned-great-friend Max Potters. Max, you have been a wise fun colleague, a patient listener, and a highly trusted friend, with whom I could share personal matters. I would like to thank you for keeping me motivated and for being a fantastic officemate.

I would also like to acknowledge efforts of the ladies in DCSC secretariat, Ellen van den Berg-Moor, Esther de Booij, Kitty Dukker, Marieke Versloot-Bus, Heleen Sakkee-Zaal, and Francy Ahaus-Smeets, for tirelessly supporting me with administrative tasks. Also thanks to (then) DISC secretary Saskia van der Meer for arranging practical matters with respect to the DISC program. I would like to extend my thanks to Conchita van der Stelt of TRAIL Research School for her efforts in facilitating the publication of this dissertation.

My wholehearted gratitude goes to my father, who was always there for me when I needed support and who encouraged me to persist in continuing the journey I had started. Even though he didn't live to see the completion of my PhD, his battle with cancer against all odds truly inspired me to finish this thesis. Thanks for everything dad.

There is no way I can thank you enough my dearest Vida. Navigating my way through the PhD years would have been impossible if it had not been for your unconditional love, endless support, and tremendous sacrifices. You have always been by my side even when it seemed the *world* was against me. You encouraged me the most, and helped me find the light at the end of the tunnel. Thanks for taking the journey of life with me and for giving us the most precious gift of all, our little *Nolan*.

Yashar Zeinaly Utrecht, June 2022

# 1

# Introduction

"I have had my results for a long time; but I do not yet know how I am to arrive at them."

Carl Friedrich Gauss, 1777 – 1855

This thesis is presented as a collection of papers, either published, accepted for publication, or under review. The papers discussed in this thesis cover two main categories of topics: i) model-based control design of large-scale Baggage Handling Systems (BHSs), and ii) positive linear systems and their applications in model-based control of large-scale baggage handling systems. The papers in the first part discuss modeling of BHSs and several Model Predictive Control (MPC) approaches for BHSs along with their computational complexity. The publications in the second part collectively rely on tools and methods from the theory of linear positive systems to develop a robust MPC approach for linear positive systems with applications to BHSs, and take a deeper dive into the theory of linear positive systems by providing a characterization for the infinite-time and the finite-time reachable subsets of discrete-time linear positive systems.

## 1.1. Model-Based Control of BHSs

The interest in large-scale BHSs has recently increased due to the increase in demand for air travel and cargo shipment, airport competition to attract passengers and airlines, and labor cost savings. Despite the recent downturn in the air travel demand due to the COVID-19 pandemic travel restrictions, the International Air Transportation Association predicts an average annual passenger growth rate of 3.3% [1], based on which 2.3 and 3.4 billion passengers worldwide are estimated to fly respectively in 2021 and 2022 with the recovery to the 2019 level taking place in 2024. In addition, robust growth of cargo demand is expected for 2021 and

2022, respectively, at 7.9% and 13.2% above the 2019 level [2]. The unexpected increase in cargo demand is most likely due to congested supply chains, which have caused a temporary shift of freight from sea to air. With the increasing demand, baggage handling service quality becomes a key factor for passenger and airline satisfaction, which, together with a reduction of expensive manpower, are crucial for long-term success and sustained business of airports.

Addressing the challenges mentioned above necessitates design and development of highly automated BHSs of a larger scales that are able to cope with the increased demand. More sophisticated designs, however, pose new challenges with respect to operational cost and efficiency. With modern baggage handling systems of large scale, it is of paramount importance to develop a systematic scheme to operate the system as efficiently as possible for two main reasons: i) to reduce both energy and operational costs to maintain affordability, ii) to fully exploit the capability of the sophisticated design. In order to develop design methods that are applicable to real-world BHSs, we focus on computational complexity and scalability for large-scale BHSs. Hence, the aim is not to develop the "best" performing MPC scheme, but rather the one that is also applicable to a large-scale BHS.

Chapter 2 of this thesis proposes a modeling framework for BHSs, in which the BHS is considered as a network of origin, transition, and destination nodes interconnected by links. The movement of pieces of baggage along the network links is modeled by baggage accumulation on the links driven by link-to-link baggage flows. The model developed in this chapter also takes into account the link-to-link flow travel time, which is generally a function of baggage accumulation on specific links.

Based on how the link-to-link travel time is treated, three classes of MPC [3, 4] approaches are proposed, namely a Nonlinear Programming (NLP) based approach, an Iterative Linear Programming (ILP) based approach, and a Linear Programming (LP) based approach. Given full information on the current and future baggage demand, the MPC approaches decide on modulating link-to-link baggage flows such that an optimal balance between timely delivery of pieces of baggage to the destination nodes and the overall energy consumption is achieved while guaranteeing the operational constraints. The recursive feasibility of the optimization problem and the asymptotic stability of the closed-loop system are either validated empirically via simulation or assured by employing sufficiently long prediction horizons.

The performance and computational complexity of the three MPC approaches are compared among each other and against a heuristic method commonly used for control of BHSs. It is shown that the MPC-based approaches can outperform the heuristic state-of-the art method, with further performance enhancement achieved for a longer prediction horizon. Furthermore, while the NLP approach provides the most accurate predictions, the optimization problem quickly becomes intractable for increasing values of the prediction horizon or for larger BHSs. The LP approach is the most efficient one, at the cost of low accuracy of predictions, especially for large prediction horizons. The ILP approach offers a good balance between accuracy of predictions and complexity of the optimization problem for long prediction horizons. Since it benefits from accurate predictions over a long horizon, it outperforms the heuristic method the most while being the most scalable to more complex BHS networks.

It is worth mentioning that there is a rich body of literature on Autonomous Mobile Robot (AMR) and Automated Guided Vehicle (AGV) based material handling and sorting systems [5–11]. A good overview on the application of AMRs in intralogistics is provided in [10]. For the AGV-based systems, [5, 6] provide a good review of the literature on the design and control of AGV systems for manufacturing, warehousing, and transport of material. While there is overlap between the problem investigated in Chapter 2 of this thesis and the three main category of problems in AMR- or AGV-based systems, namely route planning, scheduling of tasks, and dispatching, there are notable differences that does not allow direct application of those methods to the control of DCV-based BHSs.

These difference pertain to the objectives, complexity of solutions, and algorithms. The DCV-based BHSs control problem we discuss in Chapter 2, combines "light" versions of the three problem categories for AMR-based material handling systems. For example, in the approach of Chapter 2, we do not plan a route for each individual DCV, but we actually plan a time-based distribution of DCV flows. This is due to the fact that the main control objectives in Chapter 2, unlike AMRor AGV-based systems, are not minimizing throughput time, finding the shortest path, minimizing travel time, or minimizing travel distance, but rather constrained time-based distribution of DCV flows in the network with minimum energy consumption. In addition, while the scheduling problem requires complex solutions for AMR-based manufacturing systems with multiple jobs and machine centers, the DCV-based BHS system only involves loading and unloading of DCVs, which makes the scheduling problem much simpler. The multi-objective MPC-based control algorithm developed in Chapter 2 can be implemented efficiently in a centralized manner for large-scale BHSs with several hundred DCVs. In addition, it provides a standard way of incorporating time-varying (dynamic) baggage demand profiles. In contrast, centralized routing and scheduling algorithms for AMR-based systems are practically feasible only for systems with a small number of vehicles. For large-scale systems, decentralized control structures are preferred, which come at the cost of sub-optimal control performance.

In Chapter 3, we focus on the predictability of the baggage demand profile and we consider the situations in which the baggage demand is only partially known in contrast to Chapter 2, where full knowledge of the current and future baggage demand was assumed. In this framework, the unpredictable baggage demand is considered as an additive disturbance to the system. We propose a two-level control approach consisting of i) a top-level controller, designed for the nominal system<sup>1</sup>, that generates an optimal nominal control input based on full information of the baggage demand, and ii) a bottom-level controller, designed for the error system<sup>2</sup>, that stabilizes the error system. Hence, the actual trajectory converges to the nominal trajectory driven by the top level controller. In addition to closed-loop stability of the error system, the bottom-level controller can optionally be designed

<sup>&</sup>lt;sup>1</sup>The nominal system is the system without disturbance.

<sup>&</sup>lt;sup>2</sup>The error considered here is the difference between actual trajectory and the nominal trajectory.

to achieve a defined input-output optimal performance.

Assuming that the baggage demand is composed of a nominal part and a small additive disturbance part, the modeling framework and the MPC approaches developed in Chapter 2 are used to develop an ILP-based MPC controller at the top level, which generates nominal link-to-link flows for the nominal component of the baggage demand. A state feedback controller is designed for the error system to minimize the  $L_2$  gain of the error system from the disturbance input to the error output, hence minimizing the impact of disturbances on the nominal trajectory in the L<sub>2</sub>-norm sense and providing robustness against perturbations caused by deviations of the demand from its nominal value. Using the bounded real lemma [12-15], the problem of searching for the feedback gain K minimizing the induced  $L_2$  gain of the closed-loop error system can be expressed in terms of linear matrix inequalities (LMIs) [16], which results in a convex optimization problem. The final control input applied to the system is then  $u_{mpc} + Ke$ , where  $u_{mpc}$  and e are, respectively, the output of the MPC controller and the error. An important aspect of this approach is that the state and control constraints of the overall system, derived from operational constraints of the BHS, are ultimately handled by the MPC controller. For this purpose, assuming a disturbance input with a known bound on the  $L_2$  norm, a positively invariant ellipsoid over the state space of error system is calculated offline. This state ellipsoid and the corresponding ellipsoid over the control input space can then be used to "tighten" the state and control constraints of the nominal MPC design such that the combined output of the controllers and the resulting system trajectory does not violate the control and state constraints.

In cases where the positively invariant ellipsoid associated with the unconstrained  $L_2$ -gain-optimal closed-loop error system turns out to be too "large" for the top level MPC controller to handle, a constrained approach is developed, where the  $L_2$  gain of the error system in minimized subject to ellipsoidal state and control hard constraints. This approach is further extended by formulating the desired state and control constrains as soft constraints, hence avoiding infeasibility of the LMIs due to the overly restrictive state or control constraints. A small case study illustrates the implementation and performance of the suggested two-level control scheme.

### **1.2. Robust MPC for Linear Positive Systems**

Chapter 4 discusses output- $L_{\infty}$ -norm-optimal feedback control design for linear positive systems, which is meant to be incorporated as the bottom-level controller in a tube-based MPC approach. For linear systems subject to additive disturbances, tube-based MPC [4, Chapter 3] is a low complexity robust MPC approach. It consists of a nominal MPC controller generating a nominal control input for the nominal (i.e., disturbance-free) system and a stabilizing state feedback controller governing the error system, which pertains to the difference between the uncertain system and the nominal system. Assuming a stable error system, the state trajectory of the disturbed system is contained in a bounded neighborhood of the nominal trajectory, which is called a tube. The combined goal of the controllers is to drive the tube center in a defined optimal manner while guaranteeing satisfaction of the constraints for the entire tube. This is achieved by ensuring that the nominal state and input trajectories satisfy appropriately "tightened" versions of the original constraints. The process of constraint tightening normally involves computation of the minimal robust invariant set, which is difficult to compute, especially for large-scale systems.

For stable linear positive systems, a robustly positively invariant hypercube containing the minimal robust invariant set can be obtained via a linear program, which can be used for constraint tightening in the tube-based MPC scheme. Based on this and given a disturbance set characterized by infinity-norm constraints (i.e., box constraints), we propose a state feedback design method for the error system that renders the closed-loop error system positive while minimizing the disturbance driven output of the error system in the  $L_{\infty}$  sense. For linear systems subject to additive disturbances, the  $L_{\infty}$  norm of output is a more general measure to capture the effect the of disturbances than the  $L_{\infty}$ -induced norm (i.e.,  $L_{\infty}$  gain) of the system as the  $L_{\infty}$  norm of the output provides a tighter bound on the "worst-casescenario performance" of the system. We show that the joint problem of searching for the feedback gain K and the smallest robustly positively invariant hypercube Xcontaining the minimal robust invariant set can be expressed as a linear program. Our problem formulation also allows for specifying a minimum size for X and for incorporation of infinity-norm hard and soft constraints on the control effort of the state feedback controller. Hence, in contrast to the approach of Chapter 3, where symmetric bounds on the state and the control input can only be imposed via ellipsoidal approximations, the approach developed in this chapter allows for direct inclusion of capacity constraints that arise in systems such as BHSs. In addition to providing a natural way of expressing the capacity constraints in BHSs, this method is better suited for large-scale BHSs since the conservatism introduced by the ellipsoidal approximations of the capacity constraints in the LMI-based approach may lead to infeasibility of the LMIs even though the original capacity constraints are feasible.

The developed method of state feedback design is applied, within a tube-based MPC scheme, to a BHS case study, where the nominal control input is generated by a linear-programming-based MPC controller. In contrast to Chapter 2, the design of the nominal MPC controller in this chapter explicitly enforces recursive feasibility and asymptotic stability by including an appropriate terminal constraint set and a terminal cost function, which are determined using linear programs. Therefore, the proposed tube-based control scheme is entirely a linear-programming-based approach, which scales well for large-scale BHSs.

#### **1.3. Reachability of Linear Positive Systems**

The class of positive dynamical systems, also known as nonnegative dynamical systems, generally refers to dynamical systems, the state trajectories of which reside in the positive orthant for nonnegative initial conditions and nonnegative inputs. This property, however, does not require the initial states or the control inputs to be nonnegative even though many real-word examples of positive systems are only defined for nonnegative inputs and nonnegative initial states. Positive systems arise

in many applications such as econometrics, bio-chemical reactors, compartmental systems, and transportation systems. BHSs, as the main application area of this thesis, are prime examples. The variables in positive systems represent growth rates, concentration levels, mass accumulation, flows, etc. Obviously, variables of this nature can only assume nonnegative values. Many classical control problems such as  $H_{\infty}$ ,  $L_1$ , and  $L_{\infty}$  problems, which are the subject of Chapter 4, can be made significantly simpler for positive systems. Inspired by the theory of positive dynamical systems and its applicability to several application domains, this chapter focuses on a fundamental property of linear positive systems: *reachability*. Following prior work in the literature on the reachability of linear positive systems are defined over a cone rather than over a linear subspace, their reachability is not in general equivalent to the reachability of linear systems.

We propose an alternative formulation of the reachability problem for discretetime linear time-invariant positive systems, where the goal is to check whether a certain subset or subcone of the positive orthant can be reached from the origin in finite or infinite time, which is motivated from an application point-of-view and which is in contrast with the classical view of requiring the entire positive orthant to be reachable from the origin. The latter view of reachability leads to strong conditions, which are very difficult to satisfy in practice. To formulate our view of the reachability problem, first a characterization of the finite-time and infinite-time controllable subsets, which are the cones of all reachable states from the origin in finite time and infinite time, is given for a single-input positive system. We focus our attention to the geometry of the controllable subsets and to the cases where the finite or infinite controllable subsets are polyhedral (i.e., they can be expressed by a finite number of extremal rays). Then, necessary and sufficient conditions for polyhedrality of the controllable subsets are derived for a pair (A, b), with A and b being the system matrices. For a generic **b** without any imposed structure, it turns out that such conditions are characterized solely in terms of the spectrum of A.

Finally, the reachability of a polyhedral target sub-cone or polyhedral target subset of the positive orthant is established by requiring all extremal rays of the target sub-cone or extremal points of the polyhedral subset to be included in the reachable finite-time or infinite-time controllable subsets, which leads to solving a set of constrained linear equations.

#### 1.4. Organization

The organization of thesis is conceptually illustrated by Fig. 1.1, where the publication corresponding to each chapter is highlighted, and where relation of the thesis chapters to each other, in terms of common themes, is indicated.

Chapter 2 describes the model of BHSs, where various MPC-based approaches are discussed and their performance is evaluated using a detailed case study. For a BHS subject to additive disturbances, the proposed 2-level control scheme is introduced in Chapter 3, and the design of a constrained state feedback controller minimizing the  $L_2$ -induced-gain of the closed-loop system is discussed. Chapter 4

discusses the output- $L_{\infty}$ -norm-optimal state feedback design problem for discretetime linear positive systems and implements the developed design approach in a tube-based MPC scheme for a BHS. In Chapter 5, we provide our view on the reachability of discrete-time linear time-invariant positive systems, where necessary and sufficient conditions for checking the reachability of a subset of the positive orthant are developed. Finally, Chapter 6, concluding remarks are laid out and potential future research directions in continuation of this work are pointed out.

#### References

- [1] IATA, 20-year air passenger forecast, https://www.iata.org/en/ publications/store/20-year-passenger-forecast, (2021).
- [2] IATA, Losses reduce but challenges continue, https://www.iata.org/ en/pressroom/2021-releases/2021-10-04-01, (2021), IATA Press Release No. 64.
- [3] C. E. Garcia, D. M. Prett, and M. Morari, *Model predictive control: Theory and practice–A survey*, Automatica 25, 335 (1989).
- [4] J. B. Rawlings, D. Q. Mayne, and M. M. Diehl, *Model Predictive Control: Theory, Computation, and Design*, 2nd ed. (Nob Hill Publishing, Madison, Wisconsin, 2018) p. 623.
- [5] T. Le Anh and R. de Koster, A review of design and control of automated guided vehicle systems, European Journal of Operational Research 171, 1 (2006).
- [6] I. F. A. Vis, Survey of research in the design and control of automated guided vehicle systems, European Journal of Operational Research 170, 677 (2006).
- [7] T. Le Anh, *Intelligent Control of Vehicle-Based Internal Transport Systems*, Ph.D. thesis, Department of Management of Technology and Innovation (2005), ERIM PhD Series Research in Management.
- [8] B. Zou, R. de Koster, Y. Gong, X. Xu, and G. Shen, *Robotic sorting systems: Performance estimation and operating policies analysis*, Transportation Science 55, 1430 (2021).
- [9] P. Chevalier, Y. Pochet, and L. Talbot, *Design and performance analysis of a heavily loaded material handling system*, CORE Discussion Papers **37** (2001).
- [10] G. Fragapane, R. de Koster, F. Sgarbossa, and J. O. Strandhagen, *Planning and control of autonomous mobile robots for intralogistics: Literature review and research agenda*, European Journal of Operational Research **294**, 405 (2021).
- [11] T. Le Anh and R. de Koster, *Multi-attribute dispatching rules for agv systems with many vehicles,* ERIM REPORT SERIES RESEARCH IN MANAGEMENT (2004).

- [12] V. A. Yakubovich, The solution of certain matrix inequalities in automatic control theory, Dokl. Akad. Nauk SSSR. 143, 11304 (1962).
- [13] R. E. Kalman, Lyapunov functions for the problem of Lur'e in automatic control, Proceedings of the National Academy of Sciences of the United States of America 49, 201 (1963).
- [14] A. Rantzer, On the Kalman—Yakubovich—Popov lemma, Systems & Control Letters 28, 7 (1996).
- [15] S. V. Gusev and A. L. Likhtarnikov, Kalman-popov-yakubovich lemma and the S-procedure: A historical essay, Automation and Remote Control 67, 1768 (2006).
- [16] S. Boyd, L. El Ghaoui, E. Feron, and V. Balakrishnan, *Linear Matrix Inequalities in System and Control Theory* (Society for Industrial and Applied Mathematics, 1994) p. 185.
- [17] Y. Zeinaly, B. De Schutter, and H. Hellendoorn, An integrated model predictive scheme for baggage-handling systems: Routing, line balancing, and emptycart management, IEEE Transactions on Control Systems Technology 23, 1536 (2015).
- [18] Y. Zeinaly, J. H. van Schuppen, and B. De Schutter, *Linear positive systems may have a reachable subset from the origin that is either polyhedral or non-polyhedral*, SIAM Journal on Matrix Analysis and Applications **41**, 279 (2020).
- [19] Y. Zeinaly, B. De Schutter, and H. Hellendoorn, *Towards a robust multi-level* control approach for baggage handling systems, in 52nd IEEE Conference on Decision and Control (2013) pp. 281–287.
- [20] Y. Zeinaly and B. De Schutter, Robustly positively invariant sets for discretetime linear positive systems: Application to a tube-based mpc approach, (2022), to be submitted to IEEE Transactions on Control Systems Technology.



Figure 1.1: Schematic diagram of the thesis organization highlighting the publication used for each chapter along with the common themes between chapters. Solid links indicate that the chapters are strongly related with respect to the mentioned themes. Dashed links indicate that the chapters are weakly related with respect to the mentioned themes.

9

\_ I

\_ I

# 2

# An Integrated Model Predictive Scheme for Baggage Handling Systems: Routing, Line Balancing, and Empty-Cart Management

"True optimization is the revolutionary contribution of modern research to decision processes."

George Bernard Dantzig, 1914 – 2005

This chapter proposes a new strategy for integrated control of destination-codedvehicle-based baggage handling systems. Three main control issues in baggage handling systems, namely, routing and scheduling, empty-cart management, and line balancing, are identified and a combined control approach based on model predictive control is proposed to tackle these issues in an optimal way. It is shown that the control approach can be formulated as a linear programming problem, which can be solved very efficiently, and hence the proposed approach can be extended to large-scale baggage handling systems. We illustrate the applicability and performance of the proposed approach by a case study, and we compare the results with the state-of-the-art method currently used for baggage handling systems.

This chapter is based on [1].

#### **2.1. Introduction**

 $\mathbf{I}$  n the past decade, modern baggage handling systems [2, 3] (BHS) have been implemented in large airports to accommodate the rising demand in air travel. Such baggage handling systems are controlled by state-of-the-art techniques that are mostly tailor-made for a specific layout. However, with increasing demand, it becomes necessary to increase the efficiency and reliability of the baggage handling systems by utilizing a systematic controller design approach. Such a control approach should optimize the performance of the baggage handling system in terms of reliability and costs. A modern baggage handling system, an essential layout of which is depicted in Fig. 2.1, is composed of the following components [2]: Destination coded vehicles (DCVs), which are high-speed vehicles powered by linear induction motors transporting the pieces of baggage between various locations in the system. Each DCV can carry only one piece of baggage. The term DCV refers to both carts move powered by linear induction motors and passive tubs on modular conveyor elements, as illustrated in Fig. 2.2; loading stations, where the pieces of baggage are loaded onto DCVs after entering the system (either from the check-in desks or from the transfer flights); unloading stations, where the DCVs unload the pieces of baggage; the pieces of baggage are then transported to the planes; an early baggage storage (EBS), which is an automated storage/retrieval system used to temporarily store the loaded DCVs; a network of uni-directional tracks on which the DCVs travel. This network connects the loading stations, the unloading stations, and the EBS; and a switch controller at each junction that determines the path of the DCVs that pass through that junction. From a high-level control perspective, there are three main control challenges related to the baggage handling systems [3–5], namely, i) routing and scheduling of DCVs, ii) line balancing, and iii) emptycart management. The routing problem is the problem of routing loaded DCVs from the loading stations to the unloading stations or to the EBS and the problem of routing the DCVs from the EBS to the unloading stations. Line balancing is the problem of dynamically assigning empty DCVs located at the unloading stations to the loading stations. Closely related to line balancing is empty-cart management, which is the problem of routing empty DCVs from the unloading stations, through the network, to their assigned loading stations.

The control problems in baggage handling system can, in general, be related to operation scheduling, flow shop scheduling, and production scheduling [6, 7], or to predictive routing and flow control [8, 9]. Recently, the application of model predictive control (MPC) [10–12] to supply chain management has been studied in the literature [13–15]. In [8] the authors propose a two-level decision making process: long-term strategic level based on offline branch and bound optimization and short-term tactical level based on MPC. In [14], the authors propose a MPC scheme based on quadratic programming (QP) optimization, and [13] proposes and MPC approach based on mixed integer linear programming (MILP). Our proposed approach here differs from the literature in the following ways: i) Unlike the supply chain networks where transport times are fixed or known functions of time, BHS involves highly variable queue-length-dependent transport delays, which needs special treatment when developing the model and defining the optimization



Figure 2.1: Basic configuration of a baggage handling system showing the loading stations, unloading stations, DCV storage (EBS), and the DCV depot. Vertical baggage and loaded-DCV stacks are assumed at the loading stations and the EBS, respectively. Vertical empty DCV stacks are assumed at the loading station and the central depot. In the rest of the network, loaded and empty DCVs accumulate along the network links.

problem. ii) To achieve timely arrival of DCVs within their time windows as well as possible, we introduce time-window constrains as soft, rather than hard, constraints in the objective function to ensure that the optimization problem remains feasible at all times. To this end, the objective function includes a time-weighted sum of DCV queues and DCV flows. In this way, we employ a different objective function from those in the literature, where travel times are often explicitly penalized and the time window requirements are imposed as optimization constraints. iii) For realtime control purposes, the proposed algorithm has to be computationally efficient for large-scale BHS. This makes MPC schemes with long prediction horizons based on MILP or nonlinear optimization practically unsuitable for large-scale BHS. Therefore, the solutions developed in the literature are not directly applicable to the baggage handling systems. To address this issue, we need to develop numerically-less-intensive schemes. In this chapter we have opted for an LP-based approach.

The state-of-the-art control method for baggage handling systems addresses the

routing problem by using look-up tables to control the switches at the junctions [2, 5]. These look-up tables are computed off-line for different system operation scenarios. However, such a control scheme cannot guarantee optimal performance of the system for complex network layouts. In addition, this control method addresses the empty-cart management by decoupling the empty-DCV and loaded-DCV

traffic flows using dedicated loops to transfer empty carts to loading stations. Moreover, it relies on heuristics for line balancing. An immediate shortcoming of such a control system is that baggage handling systems have to adopt simple design layouts (e.g., composed of few loops). This limits the achievable performance of the system in terms of flexibility and baggage throughput.

Recently, it has been shown [4, 5, 16, 17] that a systematically designed control system can cope with complex network layouts. In [16], a multi-agent control strategy has been proposed for conveyor-based baggage handling systems, where each bag is dynamically assigned a path using the shortest path algorithm. The authors of [16] have shown that their algorithm outperforms the conventional control scheme used in practice, but they have not included optimal performance in their problem formulation. Model-based control of DCV-based baggage handling systems is considered in [4], where an automated way of learning routing rules has been proposed to solve the routing problem. In comparison with [4], we discuss optimality of our approach as well as on-time arrival of DCVs to the unloading stations. In [5], a solution has been proposed for the routing problem based on model predictive control (MPC) [10-12]. In this approach, a dynamic sequence of optimal switch positions is assigned to each DCV in order to guide it along an optimal route to its destination. This approach guarantees optimal performance of the system, but it is computationally prohibitive for large-scale systems<sup>1</sup>. A more computationally efficient MPC-based approach has been proposed in [18], which arrives at a mixed integer linear programming (MILP) formulation of the problem. In [19], it has been shown that the problem in [18] can in fact be recast as a linear programming (LP) problem, which can be solved efficiently for large-scale systems [20]. Nevertheless, to the best of our knowledge, the previous works have only focused on a particular control issue of the baggage handling systems. The aim of this chapter is to propose an optimal integrated solution to routing and scheduling, empty-cart management, and line balancing.

We consider the following two criteria for an effective baggage handling system. First, the pieces of baggage should reach the assigned unloading stations within pre-specified time windows. Second, the cost of operating the system should be minimized. In order to achieve an overall optimal performance with respect to these criteria, in this chapter, we propose a control scheme based on MPC that addresses the aforementioned control problems in *one integrated approach*, rather than treating them as individual sub-problems. Moreover, based on the model we develop, we show that the resulting optimization problem is in general a nonlinear optimization problem the solution of which can be obtained using a general nonlinear programming (NLP) approach or a iterative linear programming (ILP) approach. We also propose a suboptimal solution based on linear programming. We then compare the performance of these three approaches to each other and to the state-of-the-art method used for BHS. In addition, we compare our proposed control approaches in terms of their computational complexity for the case study at hand.

<sup>&</sup>lt;sup>1</sup>For instance, Amsterdam Schiphol airport operates 550 DCVs [2], and Denver International Airport operates about 4000 DCVs [3], which indicates real-life systems can indeed be large scale.

The rest of this chapter is organized as follows. In Section 2.2, we develop the dynamical model. In Section 2.3, we define the MPC optimization problem. A case study is presented in Section 2.4 to illustrate the performance of the proposed control approach for a given scenario and finally, Section 2.5 concludes the chapter.





(a) Carts powered by linear induction motors. (b) Passive tubs on modular conveyor elements.

Figure 2.2: Two types of DCVs used for transporting luggage within BHSs. Note that for both motorized carts and passive tubs power by modular conveyor belts one can assign a travel route within the network. Photos courtesy of Vanderlande.

#### **2.2. Dynamical Model**

#### **2.2.1.** Notation and Assumptions

The baggage handling system network can be seen as a directed graph, where nodes of the graph are composed of loading stations, unloading stations, junctions, and the EBS, and where the links represent the tracks of the system. The relation between the graph representation and the real network is rather symbolic. Not all components of DCV-based BHS are shown on the graph. However, the most important components of BHS, namely, the loading stations, the junctions, the unloading stations, and the EBS are present in our graph representation of the network. In





Figure 2.3: Physical node v, which represents a loading station, an unloading station, or the EBS (left) and its "extended" description (right)

Figure 2.4: Flow variables associated with an extended loading station connected to external links 1 and 2.

our mathematical description of the system, as depicted in Fig. 2.3, we will replace physical loading stations, unloading stations, and the EBS with their "extended" description, which is a super node comprising two nodes, a unique virtual incoming link, and a unique virtual outgoing link. Please note that there is no virtual link associated with the junctions of the network, so the links connected to the junctions do represent the tracks. There are no DCV queues on the virtual links and the travel time on the virtual links represents the time needed for storing DCVs in their corresponding stacks or the time needed for loading pieces of baggage onto DCVs. Hence, each loading station and each unloading station, and the EBS has only one incoming link and one outgoing link. This considerably simplifies representation of the model. Hereafter, unless otherwise mentioned, we drop explicit reference to the "extended" prefix and use the term graph to refer to its extended version. The graph representation of the network is denoted as G = (V, A), where  $V = V_1 \cup V_2 \cup V_3 \cup \{v^*\}$  is the set of nodes composed of set  $V_1$  associated with the loading stations, set  $V_2$  associated with the intermediate nodes (i.e., junctions), set  $V_3$  associated with the unloading stations, and the node  $v^*$  associated with the EBS. Moreover, A is the set of arcs composed of links, (i.e., physical tracks as well as virtual links) connecting the elements of V. In the sequel, we make the following assumptions regarding configuration of the network:

- A1 Only loaded DCVs are dispatched from the loading stations.
- A2 The baggage queues at the unloading stations are ignored. This is because we assume either destination nodes have sufficient capacity or the pieces baggage are immediately transported to the planes upon arrival.
- A3 The movement of DCVs on the network is approximated by a continuous flow of DCVs.
- A4 The DCV travel time on each link is an integer multiple of the sampling time  $\Delta t$ .

Assumption A3 is necessary for tractability of the control problem. Even though the number of DCVs is an integer in reality, for a fairly large number of DCVs, the movement of DCVs can be approximated by continuous flows. This is not very restrictive as the computed flows can then be realized as well as possible by a lower-level control loop that determines the optimal switching pattern for the switch controllers at the junctions. Assumption A4 allows us to arrive at a linear discrete-time model of the system. In this setup, we control the flows of DCVs within the network. The flows are indexed based on their destinations, enabling us to distinguish between loaded DCVs and empty DCVs, and also loaded or empty DCVs with different destinations among themselves. The DCV flows with an index  $o \in V_1$  refer to empty-DCV flows whereas the DCV flows with an index  $d \in V_3$  refer to loaded-DCV flows. Consequently, partial DCV queues associated with different destinations occur along the links of the network. The total DCV queue length along a link is then given as the sum of such partial queue lengths. The system is composed of baggage and empty-DCV vertical queues at the loading stations, empty-DCV vertical queues at the unloading stations, loaded-DCV queues at the EBS, and empty and loaded DCV queues along the links of the network, as shown in Fig. 2.4. Note that certain links of the network may carry both empty and loaded DCV flows. Moreover, the loading stations, the EBS, and the links have limited capacity. In our mathematical model, we make use of the following notation:

- For each node  $v \in V$ ,  $L_v^{\text{in}}$  is the set of incoming links of v and  $L_v^{\text{out}}$  is the set of outgoing links of v.
- For each link l = (v, w) of the network,  $q_{l,p,z}$  is the flow from the end of link l to link  $p \in L_w^{\text{out}}$ , with destination z.
- For each destination *z*, *L<sub>z</sub>* denotes to the set of links that are on some directed path to *z*.
- $s_l$  is the length of link l. The speed of DCVs is denoted by  $v_{\text{DCV}}$ . Moreover  $\Delta t$  is the sampling time.
- For each  $z \in V_3$ ,  $k_z^{\text{open}}$  and  $k_z^{\text{close}}$  mark, respectively, the beginning and the end of the time window of destination z, and  $k_{v,z}^{\text{nom}}$  is the nominal travel time from  $v \in V_1 \cup V_2 \cup \{v^*\}$  to z. Moreover,  $k_{v,z}^1 = k_z^{\text{open}} k_{v,z}^{\text{nom}}$  and  $k_{v,z}^2 = k_z^{\text{close}} k_{v,z}^{\text{nom}}$  are respectively the *relative* opening time-step and relative closing time-step of destination z as seen from v.

#### 2.2.2. Model Description

Now we will derive the dynamical model of the baggage handling system in discrete time under the assumptions A1-A7. In the sequel, x(k), k = 0, 1, ... denotes the value of x at time step k, and  $\begin{bmatrix} x \end{bmatrix}$  denotes the smallest integer bigger than or equal to x.

#### **Loading Stations**

For each loading station node o, let  $l_o^{\text{in}} = (w_o, o)$  and  $l_o^{\text{out}} = (o, w_o)$  respectively be the virtual incoming link and virtual outgoing link of o for some  $w_o \in V \setminus V_1$ . The control variables at each loading station are the flows of loaded DCVs,  $q_{o,l_o^{\text{out}},d}$ , from the DCV stack at o to  $l_o^{\text{out}}$  with destination  $d \in V_3$ , and the flow of empty DCVs,  $q_{l_o^{\text{in}},o,o}$ , from  $l_o^{\text{in}}$  to o with destination o. Fig. 2.4 illustrates how the flow variables are defined for the loading stations. For each outgoing link p of  $w_o$ , we need to impose the following constraints for all  $k \in \mathbb{N}$ :

$$\begin{aligned} q_{l_0^{\text{out}},p,z}(k) &= 0, \forall \ z \in V_1 \\ q_{l_0^{\text{out}},p,z}(k) &= 0, \forall \ z \in V_3 \text{ s.t. } p \notin L_z \\ q_{l_0^{\text{out}},p,z}(k) &\ge 0, \forall \ z \in V_3 \text{ s.t. } p \in L_z. \end{aligned}$$

$$(2.1)$$

For each incoming link p of  $w_{\alpha}$ , we impose the following constraints for all  $k \in \mathbb{N}$ :

$$q_{p,l_{o}^{\text{in}},z}(k) = 0, \forall \ z \in V_{3}$$

$$q_{p,l_{o}^{\text{in}},z}(k) = 0, \forall \ z \in V_{1} \setminus \{o\}$$

$$q_{p,l_{o}^{\text{in}},z}(k) \ge 0, z = o.$$
(2.2)

Note that constraint (2.1) implies that the loading station o can only send loaded-DCV flows to the unloading stations that are reachable from o. Constraint (2.2) implies that the loading station o can only accept empty-DCV flows that are designated for o. The evolution of DCV stack at loading station o,  $x_o(k)$ , is then given by

$$\begin{aligned} \mathbf{x}_{o}(k+1) &= \mathbf{x}_{o}(k) + \Delta t \Big( q_{l_{o}^{\text{in}},o,o}(k) - \sum_{d \in V_{3}} q_{o,l_{o}^{\text{out}},d}(k) \Big) \\ 0 &\leq \mathbf{x}_{o}(k) \leq \mathbf{x}_{o,\max}, \end{aligned}$$
(2.3)

where  $x_{0,\max}$  is the maximum capacity of the DCV stack at loading station o. Let  $x_{o,d}^{\text{bag}}(k)$  be the length of the baggage queue, with destination  $d \in V_3$ , at loading station *o*. Then,  $\boldsymbol{x}_{o,d}^{\text{bag}}(k)$  is described as

$$\begin{aligned} \mathbf{x}_{o,d}^{\text{bag}}(k+1) &= \mathbf{x}_{o,d}^{\text{bag}}(k) + \Delta t \big( Q_{o,d}(k) - q_{o,l_o^{\text{out}},d}(k) \big) \end{aligned} \tag{2.4} \\ \mathbf{x}_{o,d}^{\text{bag}}(k) &\geq 0, \ \forall \ d \in V_3, \end{aligned}$$

where  $Q_{o,d}(k)$  is the time varying baggage demand at loading station o that needs to be transported to destination d. The total length of baggage queue at node o is given by  $x_o^{\text{bag}}(k) = \sum_{d \in V_3} x_{o,d}^{\text{bag}}(k)$ . In order to guarantee that there is no DCV queue along the virtual outgoing link  $l_o^{out}$  and the virtual incoming link  $l_o^{in}$ , their inflow and outflow must be set equal, or equivalently

$$q_{o,l_o^{\text{out}},d}(k) = \sum_{p \in L_{w_o}^{\text{out}}} q_{l_o^{\text{out}},p,d}(k+k_{l_o^{\text{out}}}), \ \forall \ d \in V_3$$

$$q_{l_o^{\text{in}},o,o}(k+k_{l_o^{\text{in}}}) = \sum_{p \in L_{w_o}^{\text{in}}} q_{p,l_o^{\text{in}},o}(k),$$
(2.5)

where  $k_{l_{0}^{out}}$  is the number of time steps required to load a piece of baggage onto the DCVs, and  $k_{l_a^{in}}$  is the number of time steps that is required to store empty DCVs in the DCV stack.

#### **Unloading Stations**

For each unloading station node d, let  $l_d^{\text{in}} = (w_d, d)$  and  $l_d^{\text{out}} = (d, w_d)$  respectively be the virtual incoming link and virtual outgoing link of d for some  $w_d \in V_2$ . The control variables at each unloading station are the flows of empty DCVs,  $q_{d,l^{out},o}$ ,

from *d* to  $l_d^{\text{out}}$  with destination *o*, and the flows of loaded DCVs,  $q_{l_d^{\text{in}},d,d}$ , from  $l_d^{\text{in}}$  to *d* with destination *d*. For each outgoing link *p* of  $w_d$ , we impose the following constraints for all  $k \in \mathbb{N}$ :

$$\begin{aligned} q_{l_d^{\text{out}}, p, z}(k) &= 0, \forall \ z \in V_3, \\ q_{l_d^{\text{out}}, p, z}(k) &= 0, \forall \ z \in V_1 \text{ s.t. } p \notin L_z \\ q_{l_d^{\text{out}}, p, z}(k) &\ge 0, \forall \ z \in V_1 \text{ s.t. } p \in L_z. \end{aligned}$$

$$(2.6)$$

For each incoming link p of  $w_d$ , we need to impose the following constraints for all  $k \in \mathbb{N}$ :

$$\begin{aligned} q_{p,l_{d}^{\text{in}},z}(k) &= 0, \forall \ z \in V_{1}, \\ q_{p,l_{d}^{\text{in}},z}(k) &= 0, \forall \ z \in V_{3} \setminus \{d\}, \\ q_{p,l_{d}^{\text{in}},z}(k) &\geq 0, z = d. \end{aligned}$$
(2.7)

Constraint (2.6) implies that unloading station d can only send empty-DCV flows to the loading stations that are reachable from d. Constraint (2.7) implies the unloading station d can only accept loaded-DCV flows the final destination of which is d. The evolution of the DCV stack at unloading station d is given by

$$\mathbf{x}_{d}(k+1) = \mathbf{x}_{d}(k) + \Delta t \left( q_{l_{d}^{\text{in}},d,d}(k) - \sum_{o \in V_{1}} q_{d,l_{d}^{\text{out}},o}(k) \right),$$

$$0 \le \mathbf{x}(k) \le \mathbf{x}_{d,\max},$$

$$(2.8)$$

where  $x_{d,\max}$  is the maximum capacity of the DCV stack at unloading station *d*. Since, by definition, no DCV queues can occur along the virtual links of *d*, we have to set their inflow equal to their outflow, or equivalently

$$q_{d,l_{d}^{\text{out}},o}(k) = \sum_{p \in L_{w_{d}}^{\text{out}}} q_{l_{d}^{\text{out}},p,o}(k+k_{l_{d}^{\text{out}}}), \forall \ o \in V_{1},$$

$$q_{l_{d}^{\text{in}},d,d}(k+k_{l_{d}^{\text{in}}}) = \sum_{p \in L_{w_{d}}^{\text{in}}} q_{p,l_{d}^{\text{in}},d}(k),$$
(2.9)

where  $k_{l_d^{\text{out}}}$ , and  $k_{l_d^{\text{in}}}$  are respectively the number of time steps that is required to release the DCVs stored in the DCV stack, and the number of time steps that is required to unload and store the DCVs in the DCV stack at the unloading station.

EBS

Let  $l_{v^*}^{\text{out}} = (v^*, w^*)$  and  $l_{v^*}^{\text{in}} = (w^*, v^*)$  be the virtual outgoing and incoming links of EBS, respectively. For the EBS node  $v^*$  and for each  $d \in V_3$ , the control variables are the outflows of loaded DCVs,  $q_{v^*, l_{v^*}^{\text{out}}, d}$ , with destination d, and the inflows of

loaded DCVs,  $q_{l_{v^*}^{in},v^*,d}$ , whose final destination is d. For each outgoing link p of  $w^*$ , we introduce the following constraints for all  $k \in \mathbb{N}$ :

$$\begin{aligned} q_{l_{v^*}^{\text{out}},p,z}(k) &= 0, \forall \ z \in V_1, \\ q_{l_{v^*}^{\text{out}},p,z}(k) &= 0, \forall \ z \in V_3 \text{ s.t. } v^* \notin L_z, \\ q_{l_{v^*}^{\text{out}},p,z}(k) &\ge 0, \forall \ z \in V_3 \text{ s.t. } v^* \in L_z. \end{aligned}$$
(2.10)

For each incoming link p of  $w^*$ , we impose the following constraints for all  $k \in \mathbb{N}$ :

$$\begin{aligned} q_{p,l_{v^{*},z}^{\text{in}}}(k) &= 0, \forall \ z \in V_{1}, \\ q_{p,l_{v^{*},z}^{\text{in}}}(k) &= 0, \forall \ z \in V_{3} \text{ s.t. } v^{*} \notin L_{z} \\ q_{p,l_{v^{*},z}^{\text{in}}}(k) &\geq 0, \forall \ z \in V_{3} \text{ s.t. } v^{*} \in L_{z}. \end{aligned}$$
(2.11)

Constraints (2.10) and (2.11) jointly imply that the EBS cannot accept or send out empty-DCV flows, and that it can only receive and send loaded-DCV flows the final destination of which is reachable from the EBS. The evolution of the loaded-DCV queue lengths at the EBS with final destination  $d \in V_3$  is given as

$$\begin{aligned} \boldsymbol{x}_{\nu^*,d}(k+1) &= \boldsymbol{x}_{\nu^*,d}(k) + \Delta t \big( q_{l_{\nu^*}^{\text{in}},\nu^*,d}(k) - q_{\nu^*,l_{\nu^*}^{\text{out}},d}(k) \big), \end{aligned} \tag{2.12} \\ \boldsymbol{x}_{\nu^*,d}(k) &\geq 0, \end{aligned}$$

The total length of the DCV queues at the EBS is, therefore, given by  $x_{v^*}(k) = \sum_{d \in V_3} x_{v^*,d}(k)$  with the constraint  $x_{v^*}(k) \le x_{v^*,\max}$ , where  $x_{v^*,\max}$  is the maximum capacity of EBS. The following guarantee that no queues occur along the virtual

capacity of EBS. The following guarantee that no queues occur along the virtual links of EBS:

$$q_{l_{\nu^*,\nu^*,d}^{\text{in}}(k+k_{l_{\nu^*}^{\text{in}}})} = \sum_{p \in L_{w^*}^{\text{in}}} q_{p,l_{\nu^*,d}^{\text{in}}(k)}$$

$$q_{\nu^*,l_{\nu^*}^{\text{out}},d}(k) = \sum_{p \in l_{w^*}^{\text{out}}} q_{l_{\nu^*}^{\text{out}},p,d}(k+k_{l_{\nu^*}^{\text{out}}}),$$
(2.13)

where  $k_{l_{v^*}^{\text{in}}}$ , and  $k_{l_{v^*}^{\text{out}}}$  are respectively the number of time steps that is required to store loaded DCVs in the EBS, and the number of time steps that is required to release loaded DCVs from the EBS.

Links

For each real link l = (v, w) and for each  $z \in V_1 \cup V_3$ , the controls are the empty and loaded DCV flows,  $q_{l,p,z}$ , from the link l to each of its outgoing links p with destination z. For each  $p \in L_w^{out}$ , the flows of DCVs from l to p must satisfy

$$q_{l,p,z}(k) = 0, \forall z \in V_1 \cup V_3 \text{ s.t. } p \notin L_z,$$

$$q_{l,p,z}(k) \ge 0, \text{otherwise},$$

$$(2.14)$$

for all  $k \in \mathbb{N}$ , which implies that the DCVs with final destination z can be sent from l to p if z is reachable from p. Let  $F_{l,z}^{\text{in}}(k)$  be the sum of all DCV flows with destination z that enter link l, and let  $F_{l,z}^{\text{out}}(k)$  be the sum of all DCV flows with destination z that leaves link l. We then have

$$F_{l,z}^{\text{in}}(k+k_{l}(k)) = \sum_{p \in L_{\nu}^{\text{in}}} q_{p,l,z}(k),$$

$$F_{l,z}^{\text{out}}(k) = \sum_{p \in L_{\nu}^{\text{out}}} q_{l,p,z}(k),$$
(2.15)

where  $k_l(k)$  is the number of travel time steps for DCV on the link l given by  $k_l(k) = \left[\frac{s_l - x_l(k)l_{\text{DCV}}}{v_{\text{DCV}}\Delta t}\right]$ , where  $x_l(k) = \sum_{z \in V_1 \cup V_3} x_{l,z}(k)$ ,  $x_l(k) \le x_{l,\text{max}}$  is the total DCV

queue length along link l with  $x_{l,\max}$  being the maximum allowed queue length on link l and with  $x_{l,z}(k)$  being the partial DCV queue length along link l associated with destination d described by

$$\begin{aligned} \mathbf{x}_{l,z}(k+1) &= \mathbf{x}_{l,z}(k) + \Delta t \big( F_{l,z}^{\text{in}}(k) - F_{l,z}^{\text{out}}(k) \big), \\ \mathbf{x}_{l,z}(k) &\ge 0. \end{aligned}$$
(2.16)

The equality and inequality constraints (2.1)-(2.16) define the set of feasible trajectories of the system. Let  $x(k) \in \mathbb{R}^n$  be the state vector at time step k the elements of which are the baggage queue lengths at the loading stations; the empty-DCV queues at the loading stations, at the unloading stations, and on the links; the loaded-DCV queues on the links and in the EBS; the empty- and loaded-DCV flows at the past time steps, where the number of past time steps for which we need to store the flow value depends on the capacity of the link. Let  $u(k) \in \mathbb{R}^{m_1}$  be the input vector the elements of which are empty- and loaded-DCV flows from the end of each link to its outgoing links; the inflow of empty DCVs from the virtual incoming link of loading stations to the loading stations and the outflow of empty DCVs from unloading stations to their virtual outgoing link; the flow of loaded DCVs to the EBS from its virtual incoming link, and from the EBS to its virtual outgoing link, and let  $z(k) \in \mathbb{R}^{m_2}$  be the disturbance vector the elements of which are  $Q_{o,d}(k)$ for all origin and destination pairs. Note that we consider the baggage demand  $Q_{o,d}(k)$  as a disturbance since it is an exogenous input that is fully measurable or predictable. We define the output vector  $y(k) \in \mathbb{R}^p$  as the collection of queue, i.e.,  $x_o(k)$ ,  $x_d(k)$ ,  $x_{v^*,d}(k)$ ,  $x_{l,z}(k)$ ,  $x_{o,d}^{bag}(k)$ . Then, the system description can be expressed in the form

$$\begin{aligned} \mathbf{x}(k+1) &= A(\mathbf{x}(k))\mathbf{x}(k) + B\mathbf{u}(k) + G\mathbf{z}(k), \\ \mathbf{y}(k) &= C\mathbf{x}(k), \\ E\mathbf{u}(k) &= F\mathbf{x}(k), \\ 0 &\leq \mathbf{y}(k) \leq \mathbf{y}_{\max}, \\ 0 &\leq \mathbf{u}(k) \leq \mathbf{u}_{\max}, \end{aligned}$$
(2.17)



for properly defined matrices A, B, C, E, F, and G. Note that initial state of the system 2.17, and the demand profile  $Q_{o,d}(\cdot)$  have to be properly defined for the system to have solution for any  $k \in \mathbb{N}$ . Note that depending on the value of x(k), certain elements of A(x(k)) have non-zero values. Therefore, (2.17) *does not* define a linear discrete-time system with linear constraints.

#### **2.3. MPC Problem Formulation**

In this section, we use the dynamic model introduced in Section 2.2 within the context of MPC. At every time step, based on the current state of the system and a future prediction of baggage demands, a constrained finite horizon optimization problem will be solved yielding a sequence of optimal controls. According to the receding horizon policy, only the first step of this sequence is applied to the system, and this process is repeated at the next time step [10–12].

#### **2.3.1.** Objective Function

The aim of the control scheme is to assure delivery of pieces of baggage to the unloading stations within the given time windows. Imposing explicit constraints on delivery times would only be possible using a model that gives exact arrival times. However, due to its complexity, the resulting optimization problem would be intractable. Moreover, imposing explicit constraints on delivery times could lead to an infeasible optimization problem. Using the proposed flow model, the arrival time of DCVs to the unloading stations cannot be explicitly computed. Therefore, we include time-window constraints as a soft constraint in the objective function. Hence, the penalty functions penalize the flows and DCV queue lengths in such a way that the loaded DCVs are delivered to the unloading stations within their respective time windows as well as possible. Recall from Section 2.2 that  $k_{v,z}^1$  and  $k_{v,z}^2$  are respectively the relative opening and closing time steps of destination z, as seen from node v. For  $v \in V_1 \cup V_2 \cup v^*$  and  $z \in V_3$ , the weighting function

$$C^{q}(k, v, z) \coloneqq \begin{cases} r_{0} & \text{if } k \leq k_{v,z}^{1} \\ r_{0} + m_{1}(k - k_{v,z}^{1}) & \text{if } k_{v,z}^{1} < k \leq k_{v,z}^{2} \\ r_{0} + m_{1}(k_{v,z}^{2} - k_{v,z}^{1}) + m_{2}(k - k_{v,z}^{2}) & \text{if } k > k_{v,z}^{2}, \end{cases}$$
(2.18a)
depicted in Fig. 2.7, will be used in the penalty terms for the baggage queues in the loading stations, loaded DCV-queues in the EBS and along the links. For  $v \in V_1 \cup V_2$  and  $z \in V_3$ , the weighting function

$$C^{f}(k, v, z) \coloneqq \begin{cases} s_{1} & \text{if } k \leq k_{v, z}^{2} \\ s_{1} + n_{2}(k - k_{v, z}^{2}) & \text{if } k > k_{v, z}^{2}, \end{cases}$$
(2.18b)

shown in Fig. and Fig. 2.8, will be used in the penalty terms associated with the loaded-DCV flows in the loading stations and along the links. For  $z \in V_3$ , the weighting functions

$$C_{v^*}^{\text{f,in}}(k,z) \coloneqq \begin{cases} 0 & \text{if } k \le k_{v^*,z}^1 \\ n_1(k-k_{v^*,z}^1) & \text{if } k > k_{v^*,z}^1 \end{cases}$$
(2.18c)

depicted in Fig. 2.5, and

$$C_{v^*}^{\text{f,out}}(k,z) \coloneqq C_{v^*}^{\text{f,in}}(k,z) - n_1(k - k_{v^*,z}^1)$$
(2.18d)

will be respectively used to penalize the the EBS inflows and the EBS outflows. For  $z \in V_3$ , the weighting function

$$C^{f}(k,z) \coloneqq \begin{cases} s_{0} - n_{1}(k - k_{z}^{\text{open}}) & \text{if } k < k_{z}^{\text{open}} \\ s_{0} & \text{if } k_{z}^{\text{open}} < k \le k_{z}^{\text{close}} \\ s_{0} + n_{2}(k - k_{z}^{\text{close}}) & k > k_{z}^{\text{close}}, \end{cases}$$
(2.18e)

depicted in Fig. 2.6, will be used to penalize the empty-DCV outflows at unloading stations. In above,  $r_0$ ,  $m_2 > m_1$ ,  $s_0$ ,  $s_1$ ,  $n_1$ , and  $n_2$  are strictly positive constants that determine the shape of the weighting function. Note that the relative magnitude of these constants is a design parameter that determines how much in-time delivery of DCVs is favored to energy consumption.

Using 2.18, we will define the following penalty terms, which penalize the loaded-DCV flows and queues. For loading station o, the penalty terms at time step k associated with the baggage queues and the loaded-DCV flows are defined as:

$$J_{\rm LS}^{\rm bag}(k) \coloneqq \sum_{o \in V_1} \sum_{d \in V_3} C^{\rm q}(k, o, d) x_{o,d}^{\rm bag}(k),$$
(2.19)  
$$J_{\rm LS}^{\rm flow}(k) \coloneqq \sum_{o \in V_1} \sum_{d \in V_3} C^{\rm f}(k, o, d) q_{o,l_0^{\rm out}, d}(k).$$

For links l = (v, w), the loaded-DCV queues and loaded-DCV flows are penalized as follows:

$$J_{\mathrm{L}}^{\mathrm{DCV}}(k) \coloneqq \sum_{d \in V_3} \sum_{l=(v,w) \in L_d} C^{\mathrm{q}}(k,w,d) x_{l,d}(k), \qquad (2.20)$$
$$J_{\mathrm{L}}^{\mathrm{flow}}(k) \coloneqq \sum_{d \in V_3} \sum_{l=(v,w) \in L_d} \left( C^{\mathrm{f}}(k,w,d) \sum_{p \in L_w^{\mathrm{out}}} q_{l,p,d}(k) \right),$$

For the EBS, penalty term associated with the loaded-DCV queues is defined as:

$$J_{\rm EBS}^{\rm DCV}(k) \coloneqq \sum_{d \in V_3} (C^{\rm q}(k, v^*, d) - r_0) x_{v^*, d}(k),$$
(2.21)

and the penalty terms associated with loaded-DCV flows in and out of the EBS are defined as:

$$J_{\text{EBS}}^{\text{inflow}}(k) \coloneqq \sum_{d \in V_3} C_{v^*}^{\text{f,in}}(k,d) q_{l_{v^*}^{\text{in}},v^*,d}(k), \qquad (2.22)$$
$$J_{\text{EBS}}^{\text{outflow}}(k) \coloneqq \sum_{d \in V_3} C_{v^*}^{\text{f,out}}(k,d) q_{v^*,l_{v^*}^{\text{out}},d}(k),$$

For the unloading stations *d*, the penalty term associated with the inflow of loaded DCVs to *d* is defined as:

$$J_{\mathrm{US}}^{\mathrm{flow}}(k) \coloneqq \sum_{d \in V_3} C^{\mathrm{f}}(k, d) q_{l_d^{\mathrm{in}}, d, d}(k).$$
(2.23)

One can observe that for  $k \leq k_{o,d}^1$ , as depicted in Fig. 2.7, we assign a constant weight  $r_0$  to the baggage queues at the loading station and to the loaded-DCV queues along the links. With the choice of  $s_0 \ll r_0$  this allows for early release of baggage, hence loaded-DCVs, into the network. These DCVs will move to the EBS since the loaded-DCV inflow of the unloading stations is highly penalized (see Fig. 2.6) and since the inflow of DCVs to the EBS inflicts no cost (see Fig. 2.5). Please note that, during this period, the DCVs will remain in the EBS since dispatching the DCVs is more expensive. For  $k_{o,d}^1 < k \le k_{o,d}^2$ , the weighting functions for the baggage queues at the loading stations and for the loaded-DCV queues along the links and in the EBS increase with the constant slope  $m_1$  to have more loaded DCVs released in the network. The weight of the loaded-DCV inflows to the EBS increases with the constant slope  $n_1$  to prevent DCVs from entering the EBS. Moreover the inflow of loaded DCVs to the unloading station involves no cost. Hence, the released DCVs arrive at the specified unloading stations. For  $k > k_{o,d}^2$ the weighting functions for the baggage queues at the loading stations and for the loaded-DCV queues along the links increase with the constant slope  $m_2$ . In addition the weighting functions of the loaded-DCV flows along the links and into the unloading stations increases with the constant slope  $n_2$ . Since the slope of the second part of the weighting functions is larger than the slope of the first part, the case of having loaded DCVs on the links, or having loaded-DCV flows arriving at the unloading stations during this time interval becomes expensive.

To take into account the energy consumption, we define the following penalty terms, which penalize the flow of empty DCVs. The cost of empty-DCV flows,  $q_{l,p,o}(k)$  is given by

$$J^{e}(k) \coloneqq t_{1} \sum_{o \in V_{1}} \sum_{l = (v, w) \in L_{o}} \sum_{p \in L_{w}^{out}} q_{l, p, o}(k),$$
(2.24)

where we assign a constant weight  $t_1 > 0$  to empty DCV flows to avoid arbitrary empty DCV flow circulations, which is in accordance with our objective of minimizing the energy consumption. We define

$$J_1(k) \coloneqq J_{\rm LS}^{\rm pag}(k) + J_{\rm LS}^{\rm DCV}(k) + J_{\rm L}^{\rm DCV}(k) + J_{\rm EBS}^{\rm DCV}(k),$$
(2.25)

and

 $J_2(k) \coloneqq J_{\text{LS}}^{\text{flow}}(k) + J_{\text{L}}^{\text{flow}}(k) + J_{\text{US}}^{\text{flow}}(k) + J_{\text{EBS}}^{\text{inflow}}(k) + J_{\text{EBS}}^{\text{outflow}}(k)$ (2.26)

respectively as the total cost of baggage stacks and loaded DCV queues and the total cost of loaded DCV flows. The total cost function at time step k over the prediction horizon  $N_{\rm p}$  is then given by

$$J_{N_{p}}(k) \coloneqq \frac{1}{J_{1,\text{nom}}} \sum_{j=1}^{N_{p}} J_{1}(k+j) + \frac{\alpha_{1}}{J_{2,\text{nom}}} \sum_{j=0}^{N_{p}-1} J_{2}(k+j) + \frac{\alpha_{2}}{J_{\text{nom}}^{e}(k)} \sum_{j=0}^{N_{p}-1} J^{e}(k)(k+j),$$
(2.27)

where  $\alpha_1 > 0$ , and  $\alpha_2 > 0$  are constants indicating the relative importance of the respective component of the objective function, and where  $J_{i,\text{nom}}$ , i = 1, 2 is the nominal value<sup>2</sup> of  $J_i(k)$ .

#### **2.3.2. Linear Programming Approach**

In general, finding the optimal flow values and optimal queue lengths is a nonlinear programming problem. To arrive at a linear programming formulation we first make the following additional assumption:

#### A-LP The queue lengths remain constant over the prediction horizon.

Assumption A-LP enables us to arrive at a linear programming formulation of the optimization problem. Under A-LP, the first equation in (2.17) can be written as

$$x(k+j+1) = A(x(k))x(k+j) + Bu(k+j) + Gz(k+j),$$
(2.28)

for  $j = 1, ..., N_p - 1$ , which, together with constraints of (2.17), defines a linear discrete time-invariant system with linear constraints. Since (2.27) is weighted sum of the state variables and the input variables, it can be expressed as  $J_{N_p}(k) = F_0^T(k)\mathbf{z}_{N_p}(k) + F_1^T(k)\mathbf{x}(k) + F_2^T(k)\mathbf{u}_{N_p}(k)$  by successive substitution in (2.28). Here,  $\mathbf{z}_{N_p}(k) = [\mathbf{z}^T(k), ..., \mathbf{z}^T(k + N_p - 1)]^T \in \mathbb{R}^{m_2N_p}$  is the predicted disturbance vector,  $\mathbf{u}_{N_p}(k) = [\mathbf{u}^T(k), ..., \mathbf{u}^T(k + N_p - 1)]^T \in \mathbb{R}^{m_1N_p}$  is the control input sequence and  $F_0(k) \in \mathbb{R}^{m_2N_p}$ ,  $F_1(k) \in \mathbb{R}^n$ , and  $F_2(k) \in \mathbb{R}^{m_1N_p}$  are coefficient vectors. Note that since the values of the weighting functions are known for  $k, k + 1, ..., k + N_p - 1$ ,

<sup>&</sup>lt;sup>2</sup>The nominal value  $J_{i,\text{nom}}$ , i = 1, 2 can be computed by averaging over  $J_i(k)$  for k = 1, ..., N, where  $J_i(k)$  is obtained based on simulating the system under the nominal input  $u_{\text{nom}}(k)$  and the nominal baggage demand  $z_{\text{nom}}(k)$ , for k = 1, ..., N.

 $F_0(k)$ ,  $F_1(k)$ , and  $F_2(k)$  are fully determined at time step k. Therefore, at every time step k, we solve an optimization problem of the form

$$\min_{\boldsymbol{u}_{N_{p}}(k)} \boldsymbol{F}_{2}^{1}(k) \boldsymbol{u}_{N_{p}}(k)$$
  
subject to  $A^{eq}(k) \boldsymbol{u}_{N_{p}}(k) = \boldsymbol{b}^{eq}(k),$   
 $A^{ineq}(k) \boldsymbol{u}_{N_{p}}(k) \leq \boldsymbol{b}^{ineq}(k),$  (2.29)  
 $0 \leq \boldsymbol{u}_{N_{p}}(k) \leq \boldsymbol{u}_{max},$ 

where the matrices  $A^{eq}(k)$ , and  $A^{ineq}(k)$  and the vectors  $b^{eq}(k)$ , and  $b^{ineq}(k)$  are obtained based on (2.17), the current state of the system x(k), and the predicted baggage demand  $\mathbf{z}_{N_p}(k) = [\mathbf{z}^T(k), ..., \mathbf{z}^T(k + N_p - 1)]^T \in \mathbb{R}^{m_2 N_p}$ . This is an LP problem, which, among others, can be solved efficiently using the simplex, active-set, or interior point methods [21].

#### **2.3.3.** Iterative Linear Programming Approach

The prediction model (2.28) can yield inaccurate predictions. To remedy this problem and still arrive at a tractable formulation for the optimization problem, we propose an iterative version of (49): at each iteration, a problem of the form (2.29) is solved to find the optimal sequence of flow variables. This sequence is then used in the model (2.17) in forward simulation to compute the updated values for queue lengths resulting from the obtained flow values. The updated queue lengths are then used again in the LP problem formulation to find updated flow values. This process is repeated for a certain number of iterations. More formally, we replace the first equation in (2.17) by the prediction model

$$\boldsymbol{x}_{(i)}(k+j+1) = A(\boldsymbol{x}_{(i-1)}(k+j))\boldsymbol{x}_{(i)}(k+j) + B\boldsymbol{u}_{(i)}(k+j) + G\boldsymbol{z}(k+j), \quad (2.30)$$

for  $j = 0, ..., N_p - 1$  with the initial conditions  $x_{(0)}(k + j) = x(k)$  for  $j = 0, ..., N_p - 1$ ,  $x_{(i)}(k) = x(k)$  for i = 1, 2, ..., where *i* is the iteration index. Hence, at time step *k*, initializing the algorithm with the initial predicted state <sup>3</sup>  $x_{N_p,(0)}(k) = [x^T(k), ..., x^T(k)]^T$ , the predicted state vector

$$\boldsymbol{x}_{N_{\text{p}},(i)}(k) = [\boldsymbol{x}_{(i)}^{\text{T}}(k), ..., \boldsymbol{x}_{(i)}^{\text{T}}(k+N_{\text{p}}-1)]^{\text{T}}$$

at iteration *i* is computed based on the input sequence,

$$\boldsymbol{u}_{N_{\mathrm{p}},(i)}(k) = \left[\boldsymbol{u}_{(i)}^{\mathrm{T}}(k), ..., \boldsymbol{u}_{(i)}^{\mathrm{T}}(k+N_{\mathrm{p}}-1)\right]^{\mathrm{T}}$$

at iteration *i*, the predicted disturbance vector  $\mathbf{z}_{N_p}(k)$ , and the predicted state vector at iteration i - 1. Using (2.30), the objective function at time step k in iteration i

<sup>&</sup>lt;sup>3</sup>If the optimal input sequence from the previous time step is availabe, it can be used to compute the initial predicted state.

be expressed as  $J_{N_{p},i}(k) = F_{0,(i-1)}^{T}(k)\mathbf{z}_{N_{p}}(k) + F_{1,(i-1)}^{T}(k)\mathbf{x}(k) + F_{2,(i-1)}^{T}(k)\mathbf{u}_{N_{p},(i)}(k)$ . Therefore, at time step k and in iteration i, we solve an optimization problem of the following form:

$$\min_{\boldsymbol{u}_{N_{p},(i)}(k)} \boldsymbol{F}_{2,(i-1)}^{T}(k) \boldsymbol{u}_{N_{p},(i)}(k) 
\text{subject to } A_{(i-1)}^{\text{eq}}(k) \boldsymbol{u}_{N_{p},(i)}(k) = \boldsymbol{b}_{(i-1)}^{\text{eq}}(k), \qquad (2.31) 
A_{(i-1)}^{\text{ineq}}(k) \boldsymbol{u}_{N_{p},(i)}(k) \leq \boldsymbol{b}_{(i-1)}^{\text{ineq}}(k),$$

where the matrices  $A_{(i-1)}^{eq}(k)$ , and  $A_{(i-1)}^{ineq}(k)$  and the vectors  $b_{(i-1)}^{eq}(k)$ , and  $b_{(i-1)}^{ineq}(k)$  are obtained based on (2.30), and the constraints of (2.17) using the predicted state vector  $x_{N_{p},(i-1)}(k)$  at iteration i - 1, the current state of the system x(k), and the predicted baggage demand  $z_{N_{p}}(k)$ . Note that even though we assume constant queue lengths at each inner iteration of (2.31), the variations in queue lengths over the prediction horizon are taken into account in the next inner iteration. Therefore, at the end of inner iterations, the computed flow values are the ones obtained having taken into account the effect of variations in the queue lengths over the prediction horizon.

#### **2.3.4.** Nonlinear Programming Approach

In this approach, without any simplifying assumptions, we make direct use of the model presented by (2.17) and the cost given by (2.27) in defining optimization problem of the following form:

$$\min_{\boldsymbol{u}_{N_{\mathrm{p}}}(k)=[\boldsymbol{u}^{\mathrm{T}}(k), \dots, \boldsymbol{u}^{\mathrm{T}}(k+N_{\mathrm{p}}-1)]^{\mathrm{T}}} J_{N_{\mathrm{p}}}(k) \text{ subject to (2.17),}$$
(2.32)

which is a nonlinear nonconvex problem. Hence, one can use multi-start local nonlinear optimization algorithms (e.g., an SQP algorithm or interior-point methods [21]) or global optimization methods (e.g., pattern search and genetic algorithms [22]).

# 2.4. Case Study

First, we introduce a state-of-the-art (SOA) method, which is developed based on our understanding of the method currently used for control of BHS by a leading supplier of DCV-based baggage handling systems. In this method, the flow of loaded DCVs at each junction is dynamically assigned to its outgoing links based on the projected deviation of delivery times from the beginning of the time window of the destination. In this way, the DCVs are dynamically routed via links that make timely delivery possible while distributing the traffic in a smart way, hence avoiding congestion. The empty-DCV flows at each junction are assigned based on projected travel times, and the length of the baggage queues at the loading stations. Hence, more empty DCVs are sent via "faster" links to the loading station with longer baggage queues. a) Routing and scheduling of loaded DCVs: at each junction of the network the flow from an incoming link *l* to an outgoing link *p* with destination *d* at current time  $t = k\Delta t$  is obtained as

$$q_{l,p,d}(k) = \frac{1/|t - t_d^{\text{open}} - t_{l,p,d}(k)|}{\sum_i 1/|t - t_d^{\text{open}} - t_{l,j,d}(k)|} q_{l,\max}$$

where  $t_d^{\text{open}}$  is the opening time for the destination,  $q_{l,\max}$  is the maximum flow of the link, and  $t_{l,p,d}$  is the estimated travel time from the end of link l to destination d via link p. This travel time can be obtained using historical data or by computing  $t_{l,p,d}(k) = t_p(k) + t_{l,p,d}^*$ , where  $t_{l,p,d}^*$  is the the travel time from l to d via the shortest path and  $t_p(k)$  is the estimated clearance time of the queues given by

$$t_p(k) = \frac{x_p(k)}{q_{l,\max}} + \frac{\sum_{j=0}^{n_p-1} F_p^{\text{in}}(k-j)(n_p-j)}{\sum_{j=0}^{n_p-1} F_p^{\text{in}}(k-j)} + \frac{\Delta t}{q_{l,\max}} \sum_{j=0}^{n_p-1} F_p^{\text{in}}(k-j), \quad (2.33)$$

where  $x_p(k)$  is the total queue length at the end of link p,  $F_p^{\text{in}}(k)$  is the total inflow of DCVs to link p, and  $n_p\Delta t$  is the DCV travel time from the beginning to the end of p. The first term on the right hand side of the (2.33) is the clearance time of current DCV queue at the end of link p. The second term determines the average time that the past DCV flows that are currently traveling on link p need to reach the end of the link, and the last term determines the clearance time of these flows once they have reached the end of the link.

b) Routing and scheduling of empty DCVs: at each junction of the network, the flow of empty DCVs from an incoming link *l* to an outgoing link *p* is with destination *o* is given as

$$q_{l,p,o}(k) = \frac{x_o^{\text{bag}}}{\sum_{j \in V_1} x_j^{\text{bag}}} \frac{1/t_{l,p,o}(k)}{\sum_j 1/t_{l,j,o}(k)} q_{l,\max},$$

where  $x_o^{\text{bag}}$  is the total baggage queue at loading station o, and  $t_{l,p,d}$  is the estimated travel time from the end of link l to destination d that can be obtained using historical data or the aforementioned procedure via (2.33). For the network layout depicted in Fig. 2.9 with  $V_1 = \{1,9\}$ ,  $V_3 = \{5,13\}$ ,  $V_2 = \{2,3,4,6,7,10,11,12\}$ , and  $v^* = \{8\}$ , we compare the the performance of the SOA, LP-MPC, ILP-MPC, and NLP-MPC against each other in terms of MPC-in-the-loop optimal cost and the computational burden under the demand scenario depicted in Fig. 2.10 using the parameters listed in Table 2.1. For the LP and ILP approaches, we use the CPLEX solver via TOMLAB toolbox for MATLAB, and for the NLP approach, we use the implementation of the interior-point algorithm in the MATLAB optimization toolbox. For each method, the total MPC-in-the-loop cost and the corresponding CPU time is listed in Table 2.2 for different prediction horizons using a dual core PC with Intel E8400 processor running at 3.00GHz and with 4GB of RAM. The reported CPU times are computed by averaging over the CPU times for all simulation time steps. It can be observed from Table 2.2 that while the computation time of the LP and ILP-MPC

Table 2.1: Closed-loop simulation parameters, including the total simulation time steps  $N_{sim}$ , destinations opening and closure time instants, MPC parameters, link capacities, and the initial value of the bags and empty DCVs at the loading stations. There are no DCVs initially in the other parts of the network.

| MPC Parameters                    |  |                            |  |                        |   |  |  |  |
|-----------------------------------|--|----------------------------|--|------------------------|---|--|--|--|
| Np                                | $(k_{U_1}^{\text{open}}, k_{U_1}^{\text{close}}), (k_{U_2}^{\text{open}}, k_{U_2}^{\text{close}})$ | $u_{\rm max}[{\rm DCV/s}]$ | $(\mathbf{x}_{o,\max}, \mathbf{x}_{d,\max}, \mathbf{x}_{v^*,\max}, \mathbf{x}_{l,\max})$ [DCV] | $(\alpha_1, \alpha_2)$ | $(s_0, s_1, n_1, n_2), (r_0, m_1, m_2)$ |  |  |  |
| 8                                 | (60,90), (80,120)  | 6                          | (50, 100, 120, 26)   | (1,1)                  | (1, 1, 1, 2), (0.5, 1, 1, 2)            |  |  |  |
| Closed-loop Simulation Parameters |  |                            |  |                        |   |  |  |  |
| $\Delta t[s]$                     | $(x_{L_1}(0), x_{L_2}(0), x_{U_1}(0), x_{U_2}(0))$ [DCV]   | N <sub>sim</sub>           | $v_{\rm DCV}[m/s]$   | $l_{\rm DCV}[m]$       |   |  |  |  |
| 0.5                               | (20, 49, 90, 0)  | 150                        | 10   | 1.5                    |   |  |  |  |

| Np  | Opt. Scheme (Solver) | Iter./Multi-start Iter. | min. CPU Time [s] | Avg. CPU Time [s] | max. CPU Time | Optimal Cost |
|-----|----------------------|-------------------------|-------------------|-------------------|---------------|--------------|
| N/A | SOA                  | N/A                     | 0.0015            | 0.0036            | 0.0411        | 16671.5      |
| 2   | LP (CPLEX)           | N/A                     | 0.0056            | 0.0074            | 0.0411        | 8083.1       |
| 2   | ILP (CPLEX)          | 2                       | 0.0128            | 0.0237            | 0.7401        | 8083.1       |
| 2   | ILP (CPLEX)          | 4                       | 0.0249            | 0.0389            | 0.7416        | 8083.1       |
| 2   | ILP (CPLEX)          | 6                       | 0.0475            | 0.0530            | 0.0739        | 8083.1       |
| 2   | NLP (interior-point) | 3                       | 99.6190           | 114.53            | 135.52        | 8083.1       |
| 4   | LP (CPLEX)           | N/A                     | 0.0074            | 0.0094            | 0.0239        | 7982.1       |
| 4   | ILP (CPLEX)          | 2                       | 0.0150            | 0.0271            | 0.7664        | 7869.5       |
| 4   | ILP (CPLEX)          | 4                       | 0.0309            | 0.0445            | 0.7845        | 7820.5       |
| 4   | ILP (CPLEX)          | 6                       | 0.0453            | 0.0514            | 0.0913        | 7800.5       |
| 4   | NLP (interior-point) | 4                       | 310.00            | 344.93            | 376.16        | 6813.4       |
| 6   | LP (CPLEX)           | N/A                     | 0.0087            | 0.0197            | 0.6674        | 7471.3       |
| 6   | ILP (CPLEX)          | 4                       | 0.0375            | 0.0457            | 0.0967        | 5890.2       |
| 6   | ILP (CPLEX)          | 8                       | 0.0721            | 0.0852            | 0.1133        | 5574.2       |
| 6   | NLP (interior-point) | 8                       | 419.62            | 450.15            | 486.42        | 4546.7       |
| 8   | LP (CPLEX)           | N/A                     | 0.1212            | 0.1902            | 0.3299        | 5925.3       |
| 8   | ILP (CPLEX)          | 6                       | 0.8061            | 1.0185            | 1.4325        | 4949.9       |

Table 2.2: Comparison of Closed-loop Performance and Computation Times

approaches are comparable to the SOA method, they outperform the SOA in terms of controller-in-the-loop cost. This can also be observed by comparing Fig. 2.15 with Fig. 2.14 which illustrate the controller-in-the-loop performance of the two methods for the baggage demand profile depicted in Fig. 2.10. While with the ILP-MPC with three iterations, 70% of the baggage demand for destination  $U_1 = 5$  and 80% of the baggage demand for destination  $U_2 = 13$  arrive within the respective time window of destination, with SOA method, only 10% of demand to  $U_1 = 5$  and 20% of demand to  $U_2 = 13$  arrive within the time windows. One can observe from Fig. 2.13 and from Table 2.2 that the NLP approach outperforms the LP and ILP approaches, but its computational burden increases very sharply for high values of prediction horizon. Moreover, the the optimal cost of ILP approach converges to the NLP approach by increasing the number of iterations with far less computation effort. We also observe from Table 2.2 that the ILP approach outperforms the LP approach in terms of the closed-loop cost and that the difference between the closed-loop cost of the two increases for the increasing values of the prediction horizon. Fig. 2.11 compares the computation times of SOA, LP, and ILP methods for different values of  $N_{\rm p}$ . It can be observed that the computation time of the LP approach is comparable with the SOA. More importantly, the computation time of LP and ILP methods increase linearly in the problem size. The computation time of ILP approach as a function of ILP iteration is depicted in Fig. 2.12.



Figure 2.9: Schematic layout of the case study network with loading stations  $L_1 = 1$ , and  $L_2 = 9$  and unloading stations  $U_1 = 5$ ,  $U_2 = 13$ , and EBS = 8. The nominal travel time of each link, based on the maximum DCV speed, is indicated in simulation time step.



Figure 2.10: The baggage demand (left) and number of DCVs in the EBS (right).



Figure 2.11: Computation time of LP and ILP in seconds as a function of  $N_{\rm p}$ .



Figure 2.12: Computation time of the ILP in seconds as a function of ILP iterations



Figure 2.13: Predicted cost as a function of closed-loop time step using the LP approach (dashed line), the ILP approach (square markers), and the NLP approach (triangle markers) for different prediction horizons  $N_{\rm p}$ .



Figure 2.14: ILP-MPC-based control. Triangle: total DCV inflow. Square: DCV outflow. Solid line: DCV queue for unloading stations (left)  $U_1$  and (right)  $U_2$  with the associated time window of destination.



Figure 2.15: SOA control. Triangle: total DCV inflow. Square: DCV outflow. Solid line: DCV queue for unloading stations (left)  $U_1$  and (right)  $U_2$  with the associated time window of destination.

# **2.5. Conclusions and Future Work**

In this chapter, we have revisited the problem of dynamic routing and scheduling of DCVs in baggage handling systems. We have jointly addressed the main control problems of DCV-based baggage handling systems, namely, routing and scheduling of DCVs, line balancing, and empty cart management. Our derived model was used as the prediction model within the MPC framework. The objective function was defined so as to penalize the deviation of baggage delivery time at the unloading stations from pre-specified time windows, and the energy consumption. We formulated the MPC problem as a nonlinear programming (NLP) problem, which, for the case study, was computationally very inefficient for large prediction horizons (e.g.,  $N_{\rm p} > 6$ ) due to the large-scale nature of the problem. Under some simplifying assumptions, we showed that the underlying optimization problem can be recast as a linear programming (LP) problem. However, the performance of the LP approach is suboptimal with respect to the NLP approach with significant performance loss for larger values of prediction horizons. To achieve a fast yet close-to-optimal solution, we have proposed an iterative linear programming (ILP) scheme that solves a sequence of LP problems to compute the optimal control sequence. We have illustrated that by using this method for an appropriate number of ILP iterations, it is possible to achieve a performance close to the NLP approach with significantly less computational burden. We also compared the performance of the our proposed approach to a state-of-the-art method (SOA) as is used in practice. We illustrated via simulations that our proposed method outperforms the SOA. The scalability of our approach was illustrated by numerical tests. Particularly, we showed that the required computation time increases linearly for increasing values of the prediction horizon and number of iterations.

For future work, we will compare the performance and computational requirements of our approach to an approach based on computing the exact travel time of each DCV in the network using a given distribution of passengers and baggage allowance for a predefined flight schedule. In addition, the trade-off between performance and computational efficiency of the ILP approach, especially for large-scale systems, will be analyzed in the future.

### References

- Y. Zeinaly, B. De Schutter, and H. Hellendoorn, An integrated model predictive scheme for baggage-handling systems: Routing, line balancing, and emptycart management, IEEE Transactions on Control Systems Technology 23, 1536 (2015).
- [2] Vanderlande Industries, DCV SYSTEMS: Combining Speed, Security and Reliability, Brochure (2013).
- [3] R. De Neufville, The baggage system at Denver: Prospects and lessons, Journal of Air Transport Management 1, 229 (1994).
- [4] M. Johnstone, D. Creighton, and S. Nahavandi, *Status-based routing in bag-gage handling systems: Searching verses learning*, IEEE Transactions on Systems, Man, and Cybernetics, Part C: Applications and Reviews 40, 189 (2010).
- [5] A. Tarău, B. De Schutter, and H. Hellendoorn, *Model-based control for route choice in automated baggage handling systems*, IEEE Transactions on Systems, Man, and Cybernetics, Part C: Applications and Reviews 40, 341 (2010).
- [6] L. Tang and X. Wang, An improved particle swarm optimization algorithm for the hybrid flowshop scheduling to minimize total weighted completion time in process industry, IEEE Transactions on Control Systems Technology 18, 1303 (2010).
- [7] L. Tang and X. Wang, A scatter search algorithm for a multistage production scheduling problem with blocking and semi-continuous batching machine, IEEE Transactions on Control Systems Technology 19, 976 (2011).
- [8] A. Alessandri, C. Cervellera, and M. Gaggero, *Predictive control of container flows in maritime intermodal terminals*, IEEE Transactions on Control Systems Technology 21, 1423 (2013).
- [9] N. Lechevin, C. A. Rabbath, and M. Lauzon, A decision policy for the routing and munitions management of multiformations of unmanned combat vehicles in adversarial urban environments, IEEE Transactions on Control Systems Technology 17, 505 (2009).
- [10] J. M. Maciejowski, Predictive Control with Constraints (Prentice Hall, 2002) p. 352.
- [11] J. B. Rawlings and D. Q. Mayne, *Model Predictive Control Theory and Design*, 1st ed. (Nob Hill Publishing, 2009) p. 533.
- [12] C. E. Garcia, D. M. Prett, and M. Morari, *Model predictive control: Theory and practice–A survey*, Automatica 25, 335 (1989).
- [13] E. Perea-Lopez, B. Ydstie, and I. E. Grossmann, A model predictive control strategy for supply chain optimization, Computers and Chemical Engineering 27, 1201 (2003).

- [14] W. Wang and D. Rivera, Model predictive control for tactical decision-making in semiconductor manufacturing supply chain management, IEEE Transactions on Control Systems Technology 16, 841 (2008).
- [15] A. Alessandri, M. Gaggero, and F. Tonelli, *Min-max and predictive control for the management of distribution in supply chains*, IEEE Transactions on Control Systems Technology **19**, 1075 (2011).
- [16] G. Black and V. Vyatkin, Intelligent component-based automation of baggage handling systems with IEC 61499, IEEE Transactions on Automation Science and Engineering 7, 337 (2010).
- [17] V. Vyatkin, IEC 61499 as enabler of distributed and intelligent automation: State-of-the-art review, IEEE Transactions on Industrial Informatics 7, 768 (2011).
- [18] A. Tarău, B. De Schutter, and H. Hellendoorn, *Hierarchical route choice control for baggage handling systems*, in *Proceedings of the 12th International IEEE Conference on Intelligent Transportation Systems* (2009) pp. 679–684.
- [19] Y. Zeinaly, B. De Schutter, and H. Hellendoorn, An MPC scheme for routing problem in baggage handling systems: A linear programming approach, in Proceedings of the 10th IEEE International Conference on Networking, Sensing and Control (2013) pp. 786–791.
- [20] A. Schrijver, Theory of Linear and Integer Programming (John Wiley & Sons, Inc., 1998) p. 484.
- [21] A. Antoniou and W. S. Lu, *Practical Optimization: Algorithms and Engineering Applications* (Springer, Berlin, Germany, 2007) p. 670.
- [22] T. Weise, Global Optimization Algorithms Theory and Application, 2nd ed. (Self-Published, 2009) p. 1223.

\_ I

\_ I

# 3

# Towards a Robust Multi-level Control Approach for Baggage Handling Systems

"It is my experience that proofs involving matrices can be shortened by 50% if one throws the matrices out."

Emil Artin, 1898 – 1962

This chapter revisits the routing problem in baggage handling systems. We propose a two-level control approach based on a model predictive controller at the top level and a constrained feedback controller at the bottom level that minimizes the  $L_2$  gain of the closed-loop system. The model predictive control problem is recast as a linear programming problem and the constrained feedback controller design problem is formulated as minimization of a linear objective function subject to linear matrix inequalities. The effectiveness of the proposed method is illustrated by a case study.

# **3.1. Introduction**

There has been a growing interest, in the last decade, in automated modern baggage handling systems for large airports. Such baggage handling systems have enabled big airports to achieve high throughput of passengers and cargo. The efficiency and reliability of baggage handling systems have improved over time by

This chapter is based on [1].

implementing more advanced control strategies. However, in order to meet the increasing demand for air travel and cargo shipment, we need more intelligent and reliable control methods than the currently available state-of-the-art methods. Modern baggage handling systems are composed of the following main components: i) loading stations, where the baggage demand originates. The pieces of baggage arrive at the loading stations either from a check-in desk or from a transfer flight, ii) unloading stations that are the final destination of the luggage and from where the pieces of baggage are boarded on to the planes, iii) a network of tracks that connect loading stations to unloading stations through junctions, iv) high-speed destination coded vehicles (DCV) that transport the pieces of baggage on the network from the loading stations to the unloading stations, v) switch controllers at the junctions that determine the route of DCVs. A complete description of the baggage handling system, the state-of-the-art control approaches, and the high-level control problems can be found in [2] and [3]. In this chapter, building on the work of [4], we develop a new approach for dynamic routing of DCVs within the network such that the pieces of baggage arrive at their destination within a given time window with minimum energy consumption. We also improve the robustness of our approach against variations in the baggage demand.

The proposed control structure is composed of a controller based on model predictive control (MPC) at the top level and a controller based on  $L_2$  gain optimization at the bottom level. The MPC controller computes the nominal control input based on nominal prediction of the baggage demand such that the pieces of baggage arrive at their destination within a specified time window with minimal energy consumption. The  $L_2$  based controller then minimizes the deviation of system trajectories from the nominal behavior due to unpredicted variations in the nominal predicted baggage demand. Fig. 3.1 depicts a schematic overview of the proposed two-level control approach.

The rest of the chapter is organized as follows. In Section 3.2, we present the dynamical model of the baggage handling system used for our control purposes. Section 3.3 and Section 3.4 describe the MPC approach and the  $L_2$  optimization control approach, respectively. In Section 3.5, we explain how to combine these two controller approaches into a two-level control structure. In Section 3.6, we present a case study illustrating the performance of our proposed control scheme and finally Section 3.7 concludes the chapter.

# **3.2.** Dynamical Model

The baggage handling system network can be seen as a directed graph G = (V, A), where  $V = O \cup I \cup D$  is the set of nodes composed of origin nodes O (i.e., loading stations), intermediate nodes I (i.e., junctions), and destination nodes D (i.e., unloading stations), and A is the set of arcs composed of links (i.e., tracks) connecting the nodes. The queue lengths are associated with the nodes and the control variables are defined at each node as the flows of DCVs from that node to its neighbor nodes. In a similar manner to the model in [5], the flows are indexed by their destination, enabling us to distinguish between baggage with different destinations. This is important as the baggage must end up in the right destination. Accordingly,



Figure 3.1: Schematic overview of the proposed two-level control approach, where  $u^*$  and  $y^*$  are the nominal input generated by the MPC controller and the resulting output trajectory, respectively,  $\hat{d}$  is the predicted baggage demand,  $\tilde{d}$  is the unpredictable deviation of baggage demand around  $\hat{d}$ , and u and y are respectively the actual input and output of the system.

at each node  $v \in V$  there is a partial queue of DCVs associated with each destination  $d \in D$ . The following assumptions are made in the derivation of the model:

- A1 Each node in the network belongs to at least one directed path from an origin node (i.e., a loading station) to a destination node (i.e., an unloading station).
- A2 A DCV is present at the loading station whenever a piece of baggage arrives.
- A3 The movement of pieces of baggage on the network is approximated by a continuous flow of baggage.
- A4 At each node v, with exception of destination nodes, the DCVs stack up in vertical queues according to their destination. The queue lengths at destination nodes are considered to be zero. This is because we assume either destination nodes have unlimited capacity or there is no restriction on the outflow of destination nodes so the baggage are immediately taken to the planes upon arrival.
- A5 The DCV travel time on each link is an integer multiple of the sampling time  $\Delta t$ .

Assumption A1 guarantees that there are no redundant nodes in the network. By Assumption A2, the pieces of baggage are immediately dispatched from the loading stations as they arrive. Therefore, we do not need to distinguish between baggage flows and DCV flows within the system. Otherwise, we would need to

take into account the movement of empty DCVs from the unloading stations to the loading stations. Assumption A3 is necessary for tractability of the control problem. Although the number of DCVs is an integer in reality, for a fairly large number of DCVs, the movement of DCVs can be approximated by continuous flows. This is not very restrictive as the computed flows can then be realized as well as possible by a lower-level control loop that determines the optimal switching pattern for the switch controllers at the junctions. The actual time required to travel from a node to another one depends on the length of the DCV queue at the end of the link connecting these nodes. However, if the queue lengths are sufficiently small compared to the length of the the links, the variation in the travel time is negligible. This is equivalent to having vertical queues at each node as stated in assumption A4. Assumption A5 allows us to arrive at a linear discrete-time model of the system.

We also make use of the following notation:

- The set of sending nodes of a node  $v \in V$  defined as  $V_v^{\text{send}} = \{w \in V \mid (w, v) \in A\}$ , is the set of nodes that can send flow to node v.
- The set of receiving nodes of a node  $v \in V$  defined as  $V_v^{\text{recv}} = \{w \in V \mid (v, w) \in A\}$ , is the set of nodes that can receive flow from node v.
- The set of all nodes that are on some directed path to a destination node *d* ∈ *D* is *V<sub>d</sub>*.
- For each destination node  $d \in D$  and for each origin node  $v \in O \cap V_d$ ,  $Q_{v,d}(k)$  is the baggage inflow (demand) at v with destination d during the time interval  $[k\Delta t, (k + 1)\Delta t)$ .

For each destination  $d \in D$  and each  $v \in V_d$  and each  $w \in V_v^{\text{recv}} \cap V_d$ , we define the control variable  $q_{v,w,d}(k)$  that is the partial flow of DCVs with destination node d from node v to node w during the time interval  $[k\Delta t, (k + 1)\Delta t)$ . Accordingly,  $x_{v,d}(k)$  denotes the vertical queue length at node v associated with destination d. The set of feasible trajectories of the system is described by the following linear constraints in discrete time:

$$x_{v,d}(k+1) = x_{v,d}(k) + \Delta t(F_{v,d}^{\text{in}}(k) - F_{v,d}^{\text{out}}(k))$$
(3.1a)

$$x_{\nu,d}(k) \ge 0 \tag{3.1b}$$

$$q_{\nu,w,d}(k) \ge 0 \tag{3.1c}$$

where  $F_{v,d}^{\text{in}}(k)$  is the total inflow of DCVs to node v, associated with destination d, given by

$$F_{v,d}^{\text{in}}(k) = \begin{cases} Q_{v,d}(k) + \sum_{\substack{w \in V_v^{\text{send}}}} q_{w,v,d}(k-k_{w,v}) & \text{if } v \in V_d \cap O\\ \sum_{\substack{w \in V_v^{\text{send}}}} q_{w,v,d}(k-k_{w,v}) & \text{if } v \in V_d \cap (D \cup I)\\ 0 & \text{otherwise} \end{cases}$$
(3.2)

with  $k_{w,v}\Delta t$  being the travel time<sup>1</sup> on the link (w, v), and  $F_{v,d}^{\text{out}}(k)$  is the total outflow of DCVs from node v with destination d, given by

$$F_{\nu,d}^{\text{out}}(k) = \begin{cases} F_{\nu,d}^{\text{in}}(k) & \text{if } \nu \in V_d \cap D\\ \sum\limits_{w \in V_\nu^{\text{recv}}} q_{\nu,w,d}(k) & \text{if } \nu \in V_d \cap (O \cup I)\\ 0 & \text{otherwise} \end{cases}$$
(3.3)

Equation (3.1a) describes the evolution of the queue lengths and (3.1b) constrains queue lengths to non-negative values. Likewise, (3.1c) guarantees non-negativity of the control variables (flows).

Let x(k) be the state vector that includes all queue lengths  $x_{v,d}(k)$  and *delayed* samples of  $q_{v,w,d}(k)$  with delay  $\geq 1$ . Let u(k) and d(k) be the control input vector that includes all control variables  $q_{v,w,d}(k)$ , and the demand vector composed of all individual demands  $Q_{v,d}(k)$ , respectively. Then (3.1) can be expressed by a constrained discrete-time linear system as

$$x(k+1) = Ax(k) + B_1 d(k) + B_2 u(k)$$
(3.4a)

$$x(k) \ge 0 \tag{3.4b}$$

$$\boldsymbol{u}(k) \ge 0 \tag{3.4c}$$

with properly defined matrices A,  $B_1$ , and  $B_2$ .

## **3.3. MPC Problem Formulation**

The model presented in Section 3.2 is used as internal prediction model for the MPC approach. At time step k, given the current state of the system and an estimate of future baggage demand, this model is used to compute the trajectories of the system based on which a constrained optimal control problem is solved over a horizon yielding an optimal control sequence. The first element out of the optimal control sequence is then applied to the system according to the receding horizon policy and this process is then repeated at the next time step k + 1 with new measurements [6].

The objective function must reflect the following performance criteria: i) the pieces of baggage assigned to a certain destination (unloading station) must reach the destination within a given time window, ii) the energy consumption of the system should be minimized. The time window represents the time duration in which the end point is ready to receive the luggage. It is undesirable to have the luggage arrive at the destination out of this time window. Indeed, if the pieces of luggage arrive too late, they will miss the flight. Too early arrival of the luggage at the destination point also might inflict a high storage cost on the operator. The energy consumption is associated with manipulating the actuators in the system and wear and tear inflicted on the actuators. There are two contributors to the

<sup>&</sup>lt;sup>1</sup>Assuming a constant speed for DCVs  $v_{DCV}$ ,  $k_{w,v}$  is given by  $k_{w,v} = \frac{s_{w,v}}{\Delta t v_{DCV}}$ , where  $s_{w,v}$  is the length of link (w, v).

energy consumption in the system: i) movements of DCVs in the system, which is related to the magnitude of DCV flows, and ii) variation in the DCV flows. This is particularly important when the DCV flows obtained here will be realized using switch controllers at each junction of the network. The variation in the flow then translates to switching frequency.

In order to achieve the aforementioned control objectives, we consider a cost function that is a weighted combination of four penalty terms that penalize the DCV queue lengths, DCV flows (control variables), and the variation of DCV flows. The cost associated with the DCV is defined as: The constrained linear model given in Section 3.2 cannot be used to determine the time instant at which a certain flow of baggage reaches to its destination explicitly. However, we can consider a cost function to indirectly penalize baggage arrival time deviation from a given time window. The cost function is composed of three penalty terms. The first penalty term penalizes the queue lengths being defined as

$$J_{d}^{\rm tw}(k) = \sum_{v \in V_{d}} C_{v,d}^{\rm tw}(k) x_{v,d}(k)$$
(3.5)

where  $C_{v,d}^{tw}(k)$  as illustrated in Fig. 3.2a is given as

$$C_{v,d}^{\text{tw}}(k) = \begin{cases} 0 & \text{if } k + k_{v,d} \le k_d^{\text{open}} \\ c^{\text{tw}}(k - k_d^{\text{open}} + k_{v,d}) & \text{if } k_d^{\text{open}} < k + k_{v,d} \le k_d^{\text{close}} \\ c^{\text{tw}}(k_d^{\text{close}} - k_d^{\text{open}}) & \text{if } k + k_{v,d} > k_d^{\text{close}} \end{cases}$$
(3.6)

where  $k_d^{\text{open}}$  and  $k_d^{\text{close}}$  are, respectively, the opening and the closing time steps of destination d and  $k_{v,d}\Delta t$  is the expected travel time from node v to destination d under the current nominal operating conditions<sup>2</sup>. Note that since  $C_{v,d}^{\text{tw}}(k) = 0$  for  $k \leq k_d^{\text{open}} - k_{v,d}$ , the queue lengths associated with destination d are not penalized before the destination is open, taking into account the DCVs travel time from v to d. During the time window of destination d, the weight associated with DCV queues increases linearly in time, hence, forcing the DCVs to move towards d.

The penalty term associated with the DCV flows is defined as:

$$J_d^{\text{flow}}(k) = \sum_{\nu \in V_d} \sum_{w \in V_\nu^{\text{flow}} \cap V_d} C_{\nu,d}^{\text{flow}}(k) q_{\nu,w,d}(k)$$
(3.7)

with  $C_{\nu,d}^{\text{flow}}(k)$  as depicted in Fig. 3.2b being

$$C_{\nu,d}^{\text{flow}}(k) = \begin{cases} -c_1^{\text{flow}}(k - k_d^{\text{open}} + k_{\nu,d}) & \text{if } k + k_{\nu,d} \le k_d^{\text{open}} \\ 0 & \text{if } k_d^{\text{open}} < k + k_{\nu,d} \le k_d^{\text{close}} \\ c_2^{\text{flow}}(k - k_d^{\text{close}} + k_{\nu,d}) & \text{if } k + k_{\nu,d} > k_d^{\text{close}} \end{cases}$$
(3.8)

Note that  $C_{v,d}^{\text{flow}}(k)$  is chosen in such a way that DCV flows to destination d are allowed during the time window of d. Higher values  $C_{v,d}^{\text{flow}}(k)$  outside of the time

<sup>&</sup>lt;sup>2</sup>These can be obtained based on historical data for periods with similar conditions as the current one.

window prevent early or late DCV flows to the destination *d*. Moreover, in order to allow late DCVs to reach the destination, the slope of the third part of  $C_{v,d}^{\text{flow}}(k)$  is smaller than the slope of the first part. Now will will introduce the terms in the cost function that reflect the energy consumption in the network. We penalize all flows in the network in order to avoid indefinite circulation of DCVs throughout the network. Hence, we consider the following penalty term:

$$J^{e}(k) = \sum_{d \in D} \sum_{v \in V_{d}} \sum_{w \in V_{v}^{\text{recv}} \cap V_{d}} q_{v,w,d}(k)$$
(3.9)

In addition, we use the following penalty term to penalize the total variation of the control signal (i.e, flows), which reflects the wear and tear of the DCVs:

$$J^{\rm sw}(k) = \sum_{d \in D} \sum_{l \in V_d} \sum_{w \in V_v^{\rm recv} \cap V_d} |q_{v,w,d}(k) - q_{v,w,d}(k-1)|$$
(3.10)

The total cost at time step k is therefore given as

$$J(k) = \sum_{d \in D} J_d^{\text{tw}}(k) + \alpha_1 \sum_{d \in D} J_d^{\text{flow}}(k) + \alpha_2 J^{\text{e}}(k) + \alpha_3 J^{\text{sw}}(k)$$
(3.11)

where  $\alpha_i > 0$  is a weight factor indicating the relative importance of the associated term in the objective function. The MPC performance index over the prediction horizon of  $N_{\rm p}$  step is thus given as

$$J(k, N_{\rm p}) = \sum_{i=k}^{k+N_{\rm p}-1} J(i)$$
(3.12)

Now we would like to highlight the following remarks:

- R1 The plots of Fig. 3.2a and Fig. 3.2b show respectively coefficients of the penalty terms (3.5) and (3.7), not the penalty terms themselves. In fact, at the given time step k and for a prediction horizon  $N_p$  the values of these coefficients are known for  $k, ..., k + N_p 1$ . Therefore, these coefficients have fixed values and hence the associated penalty terms (3.5) and (3.7) are linear in the control variable.
- R2 By introducing some dummy variables according to standard techniques in optimization [7], terms of the form (3.10) can be recast as a linear programming problem with linear constraints.

Consider u(k), x(k), and d(k) as introduced in Section 3.2. At every time step k we solve the following optimization problem:

$$\min_{\overline{\boldsymbol{u}}(k)} F(k)\overline{\boldsymbol{u}}(k)$$
subject to  $A_{\text{ineq}}(k)\overline{\boldsymbol{u}}(k) \le \boldsymbol{b}_{\text{ineq}}(k)$ 

$$A_{\text{eq}}(k)\overline{\boldsymbol{u}}(k) = \boldsymbol{b}_{\text{eq}}(k)$$
(3.13)

where the vector F(k) is defined based on the MPC objective function (3.11), as explained in Appendix 3.A, and the vector  $\overline{u}(k)$  includes the control inputs  $u(k), ..., u(k+N_p-1)$  and the dummy variables mentioned in Remark R2. Moreover,  $A_{\text{ineq}}(k)$ , and  $A_{\text{eq}}(k)$  are determined based on the constraints, and  $b_{\text{ineq}}(k)$ , and  $b_{\text{eq}}(k)$  are constant vectors that depend on the current state x(k) and the demand values  $d(k), ..., d(k+N_p-1)$ .

The optimization problem given by (3.13) is an LP problem, that can be solved efficiently with currently available solvers, e.g., MATLAB linprog.



Figure 3.2: (a) the coefficient for the queue length penalty term; (b) the coefficient for the flow penalty term.

## **3.4. Feedback Control Problem Formulation**

#### **3.4.1. Problem Setup**

Consider a discrete-time linear system

$$x(k+1) = Ax(k) + B_1 d(k) + B_2 u(k)$$
(3.14a)

$$\mathbf{z}(k) = C_1 \mathbf{x}(k) + D_1 \mathbf{d}(k) + D_2 \mathbf{u}(k)$$
(3.14b)

with full state feedback

$$\boldsymbol{u}(k) = K\boldsymbol{x}(k) \tag{3.15}$$

where the system matrices,  $x \in \mathbb{R}^n$ ,  $d \in \mathbb{R}^{n_d}$ , and  $u \in \mathbb{R}^m$  are those of (3.4a) and  $z \in \mathbb{R}^{n_z}$  is the controlled output vector. Assume that  $(A, B_2)$  is stabilizable and K is a stabilizing feedback gain. The  $L_2$  gain of the closed-loop system is bounded by  $\gamma > 0$  (i.e.,  $\sup_{z \neq 0} \frac{\|z\|_2}{\|d\|_2} \le \gamma$ ) if and only if there exists a P > 0 such that [8], [9]

$$\begin{bmatrix} \mathcal{A}^{\mathrm{T}}P\mathcal{A} - P + \frac{1}{\gamma}\mathcal{C}^{\mathrm{T}}\mathcal{C} & \mathcal{A}^{\mathrm{T}}P\mathcal{B} + \frac{1}{\gamma}\mathcal{C}^{\mathrm{T}}\mathcal{D} \\ \mathcal{B}^{\mathrm{T}}P\mathcal{A} + \frac{1}{\gamma}\mathcal{D}^{\mathrm{T}}\mathcal{C} & \mathcal{B}^{\mathrm{T}}P\mathcal{B} + \frac{1}{\gamma}\mathcal{D}^{\mathrm{T}}\mathcal{D} - \gamma I \end{bmatrix} \leq 0$$
(3.16)

where

$$\begin{bmatrix} \mathcal{A} & \mathcal{B} \\ \mathcal{C} & \mathcal{D} \end{bmatrix} = \begin{bmatrix} A + B_2 K & B_1 \\ C_1 + D_2 K & D_1 \end{bmatrix}$$
(3.17)

or equivalently

$$\begin{bmatrix} -Q & \mathcal{A}Q & \mathcal{B} & 0\\ Q\mathcal{A}^{\mathrm{T}} & -Q & 0 & Q\mathcal{C}^{\mathrm{T}}\\ \mathcal{B}^{\mathrm{T}} & 0 & -\gamma I & \mathcal{D}^{\mathrm{T}}\\ 0 & \mathcal{C}Q & \mathcal{D} & -\gamma I \end{bmatrix} \leq 0$$
(3.18)

with  $Q = P^{-1} > 0$ .

Consider the problem of determining a feedback gain *K* that minimizes the  $L_2$  gain of the closed-loop system. It is well-known [9] that with the transformation Y = KQ, the matrix inequality of (3.18) can be written as

$$\begin{bmatrix} -Q & AQ + B_2Y & B_1 & 0\\ QA^{\mathrm{T}} + Y^{\mathrm{T}}B_2^{\mathrm{T}} & -Q & 0 & QC_1^{\mathrm{T}} + Y^{\mathrm{T}}D_2^{\mathrm{T}}\\ B_1^{\mathrm{T}} & 0 & -\gamma I & D_1^{\mathrm{T}}\\ 0 & C_1Q + D_2Y & D_1 & -\gamma I \end{bmatrix} \le 0$$
(3.19)

with Q > 0.

Note that for the closed-loop system given by (3.14) and (3.15), (3.16) implies

$$\boldsymbol{x}^{\mathrm{T}}(k+1)\boldsymbol{P}\boldsymbol{x}(k+1) - \boldsymbol{x}^{\mathrm{T}}(k)\boldsymbol{P}\boldsymbol{x}(k) + \frac{1}{\gamma}\boldsymbol{z}^{\mathrm{T}}(\boldsymbol{k})\boldsymbol{z}(\boldsymbol{k}) \leq \gamma \boldsymbol{d}^{\mathrm{T}}(\boldsymbol{k})\boldsymbol{d}(\boldsymbol{k})$$
(3.20)

Now we define the ellipsoid  $\varepsilon_{\gamma} := \{ x | x^{T} \frac{p}{\gamma} x \le 1 \}$ . Assuming x(0) = 0, (3.20) yields

$$\boldsymbol{x}^{\mathrm{T}}(T)P\boldsymbol{x}(T) \leq \gamma \sum_{k=0}^{T-1} \boldsymbol{d}^{\mathrm{T}}(k)\boldsymbol{d}(k) < \gamma \sum_{k=0}^{\infty} \boldsymbol{d}^{\mathrm{T}}(k)\boldsymbol{d}(k)$$
(3.21)

for any  $T \in \mathbb{N}$ . Assuming<sup>3</sup>  $||\boldsymbol{d}||_2^2 = 1$ , we get

$$\boldsymbol{x}^{\mathrm{T}}(T) \frac{P}{\gamma} \boldsymbol{x}(T) < 1 \tag{3.22}$$

which shows that  $x(T) \in \varepsilon_{\gamma}$ . Since (3.21) holds for all T,  $\varepsilon_{\gamma}$  contains the set of states that are reachable by a unit energy input signal d when the  $L_2$  gain of the closed-loop system is bounded by  $\gamma$ .

#### **3.4.2. Hard State Constraints**

Now we consider the problem of searching for the feedback gain K that minimizes the  $L_2$  gain of the closed-loop system subject to polytopic state constraints of the form

$$a_i^{\mathrm{T}} x(k) \le 1, \ i = 1, \dots, r.$$
 (3.23)

To include the state constraints of (3.23), consider the polytope

$$\mathcal{P} = \{ \boldsymbol{x} \in \mathbb{R}^n | \boldsymbol{a}_i^{\mathrm{T}} \boldsymbol{x} \le 1, i = 1, \dots, r \}$$
(3.24)

<sup>&</sup>lt;sup>3</sup>It is always possible to scale *d* such that  $||d||_2^2 = 1$ .

associated with (3.23). We assume that  $\mathcal{P}$  has the origin in its interior. To guaranty that (3.23) holds for all k > 0 with x(0) = 0, we must have  $\varepsilon_{\gamma} \subseteq \mathcal{P}$  or equivalently [9]

$$\boldsymbol{a}_{i}^{\mathrm{T}} \gamma Q \boldsymbol{a}_{i} \leq 1, \ i = 1, \dots, r \tag{3.25}$$

Therefore, the following optimization problem needs to be solved:

$$\min_{Q,Y,\gamma} \gamma \text{ subject to (3.19), (3.25), } Q > 0$$
 (3.26)

This problem is not jointly convex in  $\gamma$  and Q and Y. Moreover, it can be shown in a straightforward manner that the constraints of (3.18) and (3.25) do not satisfy the monotonicity property  $G(Q, Y, \gamma_1) > G(Q, Y, \gamma_2)$  if  $\gamma_1 > \gamma_2$ , where G < 0represents constraints (3.19) and (3.25) combined. Therefore, this problem cannot even be recast as a generalized eigenvalue problem, which is a class of quasiconvex optimization problems [9].

Now we will replace constraint (3.25) by a more conservative one that is convex in the optimization variables, in the following manner. Note that

$$\gamma Q = \frac{1}{4} (\gamma I + Q)^{\mathrm{T}} (\gamma I + Q) - \frac{1}{4} (\gamma I - Q)^{\mathrm{T}} (\gamma I - Q)$$
(3.27)

Obviously,  $\gamma Q < \frac{1}{4}(\gamma I + Q)^{\mathrm{T}}(\gamma I + Q)$ . Hence,

$$\frac{1}{4}\boldsymbol{a}_{i}^{\mathrm{T}}(\gamma l+Q)^{\mathrm{T}}(\gamma l+Q)\boldsymbol{a}_{i} \leq 1 \implies \boldsymbol{a}_{i}^{\mathrm{T}}\gamma Q\boldsymbol{a}_{i} < 1$$
(3.28)

or equivalently expressed using the Schur complement

$$\begin{bmatrix} I & (\gamma I + Q)\boldsymbol{a}_i \\ \boldsymbol{a}_i^{\mathrm{T}}(\gamma I + Q) & 4 \end{bmatrix} > 0$$
(3.29)

Clearly, this introduces conservatism as the feasibility set of (3.29) is a subset of the feasibility set of (3.25). This conservatism can be reduced if one can find a lower bound for  $(\gamma I - Q)^{T}(\gamma I - Q) \ge 0$  such that  $(\gamma I - Q)^{T}(\gamma I - Q) \ge \alpha^{2}I$  or equivalently  $\|\gamma I - Q\| \ge \alpha$  (in matrix norm sense) for some  $\alpha > 0$ . Then, instead of (3.29), one obtains

$$\begin{bmatrix} I & (\gamma I + Q)\boldsymbol{a}_i \\ \boldsymbol{a}_i^{\mathrm{T}}(\gamma I + Q) & 4 + \boldsymbol{a}_i^{\mathrm{T}}\alpha^2 \boldsymbol{a}_i \end{bmatrix} > 0$$
(3.30)

Therefore we consider (3.26) with (3.25) replaced by (3.29) or by (3.30). This is an eigenvalue problem [9], which is a convex optimization problem that can be solved with currently available LMI optimization toolboxes, e.g., MATLAB LMI toolbox, YALMIP [10], and CVX [11], [12].

#### **3.4.3. Soft State Constraints**

In the view of the proposed two-level control scheme, it makes more sense to replace the hard constraints of (3.25) by soft constraints due to the following observations: i) the constraints are mainly handled at the top level by the MPC controller, ii) if the constraints are too restrictive the conservative version of the original

constraints as expressed by (3.29) may become infeasible, which is not desirable. As an alternative to the approach presented in Section 3.4.2, one can replace hard constraints by soft ones by considering a multi-objective optimization approach that penalizes the  $L_2$  gain of the closed-loop system and, indirectly, the constraint violation at the same time. More precisely, we define the following optimization problem with the objective function that penalizes  $\gamma$ , and the volume of the ellipsoid  $\varepsilon_{\gamma}$ , which is proportional to  $\sqrt{\det \gamma Q}$ :

$$\min_{Q,Y,\gamma} c_{\gamma}\gamma + \log(\det(\gamma Q)) \text{ subject to } Q > 0 \text{ and } (3.19)$$
(3.31)

where  $c_{\gamma} > 0$  is a weight factor. The magnitude of  $c_{\gamma}$  determines the trade-off between the  $L_2$  gain and the volume of the ellipsoid that represents the set of reachable states. By minimizing the volume of  $\varepsilon_{\gamma}$ , we confine the set of reachable state from the origin. This indirectly minimizes constraint violation since the origin lies in the interior of polytope  $\mathcal{P}$ . However, this objective function is not convex in the optimization variables  $\gamma$  and Q. To mitigate this problem, instead of penalizing the volume of  $\varepsilon_{\gamma}$ , we penalize an upper bound on the length of semi-major axis of  $\varepsilon_{\gamma}$ , which is  $\sqrt{\lambda_{\max}(\gamma Q)}$ , where  $\lambda_{\max}(\gamma Q)$  is the largest eigenvalue of  $\gamma Q$ . It is clear from (3.27), that  $\lambda_{\max}(\frac{1}{4}(\gamma I + Q)^{\mathrm{T}}(\gamma I + Q))$  constitutes an upper bound on  $\lambda_{\max}(\gamma Q)$ . Then we get

$$\min_{Q,Y,\gamma} c_{\gamma}\gamma + \lambda_{\max} \left(\frac{1}{4}(\gamma I + Q)^{\mathrm{T}}(\gamma I + Q)\right) \text{ subject to } Q > 0 \text{ and } (3.19)$$
(3.32)

or equivalently

$$\min_{Q,Y,\gamma,\lambda} c_{\gamma}\gamma + \lambda \text{ subject to (3.19), } Q > 0, \begin{bmatrix} \lambda I & \gamma I + Q \\ \gamma I + Q & 4I \end{bmatrix} > 0$$
(3.33)

This is an eigenvalue problem [9] that can be solved efficiently with currently available LMI solvers such as MATLAB LMI toolbox. Note that, by inspecting (3.27), the upper bound on  $\lambda_{\max}(\gamma Q)$  can be made tighter if one can find an  $\alpha > 0$  such that  $(\gamma I - Q)^{T}(\gamma I - Q) \ge \alpha^{2}I$  or equivalently, in matrix norm<sup>4</sup> sense,  $||\gamma I - Q|| \ge \alpha$ . Then the last constraint in (3.33) will be replaced by

$$\begin{bmatrix} (\lambda + \frac{\alpha^2}{4})I & \gamma I + Q\\ \gamma I + Q & 4I \end{bmatrix} > 0$$
(3.34)

<sup>&</sup>lt;sup>4</sup>For matrix norm, we use the definition  $||A|| = \sigma_{\max}(A)$ , where  $\sigma_{\max}(A)$  is the largest singular value of matrix A.

As an example, consider a discrete-time linear system given by

$$A = \begin{bmatrix} 0.1514 & 0.4377 & 0.7293 & 0.1839 \\ 0.3958 & 0.3999 & 0.7521 & 0.9368 \\ 0.9720 & 0.7636 & 0.8323 & 0.6137 \\ 0.7718 & 0.8639 & 0.4821 & 0.6050 \end{bmatrix}, C_1 = I$$
$$B_1 = \begin{bmatrix} 0 \\ 0 \\ -0.5881 \\ 0.2487 \end{bmatrix}, B_2 = \begin{bmatrix} 0.6954 \\ -0.2837 \\ -0.9723 \\ 0.6086 \end{bmatrix}, D_1 = 0, D_2 = 0$$

For  $c_{\gamma}$  taking values in the interval [0.1, 100], Fig. 3.3 illustrates the trade-off between minimizing the  $L_2$  gain and the length of the semi-major axis of  $\varepsilon_{\gamma}$ .



Figure 3.3: Trade-off curve between the optimal  $\gamma$  and the length of the semi-major axis of  $\varepsilon_{\gamma}$ .

## **3.5. Integration of MPC and Feedback Controllers**

In this section, we briefly explain how the two control schemes presented in Sections 3.3 and 3.4 can be combined. The baggage demand at each origin node is composed of a base demand  $d^*$ , which is assumed to be predictable over the prediction horizon  $N_{\rm p}$ , and a small additive perturbation  $\tilde{d}$  around the base demand that cannot be predicted. Based on a future prediction of  $d^*$ , the MPC controller computes the optimal DCV flows  $u^*$  and system trajectories  $z^*$  subject to flow and queue length constraints such that the DCVs arrive at their destinations with minimal energy consumption and with minimal deviation from the time windows. To minimize the adverse effect of  $\tilde{d}$  on optimal system trajectories computed by the MPC controller, a feedback gain K minimizing  $\frac{\|\tilde{z}\|_2}{\|\tilde{d}\|_2} = \frac{\|z-z^*\|_2}{\|d-d^*\|_2}$  based on the measurement  $y(k) - y^*(k)$  is implemented along the MPC controller in the configuration

depicted in Fig. 3.1. Therefore, the control law applied to the system at time step k is  $u(k) = u^*(k) + K(y(k) - y^*(k)) = u^*(k) + K\tilde{y}(k)$ .

When we impose constraints on the controlled output  $\mathbf{z}(k) = \mathbf{z}^*(k) + \tilde{\mathbf{z}}(k)$ , the constraints on  $\tilde{\mathbf{z}}(k)$  depend on value of  $\mathbf{z}^*(k)$ . As a result, one needs to update the feedback gain *K* whenever the value of  $\mathbf{z}^*(k)$  changes. This can be avoided if soft constraints as in Section 3.4.3 are used. Moreover, the MPC control law  $\mathbf{u}^*$  does not have to be updated at every time instant  $k\Delta t$ . Particularly, if the base demand  $d^*$  is varying slowly with time, one can use a controller sampling time  $m\Delta t$ , with m > 1 being an integer number.

# 3.6. Case Study



Figure 3.4: A layout of baggage handling system with one loading station and one unloading station. The length of each link in the network is 40 m.



Figure 3.5: Base baggage demand, the perturbations on the base demand, and the actual demand at the loading station.

In this section we present a case study to illustrate the performance of our proposed control approach for the baggage handling system. For the sake of simplicity, we consider a simple baggage handling system, the layout of which is depicted in Fig. 3.4. Here, the focus is to illustrate the effect of the feedback controller on sup-



Figure 3.6: Optimal flows of DCVs at the unloading station. One can observe that most of the DCVs arrive at the unloading station within the specified time window.

| MPC Parameters                    |                   |                           |  |   |  |  |  |
|-----------------------------------|-------------------|---------------------------|--|---|--|--|--|
| Np                                | time window       | u <sub>max</sub> [DCV/s]  | $(\alpha_1, \alpha_2, \alpha_3)$                   | $(c_1^{\text{flow}}, c_2^{\text{flow}}, c^{\text{tw}})$ |  |  |  |
| 12                                | [40, 70]          | 4                         | (1, 125, 1)  | (100, 0.2, 1)   |  |  |  |
| Feedback Controller Parameters    |                   |                           |  |   |  |  |  |
| cγ                                | $\gamma^*_{\min}$ | $\gamma_{ m min}$         | $\lambda^*_{ m max}$                               | $\lambda_{\max}$  |  |  |  |
| 40                                | 1.4420            | 1.5222                    | 2.0001   | 3.8091  |  |  |  |
| Closed-loop Simulation Parameters |                   |                           |  |   |  |  |  |
| $\Delta t[s]$                     | N <sub>sim</sub>  | $x_0$ (initial condition) | demand perturbation                                | $v_{\rm DCV}[{\rm m/s}]$                                |  |  |  |
| 1.41                              | 100               | 10                        | $\tilde{d} \in \mathcal{U}(0,1), \ d\ _2^2 = 4.02$ | 1.41  |  |  |  |

Table 3.1: Controller design and simulation parameters

pressing the adverse effects of an unpredicted baggage demand on the behavior of the system. First, assuming that the demand is fully known, the optimal flows and optimal system trajectories are computed. Next, we consider some unpredictable random perturbations on the base demand and evaluate how closely our proposed two-level control approach can follow the optimal trajectory. For the two-level control approach, we have computed the feedback gain *K* based the approach of Section 3.4.3 using the MATLAB LMI toolbox. Table 3.1 lists the parameters used for the controller design and the closed-loop simulation. In Table 3.1,  $\lambda_{\max}^*$  and  $\gamma_{\min}^*$  denote, respectively, the *actual* values of  $\lambda_{\max}(\gamma Q)$  and  $\gamma$  achieved by the closed-loop system whereas  $\lambda_{\max}^*$  and  $\gamma_{\min}^*$  denote those values obtained by solving (3.33).

For the base demand  $d^*(k)$  depicted in Fig. 3.5, the optimal flows to the destination (node 5) are illustrated in Fig. 3.6 and the resulting optimal queue length at the origin node (node 1) is depicted in Fig. 3.7. It is clear from Fig. 3.6 that the optimal flows arrive at the destination within the desired time window. The perturbation on the base demand  $\tilde{d}(k) \in \mathcal{U}(0, 1)$  is depicted in Fig. 3.5. It is obvious



Figure 3.7: Queue lengths at node 1. One can observe that for the two-level control approach, the queue length at node 1 is only slightly affected by the disturbance.

from Fig. 3.7 that in the presence of unpredictable demand perturbations, the twolevel controller follows the optimal trajectory very closely whereas the MPC based approach deviates from the optimal trajectory.

# **3.7.** Conclusions and Future Work

The routing problem in baggage handling systems was revisited. A flow-based model was derived for our control purposes, which are delivering the pieces of baggage at the unloading stations within a pre-specified time window, and minimizing the energy consumption. We proposed a multi-level control approach with an MPC controller at the top level and a constrained feedback controller at the bottom level that minimizes the  $L_2$  gain of the closed-loop system. The idea was that based on some prior knowledge on the baggage demand, the MPC controller computes the optimal control inputs and system trajectories such that the pieces of baggage arrive at their destination within a desired time window and with minimal energy consumption. The feedback controller then would guarantee minimal deviation from this optimal trajectory in face of unknown perturbations on the baggage demand.

We showed that the MPC problem can be formulated as a linear programming problem. We proposed two methods to include state constraints in design procedure of the feedback controller that can be recast as LMI constraints. Using a simple case study, we showed the effectiveness of the proposed two-level control approach.

This approach should be extendable to large-scale system. However, for large-scale systems, the conservatism introduced by (3.28) may render the LMIs in (3.33) infeasible. Hence, one may need to find a tighter lower bound  $\alpha$  in (3.34).

For future work, the scalability of the proposed two-level approach to large network layouts will be investigated. In addition, we will compare the performance of the two-level control approach with the MPC-based approach for larger network layouts and more elaborate scenarios. As a second extension to the current work, we will include non-polytopic state constraints as well as control signal constraints in the design procedure of the feedback controller.

\_\_\_\_\_

# References

- Y. Zeinaly, B. De Schutter, and H. Hellendoorn, *Towards a robust multi-level* control approach for baggage handling systems, in 52nd IEEE Conference on Decision and Control (2013) pp. 281–287.
- [2] A. Tarău, B. De Schutter, and H. Hellendoorn, *Model-based control for route choice in automated baggage handling systems*, IEEE Transactions on Systems, Man, and Cybernetics, Part C: Applications and Reviews 40, 341 (2010).
- [3] Y. Zeinaly, B. De Schutter, and H. Hellendoorn, A model predictive control approach for the line balancing in baggage handling systems, IFAC Proceedings Volumes 45, 215 (2012), 13th IFAC Symposium on Control in Transportation Systems.
- [4] Y. Zeinaly, B. De Schutter, and H. Hellendoorn, An MPC scheme for routing problem in baggage handling systems: A linear programming approach, in Proceedings of the 10th IEEE International Conference on Networking, Sensing and Control (2013) pp. 786–791.
- [5] A. Tarău, B. De Schutter, and H. Hellendoorn, *Hierarchical route choice control for baggage handling systems*, in *Proceedings of the 12th International IEEE Conference on Intelligent Transportation Systems* (2009) pp. 679–684.
- [6] C. E. Garcia, D. M. Prett, and M. Morari, *Model predictive control: Theory and practice–A survey*, Automatica 25, 335 (1989).
- [7] A. Antoniou and W. S. Lu, Practical Optimization: Algorithms and Engineering Applications (Springer, Berlin, Germany, 2007) p. 670.
- [8] C. E. de Souza and L. Xie, On the discrete-time bounded real lemma with application in the characterization of static state feedback H<sub>∞</sub> controllers, Systems & Control Letters **18**, 61 (1992).
- [9] S. Boyd, L. El Ghaoui, E. Feron, and V. Balakrishnan, *Linear Matrix Inequalities in System and Control Theory* (Society for Industrial and Applied Mathematics, 1994) p. 185.
- [10] J. Lofberg, Yalmip: A toolbox for modeling and optimization in matlab, in Proceedings of IEEE International Symposium on Computer Aided Control Systems Design (2004) pp. 284–289.
- [11] CVX Research, Inc., CVX: Matlab software for disciplined convex programming, version 2.0 beta, http://cvxr.com/cvx (2012).
- [12] M. C. Grant and S. P. Boyd, Graph implementations for nonsmooth convex programs, in Recent Advances in Learning and Control, Lecture Notes in Control and Information Sciences, Vol. 371, edited by V. D. Blondel, S. P. Boyd, and K. Hidenori (Springer London, 2008) pp. 95–110.

# Appendix

# **3.A.** Derivation of Linear Program

In this section we show how the MPC problem of linear system with a linear objective function can be formulated as a standard linear program. Consider the discrete-time linear time-invariant system

$$\begin{aligned} \mathbf{x}(t+1) &= \mathbf{A}\mathbf{x}(t) + \mathbf{B}_1 \boldsymbol{\omega}(t) + \mathbf{B}_2 \boldsymbol{u}(t), \quad (3.35) \\ \mathbf{y}(t) &= \mathbf{C}\mathbf{x}(t), \\ \mathbf{A} \in \mathbb{R}^{n \times n}, \ \mathbf{B}_1 \in \mathbb{R}^{n \times n_\omega}, \ \mathbf{B}_2 \in \mathbb{R}^{n \times n_u}, \ \mathbf{C} \in \mathbb{R}^{n_y \times n}, \end{aligned}$$

where  $x(t) \in \mathbb{R}^n$ ,  $\omega(t) \in \mathbb{R}^{n_\omega}$ ,  $u(t) \in \mathbb{R}^{n_u}$ ,  $y(t) \in \mathbb{R}^{n_y}$  are respectively the state at time *t*, the disturbance input at time *t*, the control input at time *t*, and the output at time *t*. Consider the constrained linear optimization problem defined at the current time *t* as

$$\min_{u(t+k),u(t+k+1),\dots,u(t+k+N-1)} \sum_{k=1}^{N-1} (q^{\mathrm{T}} y(t+k) + r^{\mathrm{T}} u(t+k))$$
(3.36)  
+  $s^{\mathrm{T}} y(t+N)$ 

subject to:

recursion (3.35) for 
$$k = 1, ..., N$$
 with initial state  $x(t)$ ,  
 $Gy(t + k) + g \le 0, k = 1, ..., N$ ,  
 $Mu(t + k) + m \le 0, k = 1, ..., N - 1$ ,

where  $q \in \mathbb{R}^{n_y}$ ,  $r \in \mathbb{R}^{n_u}$ , and  $s \in \mathbb{R}^{n_y}$  are weighting factors, and where  $G \in \mathbb{R}^{n_1 \times n_y}$ ,  $g \in \mathbb{R}^{n_1}$ ,  $M \in \mathbb{R}^{n_2 \times n_u}$ ,  $m \in \mathbb{R}^{n_2}$  define the output and control constraints, respectively. This problem can be written in the standard LP form using the explicit expression of the solution of (3.35), which is given as

$$\mathbf{x}(t+k) = \mathbf{A}^{k}\mathbf{x}(t) + \sum_{s=0}^{k-1} \mathbf{A}^{s} \Big( \mathbf{B}_{2}\mathbf{u}(t+k-1-s) + \mathbf{B}_{1}\boldsymbol{\omega}(t+k-1-s) \Big), \quad (3.37)$$

for all  $k \in \mathbb{N}$ . Define

$$\boldsymbol{x}_{N}(t) = \begin{bmatrix} \boldsymbol{x}(t+1) \\ \vdots \\ \boldsymbol{x}(t+N) \end{bmatrix}, \ \boldsymbol{y}_{N}(t) = \begin{bmatrix} \boldsymbol{y}(t+1) \\ \vdots \\ \boldsymbol{y}(t+N) \end{bmatrix}, \ \boldsymbol{u}_{N}(t) = \begin{bmatrix} \boldsymbol{u}(t) \\ \vdots \\ \boldsymbol{u}(t+N-1) \end{bmatrix},$$
$$\boldsymbol{\omega}_{N}(t) = \begin{bmatrix} \boldsymbol{\omega}(t) \\ \vdots \\ \boldsymbol{\omega}(t+N-1) \end{bmatrix}, \ \boldsymbol{A}_{N} = \begin{bmatrix} \boldsymbol{A} \\ \vdots \\ \boldsymbol{A}^{N} \end{bmatrix}, \ \boldsymbol{q}_{N} = \begin{bmatrix} \boldsymbol{q} \\ \vdots \\ \boldsymbol{q} \end{bmatrix}, \ \boldsymbol{C}_{N} = \begin{bmatrix} \boldsymbol{C} \\ \vdots \\ \boldsymbol{C} \end{bmatrix},$$

$$\boldsymbol{r}_{N} = \begin{bmatrix} \boldsymbol{r} \\ \vdots \\ \boldsymbol{r} \end{bmatrix}, \ \boldsymbol{G}_{N} = \begin{bmatrix} \boldsymbol{G} \\ \vdots \\ \boldsymbol{G} \end{bmatrix}, \ \boldsymbol{g}_{N} = \begin{bmatrix} \boldsymbol{g} \\ \vdots \\ \boldsymbol{g} \end{bmatrix}, \ \boldsymbol{M}_{N} = \begin{bmatrix} \boldsymbol{M} \\ \vdots \\ \boldsymbol{M} \end{bmatrix}, \ \boldsymbol{m}_{N} = \begin{bmatrix} \boldsymbol{m} \\ \vdots \\ \boldsymbol{m} \end{bmatrix},$$
$$\boldsymbol{B}_{2,N} = \begin{bmatrix} \boldsymbol{B}_{2} & 0 & \cdots & 0 \\ \boldsymbol{A}\boldsymbol{B}_{2} & \boldsymbol{B}_{2} & \cdots & 0 \\ \vdots & \cdots & \ddots & 0 \\ \boldsymbol{A}^{N-1}\boldsymbol{B}_{2} & \boldsymbol{A}^{N-2}\boldsymbol{B}_{2} & \cdots & \boldsymbol{B}_{2} \end{bmatrix}, \ \boldsymbol{B}_{1,N} = \begin{bmatrix} \boldsymbol{B}_{1} & 0 & \cdots & 0 \\ \boldsymbol{A}\boldsymbol{B}_{1} & \boldsymbol{B}_{1} & \cdots & 0 \\ \vdots & \cdots & \ddots & 0 \\ \boldsymbol{A}^{N-1}\boldsymbol{B}_{1} & \boldsymbol{A}^{N-2}\boldsymbol{B}_{1} & \cdots & \boldsymbol{B}_{1} \end{bmatrix}.$$

Then, using

$$\boldsymbol{y}_N = \boldsymbol{C}_N \Big( \boldsymbol{A}_N \boldsymbol{x}(t) + \boldsymbol{B}_{2,N} \boldsymbol{u}_N(t) + \boldsymbol{B}_{1,N} \boldsymbol{\omega}_N(t) \Big),$$

the optimization problem (3.36) can be rewritten as

$$\min_{\boldsymbol{u}_N} \begin{bmatrix} \boldsymbol{q}_N^{\mathrm{T}} & \boldsymbol{s}^{\mathrm{T}} \end{bmatrix} \boldsymbol{C}_N \Big( \boldsymbol{A}_N \boldsymbol{x}(t) + \boldsymbol{B}_{1,N} \boldsymbol{\omega}_N(t) + \boldsymbol{B}_{2,N} \boldsymbol{u}_N(t) \Big)$$

subject to:

$$G_N C_N B_{2,N} u_N + G_N C_N (A_N x(t) + B_{1,N} \omega_N) \leq g_N$$
$$M_N u_N \leq m_N,$$

or equivalently as

$$\min_{\boldsymbol{u}_N} \begin{bmatrix} \boldsymbol{q}_N^{\mathrm{T}} & \boldsymbol{s}^{\mathrm{T}} \end{bmatrix} \boldsymbol{B}_{2,N} \boldsymbol{u}_N(t)$$
subject to:

$$G_N C_N B_{2,N} u_N + G_N C_N \Big( A_N x(t) + B_{1,N} \omega_N \Big) \le g_N$$
$$M_N u_N \le m_N,$$

which is in the form of standard linear program in the variable  $\boldsymbol{u}_{\scriptscriptstyle N}.$ 

# 4

# Robustly Positively Invariant Sets for Discrete-time Linear Positive Systems: Application to Tube-based MPC Approach

"Science can amuse and fascinate us all, but it is engineering that changes the world."

Isaac Asimov, 1920 – 1992

For linear discrete-time systems subject to an infinity-norm bounded disturbance, this chapter presents a Linear-Programming-based method for simultaneous calculation of an infinity-norm bounded robustly positively invariant set and a (constrained) state feedback gain that minimizes the  $L_{\infty}$ -norm of the output over this set. This result can be applied in tube-based MPC, where the robustly positively invariant set is used to tighten the nominal state and control constraints. In addition, this method is used to derive an infinity-norm bounded terminal constraint set and an infinity-norm-based terminal cost function for (tube-based) MPC approach, guaranteeing recursive feasibility and (robust) asymptotic stability of the

This chapter is based on [1].

closed-loop system. Application of the developed approach is demonstrated in a case study involving baggage handling systems.

# 4.1. Introduction

For linear systems subject to an additive disturbance, tube-based Model Predictive Control (MPC) provides a relatively simple approach to robust MPC design. A key element of this approach is the *tube*, which is the bundle of all trajectories of the uncertain system for a given disturbance set. Characterizing the tube is very closely related to calculating robustly positively invariant and robustly control invariant sets for the error system, which describes the error dynamics between the uncertain system and the nominal system (i.e., without disturbance). The problem of finding robustly positively invariant and robustly control invariant sets for a linear system is, in general, a complex problem. For a linear system subject to linear state and control constraints, the solution is an iterative process involving polyhedral projections and Minkowski differences [2–4], which results in a finite number of iterations only under certain conditions that depend on the system matrices (*A*, *B*) and the geometry of admissible state, and control, and disturbance sets [5, 6]. This limits practical applicability of tube-based MPC to small-scale systems, as for large-scale system polyhedral projections and Minkowski differences become prohibitively expensive.

In this chapter, we propose a solution for calculating robustly positively invariant and robustly control invariant sets for discrete-time linear positive systems subject to infinity-norm bounded constraints and infinity-norm bounded disturbances. We show that the problem of finding a robustly positively invariant set can be formulated as a linear program (LP), which can be efficiently solved even for large-scale systems. The LP-based approach for finding a robustly positively invariant set is later used in designing a tube-based MPC controller, where a constrained state feedback gain is calculated to not only stabilize the error system, but also to minimize the  $||L||_{\infty}$  norm of the output subject to the given disturbance set. Therefore, the tube-based MPC controller ensures the "smallest" deviation from the nominal trajectory in the presence of disturbances for a given control "budget", which determines the maximum control effort that can be generated by the feedback controller. We show that this state feedback controller can be obtained via a linear program as well.

Based on the approach developed for calculating a robustly control invariant set, we also develop a linear program for calculating a terminal constraint set and a terminal cost function for the (tube-based) MPC approach such that recursive feasibility and asymptotic stability of the closed-loop system is theoretically guaranteed.

To illustrate its applicability, we apply the developed methods in a case study involving automated baggage handling systems (BHSs), which have recently received a lot of attention due to growing demand for air travel. The reader is referred to [7] for a complete description of modern BHSs and their basic components. From a high-level perspective, the control problems associated with BHSs revolve around routing, dispatching, and scheduling of high-speed destination coded vehicles (DCVs) that transport the pieces of baggage between loading stations and unloading stations via various routes in the system. DCVs need to be controlled in such a way that pieces of luggage are delivered at their pre-assigned unloading stations within pre-specified time windows. A detailed description of the high-level control problems of BHSs can be found in [7, 8].

Having developed a modeling framework for BHSs based on *link densities*, we design a nominal MPC control strategy based on the base value of the baggage demand profile at loading stations, which is assumed to be a known function of time and, hence, fully predictable. As long as the actual baggage demand at loading stations is the nominal demand, this controller generates a routing policy that is optimal in the sense of *energy cost* and delivery time deviations from the target time window, while satisfying the capacity constraints of the BHS. We then consider structured uncertainty in the baggage demand profile and design a tube-based MPC approach to minimize the effect of uncertainty on the performance of the system, where the feedback gain for the tube-based MPC controller is calculated via the solution proposed in this chapter. In addition, we show how the terminal constraint set and the terminal cost function for the MPC and tube-based MPC problems can be calculated using our proposed approach. The effectiveness of tube-based MPC approach in the face of baggage demand uncertainty is then demonstrated by a simulation study.

The current chapter is different with respect to [8] and [9] in several key aspects: firstly, the solution proposed solution in [8] assumes full knowledge of the baggage demand and its future prediction at the loading stations, and hence is not robust against any possible uncertainty in the baggage demand. In the current chapter, the tube-based MPC approach is meant to deal with partially uncertain baggage demand. Secondly, in both [8] and [9], stability and *recursive feasibility* of the MPC controller are achieved indirectly by choosing a sufficiently long prediction horizon, whereas, here, these problems are directly addressed by introducing an appropriate *terminal state constraint* and a *terminal cost* [10, 11]. Finally, the two-level solution proposed in [9] incorporates an  $L_2$ -optimal feedback controller, which is formulated as Linear Matrix Inequalities (LMIs). In the current chapter we take advantage of the intrinsic nonnegative dynamics of our model, which allows for Linear Programming (LP) formulations outweigh the LMI approach in terms of scalability of the solution to large-scale BHSs.

The rest of this chapter is organized as follows. In Section 4.2 the terminology and definitions used throughout the chapter are introduced. In Section 4.3, we present some known results on discrete-time LTI positive systems and extend them to discrete-time systems, and formulate the output  $L_{\infty}$ -norm optimization problem for a discrete-time linear time-invariant positive system. The nominal and tubebased MPC problems are then formulated in Section 4.4. The dynamical model of the BHS is developed in Section 4.5. Using the developed model of the BHS, the proposed control approaches are then applied to a simulated case study and the results are presented in Section 4.6. Finally, we conclude this chapter in Section 4.7, pointing out possible directions for future research. Proofs of the propositions and theorems presented in the chapter are provided in Appendix 4.A.
# 4.2. Notation, terminology and definition

For sets A and B,  $A \oplus B := \{a + b \mid a \in A, b \in B\}$  is the Minkowski sum of the two sets. For non-negative integers *i* and *j*,  $i \le j$ ,  $\mathbb{N}_{i:j}$  is the finite set  $\{i, i + 1, ..., j\}$  of integers and  $\mathbb{N}_{i,\infty}$  is the infinite set  $\{i, i + 1, ...\}$  of integers. For a vector  $x \in \mathbb{R}^n$ ,  $x_i$  denotes its *i*-th element. In a similar manner for a matrix  $A \in \mathbb{R}^{n \times m}$ ,  $A_{i,j}$  denotes the element on the *i*-th row and the *j*-th column, and  $A_{i,:}$  and  $A_{:,j}$  respectively denote the *i*-th row of the matrix and its *j*-th column. We denote the all-ones vector in  $\mathbb{R}^n_+$ , the identity matrix of dimension *n*, and any zero matrix of suitable dimension, respectively, by  $1_n$ ,  $I_n$ , and 0. For matrices  $A, B \in \mathbb{R}^{n \times m}$ ,  $A \ge B$  (A > B), or equivalently  $A - B \in \mathbb{R}^{n \times m}_+$  ( $A - B \in \mathbb{R}^{n \times m}_+$ ) means that all entries of Z := A - B are non-negative (positive).

For a matrix  $A \in \mathbb{R}^{m \times n}$ ,  $\operatorname{conv}(A) \coloneqq \{\sum_{i=1}^{n} x_i A_{:,i} \mid \sum_{i=1}^{n} x_i = 1, x_i \ge 0, i \in \mathbb{N}_{1:n}\}$  is the convex set generated by its columns. The infinity-induced -norm of  $A \in \mathbb{R}^{m \times n}$ is defined as  $||A||_{\infty-\operatorname{ind}} \coloneqq \sup_{x \in \mathbb{R}^n \setminus \{0\}} \frac{||Ax||_{\infty}}{||x||_{\infty}}$ ; it is known [12] that  $||A||_{\infty-\operatorname{ind}} = \max_{i \in \mathbb{N}_{1:m}} (|A|1_n)_i$ , where  $|\cdot|$  denotes the element-wise absolute value of its argument. For a single-column matrix  $a \in \mathbb{R}^m$ , this definition renders into the conventional vector infinity norm  $||a||_{\infty-\operatorname{ind}} = ||a||_{\infty} \coloneqq \max_{i \in \mathbb{N}_{1:m}} (|a|)_i$ . For a matrix  $A \in \mathbb{R}^{m \times n}_+$  and a positive scalar  $\gamma$ , the inequality  $||A||_{\infty-\operatorname{ind}} < \gamma$  can then be equivalently expressed as  $A1_n < \gamma 1_m$ . For a discrete-time signal of dimension n, defined as the map  $x : \mathbb{N} \to \mathbb{R}^n$ , its  $L_{\infty}$  norm is defined as  $||x||_{L_{\infty}} \coloneqq \sup_{t \in \mathbb{N}} ||x(t)||_{\infty}$ .

Define  $x \in \mathbb{R}^n$ ,  $\omega \in \mathbb{R}^{n_{\omega}}$ , and  $z \in \mathbb{R}^{n_z}$  respectively as the state, disturbance, and monitored/controlled output. An autonomous (i.e., without control input) discrete-time linear time-invariant system of the form

$$\begin{aligned} x^{+} &= Ax + B_{1}\omega, \qquad (4.1)\\ z &= Cx + D_{11}\omega \\ &\text{with } A \in \mathbb{R}^{n \times n}, B_{1} \in \mathbb{R}^{n \times n_{u}}, C \in \mathbb{R}^{n_{z} \times n}, D_{11} \in \mathbb{R}^{n_{z} \times n} \end{aligned}$$

is called a *positive system* [13–15] if  $A \ge 0$ ,  $B_1 \ge 0$ ,  $C \ge 0$ , and  $D_{11} \ge 0$ . For system (4.1) with initial state  $x_0 \coloneqq x(0)$ , and disturbance sequence  $\omega_k \coloneqq (\omega(0), \omega(1), ..., \omega(k-1), \omega(k))$  of length k + 1,  $k \in \mathbb{N}$ , the state and output trajectories at time  $t \in \mathbb{N}_{0:k}$  are given explicitly by the formula

$$\boldsymbol{x}(t;\boldsymbol{x}_0,\boldsymbol{\omega}_k) \coloneqq \boldsymbol{A}^t \boldsymbol{x}_0 + \sum_{s=0}^{t-1} \boldsymbol{A}^s (\boldsymbol{B}_1 \boldsymbol{\omega}_k (t-s)), \qquad (4.2a)$$

$$\boldsymbol{z}(t;\boldsymbol{x}_0,\boldsymbol{\omega}_k) \coloneqq \boldsymbol{C}\boldsymbol{x}(t;\boldsymbol{x}_0,\boldsymbol{\omega}_k) + \boldsymbol{D}_{11}\boldsymbol{\omega}_k(t+1). \tag{4.2b}$$

For a positive system with  $x_0 \ge 0$ , and  $\omega_k \in \mathbb{R}^{n_\omega(k+1)}_+$ ,  $k \in \mathbb{N}_+$ , it is obvious that  $x(t; x_0, \omega_k) \ge 0$  and  $z(t; x_0, \omega_k) \ge 0$  for all  $t \in \mathbb{N}_{0:k}$ .

# 4.3. Output $L_{\infty}$ -norm-optimal Feedback Control for Positive Systems

#### **4.3.1.** Preliminaries

In this section, we focus on a system of the form

$$\begin{aligned} \mathbf{x}^{+} &= \mathbf{A}\mathbf{x} + \mathbf{B}_{1}\boldsymbol{\omega} + \mathbf{B}_{2}\mathbf{u}, \\ \mathbf{z} &= \mathbf{C}\mathbf{x} + \mathbf{D}_{11}\boldsymbol{\omega} + \mathbf{D}_{12}\mathbf{u}, \\ \mathbf{A} \in \mathbb{R}^{n \times n}, \ \mathbf{B}_{1} \in \mathbb{R}^{n \times n_{\omega}}, \ \mathbf{B}_{2} \in \mathbb{R}^{n \times n_{u}}, \ \mathbf{C} \in \mathbb{R}^{n_{z} \times n}, \\ \mathbf{D}_{11} \in \mathbb{R}^{n_{z} \times n}, \ \mathbf{D}_{12} \in \mathbb{R}^{n_{z} \times n}. \end{aligned}$$
(4.3)

Given compact sets  $\mathbb{X}_0 \subseteq \mathbb{R}^n$ ,  $\mathbb{W} \subseteq \mathbb{R}^{n_\omega}$ ,  $\mathbb{U} \subseteq \mathbb{R}^{n_u}$  as the set of initial conditions, the disturbance set, and the set of admissible controls with  $0 \in \mathbb{X}_0$ ,  $0 \in \mathbb{W}$ , and  $0 \in \mathbb{U}$ , we are interested in the smallest, in the sense of the  $L_\infty$ -norm of the output, robustly positively invariant set [2]  $\mathbb{X} \supseteq \mathbb{X}_0$  achievable with a linear state feedback law of the form  $\boldsymbol{\mu} : \boldsymbol{x} \mapsto K\boldsymbol{x}$ ,  $\boldsymbol{u}(t) = \boldsymbol{\mu}(\boldsymbol{x}(t))$  that renders the closed-loop system positive. Hence, a feedback gain  $\boldsymbol{K} \in \mathbb{R}^{nu \times n}$  and a set  $\mathbb{X} \supseteq \mathbb{X}_0$  are sought such that the closed-loop system

$$x^{+} = (A + B_{2}K)x + B_{1}\omega,$$

$$z = (C + D_{12}K)x + D_{11}\omega,$$

$$u = Kx,$$
(4.4)

is positive, has the property that  $x \in \mathbb{X}$  implies  $u \in \mathbb{U}$  and  $x^+ \in \mathbb{X}$  for all  $\omega \in \mathbb{W}$ , and achieves minimum  $||z||_{L_{\infty},\mathbb{W},\mathbb{X}}$  defined as

 $\|\boldsymbol{z}\|_{L_{\infty},\mathbb{W},\mathbb{X}} \coloneqq \max_{\boldsymbol{\omega}\in\mathbb{W},\boldsymbol{x}_{0}\in\mathbb{X}} \|\boldsymbol{z}\|_{L_{\infty}} = \sup_{\boldsymbol{k}\in\mathbb{N},\,t\in\mathbb{N}_{0:k},\,\boldsymbol{x}_{0}\in\mathbb{X},\,\boldsymbol{\omega}_{k}\in\mathbb{W}^{k+1}} \|\boldsymbol{z}(t;\boldsymbol{x}_{0},\boldsymbol{\omega}_{k})\|_{\infty}.$ 

We consider the case where both the robustly positively invariant state set in question, and the given admissible control and disturbance sets are infinity-norm bounded sets of the form  $\mathbb{Y} := \{ y \in \mathbb{R}^{n_y} \mid ||Y^{-1}y||_{\infty} \leq 1, Y := \text{diag}(y_M), y_M \in \mathbb{R}^{n_y}_{+,s} \}.$ 

# 4.3.2. Upper Bound on $L_{\infty}$ -norm of Output

We first present the following result regarding establishing an upper bound on  $||\mathbf{z}||_{L_{\infty}, W, X}$  for the open-loop positive system (4.1) that will be used when discussing the state feedback design problem.

Proposition 4.1 (Bound on  $\mathbf{z}_{L_{\infty}, \mathbb{W}, \mathbb{X}}$ ). Consider the positive system (4.1) with a positive scalar  $\gamma \in \mathbb{R}_{+,s}$ ,  $\mathbb{W} := \{\boldsymbol{\omega} \in \mathbb{R}^{n_{\omega}} \mid || \boldsymbol{\Omega}^{-1} \boldsymbol{\omega} ||_{\infty} \leq 1$ ,  $\boldsymbol{\Omega} := \text{diag}(\boldsymbol{\omega}_{M}), \boldsymbol{\omega}_{M} \in \mathbb{R}^{n_{\omega}}_{+,s}\}$ ,  $\mathbb{X} := \{\boldsymbol{x} \in \mathbb{R}^{n} \mid || \boldsymbol{P}^{-1} \boldsymbol{x} ||_{\infty} \leq 1$ ,  $\boldsymbol{P} := \text{diag}(\boldsymbol{p}), \boldsymbol{p} \in \mathbb{R}^{n}_{+,s}\}$ , and  $\mathbb{X}_{0} = \text{conv}(\boldsymbol{M}^{T})$ ,  $\boldsymbol{M} \in \mathbb{R}^{n \times r}$ . Then  $\mathbb{X}$  is a robustly positively invariant set containing  $\mathbb{X}_{0}$  for the system with  $\boldsymbol{\omega} \in \mathbb{W}$ , and  $|| \boldsymbol{z} ||_{L_{\infty}, \mathbb{W}, \mathbb{X}}$  of the system is bounded as  $|| \boldsymbol{z} ||_{L_{\infty}, \mathbb{W}, \mathbb{X}} < \gamma$  if and

only if

$$(\mathbf{A} - \mathbf{I}_n)\mathbf{P}\mathbf{1}_n + \mathbf{B}_1\mathbf{\Omega}\mathbf{1}_{n_\omega} < 0, \tag{4.5a}$$

$$CP1_n + D_{11}\Omega 1_{n_\omega} - \gamma 1_{n_z} < 0,$$
 (4.5b)

$$P \ge \operatorname{diag}([\|M(:,1)\|_{\infty} \dots \|M(:,n)\|_{\infty}]^{1}).$$
(4.5c)

т

See Appendix 4.A for the proof.

*Remark* 1. Note that with  $\boldsymbol{\Omega} = I_{n_{\omega}}$  and  $\mathbb{X}_0 = \{0\}$ , inequalities 4.5 of Proposition 4.1 are necessary and sufficient conditions for establishing  $\gamma$  as an upper bound to the  $L_{\infty}$ -gain of the system, defined as  $\|\boldsymbol{G}\|_{\infty} \coloneqq \sup_{\boldsymbol{\omega} \in \mathcal{L}_{\infty}} \frac{\|\boldsymbol{z}\|_{L_{\infty}}}{\|\boldsymbol{\omega}\|_{L_{\infty}}}$  [16, 17]. Hence, the search for upper bound to  $L_{\infty}$ -gain of the system, can be considered an special case of finding an upper bound to  $\|\boldsymbol{z}\|_{L_{\infty}, \mathbb{W}, \mathbb{X}}$ .

*Remark* 2. For the positive system (4.1), the statements of proposition 4.1 are also valid when the state and disturbance are restricted to their corresponding positive orthants; that is, when X and W are respectively replaced by  $X_+ := X \cap \mathbb{R}^n_+$  and  $W_+ := W \cap \mathbb{R}^{n_{\omega}}_+$ .

#### **4.3.3.** Derivation of the Feedback Gain

Now, we will discuss the state feedback design problem. The method presented in this section will be used in Section 4.4.2 for designing a tube-based MPC controller. Given the infinity norm-bounded sets  $W := \{ \boldsymbol{\omega} \in \mathbb{R}^{n_{\omega}} \mid \|\boldsymbol{\Omega}^{-1}\boldsymbol{\omega}\|_{\infty} \leq 1, \boldsymbol{\Omega} := \text{diag}(\boldsymbol{\omega}_{M}), \boldsymbol{\omega}_{M} \in \mathbb{R}^{n_{\omega}}_{+,s} \}, \mathbb{U} := \{ \boldsymbol{u} \in \mathbb{R}^{n_{u}} \mid \|\boldsymbol{U}^{-1}\boldsymbol{u}\|_{\infty} \leq 1, \boldsymbol{U} := \text{diag}(\boldsymbol{u}_{M}), \boldsymbol{u}_{M} \in \mathbb{R}^{n_{u}}_{+,s} \},$  and the polyhedral set  $X_{0} = \text{conv}(\boldsymbol{M}^{T}), \boldsymbol{M} \in \mathbb{R}^{n \times r}$ , existence and derivation of the feedback gain  $\boldsymbol{K}$  and the robustly positively invariant set  $X \supseteq X_{0}$  are the subject of the following theorem:

**Theorem 4.2.** The feedback gain *K* renders the closed-loop system 4.4 positive with minimum  $||\mathbf{z}||_{L_{\infty}, \mathbb{W}, \mathbb{X}}$  over a robustly positively invariant set  $\mathbb{X} := \{\mathbf{x} \in \mathbb{R}^n \mid ||\mathbf{P}^{-1}\mathbf{x}||_{\infty} \leq 1, \mathbf{P} = \text{diag}(\mathbf{p}), \mathbf{p} \in \mathbb{R}^n_{+,s}\}, \mathbb{X} \supseteq \mathbb{X}_0$  such that  $K\mathbf{x} \in \mathbb{U}$  for all  $\mathbf{x} \in \mathbb{X}$  and all  $\boldsymbol{\omega} \in \mathbb{W}$ , if and only if there is a solution  $\gamma \in \mathbb{R}_{+,s}$ ,  $\mathbf{p} \in \mathbb{R}^n_{+,s}$ ,  $\mathbf{Y}_1 \in \mathbb{R}^{n_u \times n}_+$  and  $\mathbf{Y}_2 \in \mathbb{R}^{n_u \times n}_+$  to the linear program

$$\inf_{\boldsymbol{p}\in\mathbb{R}^{n}_{+,s},Y_{1},Y_{2}\in\mathbb{R}^{n_{u}\times n}_{+},\gamma>0}\gamma,$$
(4.6a)

subject to :

$$AP + B_2(Y_1 - Y_2) \ge 0$$
,  $CP + D_{12}(Y_1 - Y_2) \ge 0$ , (4.6b)

$$\begin{bmatrix} (\boldsymbol{A} - \boldsymbol{I}_n)\boldsymbol{P} + \boldsymbol{B}_2(\boldsymbol{Y}_1 - \boldsymbol{Y}_2) & \boldsymbol{B}_1\boldsymbol{\Omega}\boldsymbol{1}_{n_{\omega}} \\ \boldsymbol{C}\boldsymbol{P} + \boldsymbol{D}_{12}(\boldsymbol{Y}_1 - \boldsymbol{Y}_2) & \boldsymbol{D}_{11}\boldsymbol{\Omega}\boldsymbol{1}_{n_{\omega}} - \gamma\boldsymbol{1}_{n_z} \end{bmatrix} \begin{bmatrix} \boldsymbol{1}_n \\ \boldsymbol{1} \end{bmatrix} < 0,$$
(4.6c)

$$P \ge \operatorname{diag}([\|M(:,1)\|_{\infty} \dots \|M(:,n)\|_{\infty}]^{T}),$$
(4.6d)

$$\begin{bmatrix} Y_1 & Y_2 \end{bmatrix} S \le U \mathbf{1}_{n_u, 2^n} \text{, where} \begin{cases} S_{i,j} = 1, S_{i+n,j} = 0 & v_i^j > 0 \\ S_{i,j} = 0, S_{i+n,j} = 1 & \text{otherwise} \end{cases},$$
(4.6e)

for all  $i \in \mathbb{N}_{1:n}$ ,  $j \in \mathbb{N}_{1:2^n}$ , where  $v^j \in \mathbb{R}^n$ ,  $j \in \mathbb{N}_{1:2^n}$  is the *j*-th vertex of the unit hypercube in  $\mathbb{R}^n$ . The optimal feedback gain  $K^*$  is retrieved from the optimization variables P,  $Y_1$ , and  $Y_2$  using  $K^* = (Y_1 - Y_2)P^{-1}$ . In addition, the closed-loop system under the optimal feedback control  $u \coloneqq \mu(x) = K^*x$  is globally ultimately bounded in the set  $\mathbb{X}$ .

the proof can be found in Appendix 4.A.

Note that, as highlighted in Remark 1, with  $\Omega = I_n$ ,  $\mathbb{U} = \mathbb{R}^{n_u}$ , and  $\mathbb{X}_0 = \{0\}$ , Theorem 4.2 yields a positive closed-loop system with minimum  $L_{\infty}$ -gain. The problem of closed-loop  $L_{\infty}$ -gain minimization with control constraints has been discussed in [17, 18], where the state of the system is restricted to the positive orthant. However, that approach does not automatically apply to the case when the state of the system belongs to a infinity-norm bounded set in  $\mathbb{R}^n$ .

The linear program (4.6) involves  $n + 2nn_u + 1$  optimization variables, and  $2n_z + 3n + 2^n n_u$  linear inequalities, which may get prohibitively complex for large-scale systems with an increasing number of variables. The number of inequality constraints can be significantly reduced by replacing the  $2^n n_u$  inequalities in 4.6e by the conservative approximation  $[Y_1 Y_2]1_{2n} \leq U1_{n_u}$ , which consists of  $n_u$  inequalities. In addition, the number of optimization variables can be reduced by employing  $Y := Y_1 - Y_2$ ,  $Y \in \mathbb{R}^{n_u \times n}$  and replacing 4.6e by the conservative approximation  $\sum_{i=1}^n ||U^{-1}Y_{:,i}||_{\infty} \leq 1$ , which is equivalent to  $Y[v^1 \dots v^{2^n}] \leq U1_{n_u,2^n}$ , where  $v^j \in \mathbb{R}^n$ , is the *j*-th vertex of the unit *n*-dimensional hypercube.

# 4.3.4. Derivation of the Feedback Gain With Soft Control Constraints

Rather than specifying hard control constraints in the form of  $u \in U$ , as discussed in Section 4.3.3, one may be interested in keeping the control effort relatively small, which is applicable when the resulting feedback control only constitutes a part of the final control input to the system, and the control input constraint is handled by a higher-level controller. In addition, in this case, infeasibility of the optimization problem (4.6) due to an overly stringent specification of control constraint is avoided. The optimization problem

$$\inf_{\substack{p \in \mathbb{R}^{n}_{+,s}, Y_{1}, Y_{2} \in \mathbb{R}^{n_{u} \times n}_{+}, \gamma > 0, r \ge 0}} \gamma + c_{r}r, \qquad (4.7a)$$
subject to:
$$(4.6b), (4.6c), (4.6d), \\
\begin{bmatrix} Y_{1} & Y_{2} \\ -Y_{1} & -Y_{2} \end{bmatrix} S \le r \mathbf{1}_{2n_{u},2^{n}}, \text{ where } \begin{cases} S_{i,j} = 1, S_{i+n,j} = 0 & v_{i}^{j} > 0 \\ S_{i,j} = 0, S_{i+n,j} = 1 & \text{otherwise} \end{cases}, \quad (4.7b)$$

$$i \in \mathbb{N}_{1:n}, j \in \mathbb{N}_{1:2^{n}},$$

where  $v^j \in \mathbb{R}^n$ ,  $j \in \mathbb{N}_{1:2^n}$  is the *j*-th vertex of the unit n-dimensional hypercube in  $\mathbb{R}^n$ , imposes no constraint on the control input and achieves an arbitrary trade-off, determined by the weighting factor  $c_r > 0$ , between  $||\mathbf{z}||_{L_{\infty}, \mathbb{W}, \mathbb{X}}$  and  $||\mathbf{u}||_{L_{\infty}}$ . The optimal feedback gain is then given by  $K^* = (Y_1 - Y_2)P^{-1}$ .

Since X is robustly positively invariant, it holds that  $\sum_{i=1}^{n} \|Y_{1:i} - Y_{2:i}\|_{\infty} \ge \|Kx\|_{\infty}$ =  $\|u\|_{L_{\infty}}$  for any  $x \in X$ , hence, constituting an upper bound to  $\|u\|_{L_{\infty}}$ . Using this upper bound with  $Y := Y_1 - Y_2$ , the  $2^{n+1}n_u$  inequalities of 4.7b are decreased to  $2nn_u$  inequalities in the following linear program, which minimizes a mixed objective function composed of a penalty term for  $\|z\|_{L_{\infty},W,X}$  and a penalty term for the upper bound of  $\|u\|_{L_{\infty}}$ :

$$\inf_{\boldsymbol{p}\in\mathbb{R}^{n}_{+,s},\,\boldsymbol{Y}\in\mathbb{R}^{n_{u}\times n},\,\boldsymbol{\gamma}>0,\,\boldsymbol{r}\in\mathbb{R}^{n}_{+}}\boldsymbol{\gamma}+c_{r}\mathbf{1_{n}}^{\mathrm{T}}\boldsymbol{r},$$
subject to :
4.6b, 4.6c, 4.6d,
$$\begin{bmatrix}\boldsymbol{Y}\\-\boldsymbol{Y}\end{bmatrix}\leq\mathbf{1}_{2n_{u}}\boldsymbol{r}^{\mathrm{T}}.$$
(4.8b)

The optimal feedback gain is then given by  $K^* = YP^{-1}$ .

# **4.4. Model Predictive Control**

This section explains the concept of model predictive control (MPC), especially within the scope of its application to a BHS described by the dynamic model (4.3). The external input  $\omega(t) \coloneqq \overline{\omega}(t) + \tilde{\omega}(t)$  is assumed to be composed of  $\overline{\omega}(t)$  with known values for all t, and a nonnegative additive unmeasurable component  $\tilde{\omega}(t)$ . Within the context of BHSs,  $\overline{\omega}(t)$  corresponds to the baseline baggage demand due to the known flight schedule and  $\tilde{\omega}(t)$  corresponds to unplanned excess of demand. The system is endowed with state and control constraints of the form

$$\boldsymbol{x}(t) \in \mathbb{X} \coloneqq \{ \boldsymbol{x} \in \mathbb{R}^n | 0 \le \boldsymbol{x} \le \boldsymbol{x}_{\max} \}, \, \forall \, t \in \mathbb{N},$$

$$(4.9a)$$

$$\boldsymbol{u}(t) \in \mathbb{U} \coloneqq \{ \boldsymbol{u} \in \mathbb{R}^{n_u} | 0 \le \boldsymbol{u} \le \boldsymbol{u}_{\max} \}, \, \forall \, t \in \mathbb{N},$$

$$(4.9b)$$

corresponding to operational constrains such as capacity limits and actuation saturation.

At every time step, based on the current state of the system and future values of the known component of the external input, a constrained finite-horizon optimization problem with a linear cost subject to (4.9) will be solved yielding a sequence of optimal control inputs. According to the receding horizon policy, only the first step of this sequence is applied to the system, and this process is repeated at the next time step [10, 11].

We develop two MPC strategies. One that achieves optimal performance when  $\tilde{\omega}(t) = 0$  and one that provides robustness in presence of  $\tilde{\omega}(t)$ . In both cases, the underlying optimization problem is formulated as an LP program by employing a linear cost function together with the linear constraints (4.9). For both MPC schemes, while optimizing for the performance defined by the cost function, stability of closed-loop system is guaranteed by proper selection of weighting vectors of the cost function.

#### **4.4.1. Baseline MPC**

For the baseline MPC design, we consider the system (4.3) with  $\omega(t) \coloneqq \overline{\omega}(t)$ , which is a known function of time *t*.

#### **Baseline MPC Problem Formulation**

Let  $N_h$  be the prediction horizon. Given the current time step t and the current state of the system  $x \coloneqq x(t)$ , the finite-horizon optimization problem that needs to be solved is defined as

$$\mathcal{P}_{N_{h}}(t, \mathbf{x}) : V_{N_{h}}^{*}(t, \mathbf{x}) = \min_{u_{N_{h}}} \{ V_{N_{h}}(t, \mathbf{x}, u_{N_{h}}(t)) \mid u_{N_{h}}(t) \in \mathcal{U}_{N_{h}}(t, \mathbf{x}) \},$$
(4.10a)

where,

64

$$V_{N_{h}}(t, x, u_{N_{h}}(t)) = \sum_{k=0}^{N_{h}-1} \left( q_{z}^{T}(t+k)z(t+k) + q_{u}^{T}(t+k)u(t+k) \right)$$
(4.10b)

$$+ v_{f}(t + N_{h}, \boldsymbol{x}(t + N_{h})),$$

$$\mathcal{U}_{N_{h}}(t, \boldsymbol{x}) = \{\boldsymbol{u}_{N_{h}} \mid 0 \le \boldsymbol{x}(t + k) \le \boldsymbol{x}_{\max}, \quad 0 \le \boldsymbol{u}(t + k) \le \boldsymbol{u}_{\max}, \quad (4.10c)$$

$$\forall \ k \in \mathbb{N}_{0:N_{h}-1}, \ \boldsymbol{x}(t + N_{h}) \in \mathbb{X}_{f}\},$$

$$\mathbf{x}(t+k) = \mathbf{x}\left(t+k; \mathbf{x}, \mathbf{u}_{N_{\rm h}}(t), \overline{\mathbf{\omega}}_{N_{\rm h}}(t)\right), \ k \in \mathbb{N}_{0:N_{\rm h}},\tag{4.10d}$$

$$\mathbf{z}(t+k) = \mathbf{z}(t+k; \mathbf{x}, \mathbf{u}_{N_{\mathsf{h}}}(t), \overline{\mathbf{\omega}}_{N_{\mathsf{h}}}(t)), \ k \in \mathbb{N}_{0:N_{\mathsf{h}}-1},$$
(4.10e)

with  $\boldsymbol{u}_{N_{h}}(t) \coloneqq (\boldsymbol{u}(t), ..., \boldsymbol{u}(t + N_{h} - 1))$  and  $\overline{\boldsymbol{\omega}}_{N_{h}}(t) \coloneqq (\overline{\boldsymbol{\omega}}(t), ..., \overline{\boldsymbol{\omega}}(t + N_{h} - 1))$  respectively being the sequence of to-be-calculated control actions and the sequence of known disturbances over the prediction horizon. In above, the weighting functions for the stage cost  $\boldsymbol{q}_{z} : \mathbb{N} \to \mathbb{R}^{n_{z}}_{+,s}$  and  $\boldsymbol{q}_{u} : \mathbb{N} \to \mathbb{R}^{n_{u}}_{+,s}$ , the terminal cost function  $V_{f} : (\mathbb{N}, \mathbb{X}) \to \mathbb{R}_{+,s}^{n}$ , and the terminal constraint set  $\mathbb{X}_{f} \subseteq \mathbb{X} \subseteq \mathbb{R}^{n}_{+}$ , are design parameters, the choice of which will be discussed in the sequel.

Assume  $x \in \mathcal{X}_{N_{h}}(t)$ , where  $\mathcal{X}_{N_{h}}(t)$  is the domain of  $\mathcal{P}_{N_{h}}(t, \cdot)$ , which is the set of all states at time step t for which the optimization problem has a solution. The solution to the optimization problem  $\mathcal{P}_{N_{h}}(t, x)$  is then the optimal control sequence  $u_{N_{h}}^{*}(t, x) = (u^{*}(t; t, x), ..., u^{*}(t + N_{h} - 1; t, x))$ . Finally, according to the receding horizon policy, the control action to be applied to the system at the current time step is given as the first element of the optimal control sequence; that is

$$\boldsymbol{u}_{\text{mpc}}(t) \coloneqq \boldsymbol{u}^*(t; t, \boldsymbol{x}). \tag{4.11}$$

Note that the even though the system described by (4.3) is time-invariant, the optimization problem and the resulting optimal control sequence are time-dependent due time-varying  $\overline{\omega}_{N_h}$  and the time-dependent weighting factors.

#### 4.4.2. Tube-based MPC

Here we assume that the external input to the system (4.3) is in the form of  $\omega(t) = \overline{\omega}(t) + \widetilde{\omega}(t)$ , where, just as before,  $\overline{\omega}(t)$  is known and  $\widetilde{\omega}(t) \in \mathbb{W} := \{\omega \ge \omega \}$ 

4

 $0 \mid \|\mathcal{D}^{-1}\omega\|_{\infty} \leq 1, \ \mathcal{D} = \text{diag}(\omega_M), \ \omega_M \in \mathbb{R}^{n_{\omega}}_{+,s}$  for all  $t \in \mathbb{N}$ . The control scheme discussed in this section aims at an optimal performance of the closed-loop system while being robust against perturbations of the external input due to  $\tilde{\omega}$ . Among others, *tube-based MPC* is an MPC-based approach that allows us to achieve this goal. This approach has the advantage that it can be derived with little modification to the baseline MPC problem 4.10 and it admits a simple implementation. The design of a tube-based MPC controller involves the following steps: i) characterize the *bounding tube* of uncertain system trajectories for all  $\tilde{\omega}(t) \in \mathbb{W}$ , ii) design a stabilizing feedback controller to ensure boundedness of the generated tube, and iii) design an MPC controller, satisfying state and control constraints for all state trajectories in the tube, in order to generate the nominal trajectory. Tube-based MPC is covered in depth in [10, Chapter 3], of which we provide an overview.

#### **Bounding Tube of System Trajectories**

In tube-based MPC approach, a *bundle* of trajectories containing states of the uncertain system for all  $\tilde{\omega}(t) \in \mathbb{W}$  is used to represent the state trajectory of the uncertain system. The bundle of state trajectories is called a *tube* and is supported by the nominal trajectory, which is the trajectory of the system driven by a nominal control signal  $\overline{u}(\cdot)$  with  $\tilde{\omega}(\cdot) = 0$ . The *size* of the tube represents deviation from the nominal trajectory for all values of  $\tilde{\omega}(t) \in \mathbb{W}$ . Consider system (4.3) and let the uncertain and nominal systems be respectively expressed by

$$\mathbf{x}(t+1) = \mathbf{A}\mathbf{x}(t) + \mathbf{B}_2\mathbf{u}(t) + \mathbf{B}_1\overline{\mathbf{\omega}}(t) + \mathbf{B}_1\tilde{\mathbf{\omega}}(t), \qquad (4.12)$$

$$\mathbf{z}(t) = \mathbf{C}\mathbf{x}(t) + \mathbf{D}_{12}\mathbf{u}(t) + \mathbf{D}_{11}\overline{\mathbf{\omega}}(t) + \mathbf{D}_{11}\tilde{\mathbf{\omega}}(t)$$
(4.13)

$$\overline{\mathbf{x}}(t+1) = A\overline{\mathbf{x}}(t) + B_2\overline{\mathbf{u}}(t) + B_1\overline{\mathbf{\omega}}(t), \qquad (4.14)$$

$$\overline{\mathbf{z}}(t) = \mathbf{C}\overline{\mathbf{z}}(t) + \mathbf{D}_{12}\overline{\mathbf{u}}(t) + \mathbf{D}_{11}\overline{\mathbf{\omega}}(t), \qquad (4.15)$$

where  $\overline{x}(t)$  and  $\overline{z}(t)$  are, respectively, the state and output of the nominal system, and x(t) and z(t) are those of the disturbed (uncertain) system, and  $\overline{u}(t)$  is the nominal control signal generated by the MPC controller, and u(t) is the control input to the disturbed system, which is of the form

$$\boldsymbol{u}(t) = \overline{\boldsymbol{u}}(t) + \boldsymbol{K}(\boldsymbol{x}(t) - \overline{\boldsymbol{x}}(t)), \qquad (4.16)$$

where the feedback gain *K* is such that  $A+B_2K$  is stable, guaranteeing boundedness of the generated tube. Hence, with  $x_e(t) \coloneqq x(t) - \overline{x}(t)$  and  $z_e(t) \coloneqq z(t) - \overline{z}(t)$ ,

$$x_{e}(t+1) = (A + B_{2}K)x_{e}(t) + B_{1}\tilde{\omega}(t), \qquad (4.17a)$$

$$\mathbf{z}_{e}(t+1) = (\mathbf{C} + \mathbf{D}_{12}\mathbf{K})\mathbf{x}_{e}(t) + \mathbf{D}_{11}\tilde{\boldsymbol{\omega}}(t), \qquad (4.17b)$$

describe the error dynamics. Assuming  $x(0) = \overline{x}(0)$ , it follows from (4.17) that  $x_e(t) \in S(t)$  for all  $t \in \mathbb{N}$ , where

$$\mathbb{S}(t) \coloneqq \boldsymbol{B}_1 \mathbb{W} \bigoplus \boldsymbol{A}_e \boldsymbol{B}_1 \mathbb{W} \bigoplus \dots \bigoplus \boldsymbol{A}_e^{t-1} \boldsymbol{B}_1 \mathbb{W} = \sum_{i=0}^{t-1} \boldsymbol{A}_e^i \boldsymbol{B}_1 \mathbb{W}, \quad (4.18)$$

with  $A_e \coloneqq A + B_2 K$ . The set S(t) contains the state trajectories of the error system with  $x_e(0) = 0$  for all possible sequences of  $(\tilde{\omega}(0), ..., \tilde{\omega}(t-1))$  and for any  $t \in \mathbb{N}$ . Hence, for the disturbed system we have  $x_e(t) \in X(t) \coloneqq \{\bar{x}(t)\} \oplus S(t), \forall t \in \mathbb{N}$ .

For the MPC use case, the bundle of all trajectories over the prediction horizon  $N_{\rm h} \in \mathbb{N}$  starting from the initial state  $\mathbf{x}(t) = \overline{\mathbf{x}}(t) = \overline{\mathbf{x}}$  under the nominal control sequence  $\overline{\mathbf{u}}_{N_{\rm h}}(t) = (\overline{\mathbf{u}}(t), ..., \overline{\mathbf{u}}(t+N_{\rm h}-1))$  and the known external input sequence  $\overline{\mathbf{\omega}}_{N_{\rm h}}(t) = (\overline{\mathbf{\omega}}(t), ..., \overline{\mathbf{\omega}}(t+N_{\rm h}-1))$  for all possible values of disturbance sequence  $\widetilde{\mathbf{\omega}}_{N_{\rm h}}(t) = (\widetilde{\mathbf{\omega}}(t), ..., \widetilde{\mathbf{\omega}}(t+N_{\rm h}-1)) \in \mathbb{W}^{N_{\rm h}}$ , is the tube generated at  $\mathbf{x}(t)$  by  $\overline{\mathbf{u}}_{N_{\rm h}}(t)$  and  $\overline{\mathbf{\omega}}_{N_{\rm h}}(t)$ , defined as

$$X_{N_{h}}(\overline{x}, \overline{u}_{N_{h}}(t), \overline{\omega}_{N_{h}}(t)) \coloneqq (X(0; \overline{x}, \overline{u}_{N_{h}}(t), \overline{\omega}_{N_{h}}(t)), \dots,$$

$$X(N_{h}; \overline{x}, \overline{u}_{N_{h}}(t), \overline{\omega}_{N_{h}}(t))) = (\{\overline{x}\}, \{\overline{x}(t+1)\} \oplus \mathbb{S}(1), \dots, \{\overline{x}(t+N_{h})\} \oplus \mathbb{S}(N_{h})).$$
(4.19)

The tube described by 4.19 is generated assuming  $x_e(0) = 0$  (i.e.,  $x(t) = \overline{x}(t)$ ). An outer bounding tube taking initial conditions into account can be obtained by using  $\mathbb{S}_{\infty} \coloneqq \lim_{t \to \infty} \mathbb{S}(t)$  instead of  $\mathbb{S}(t)$ ,  $t \in \mathbb{N}_{0:N_h}$ , which is the minimal robust positive invariant set for the system (4.17) with  $A_e$  stable [10, 19]. However,  $\mathbb{S}_{\infty}$  is often very difficult to calculate requiring one to use an approximation of it. In addition, using  $\mathbb{S}_{\infty}$  one only takes into account initial conditions  $x_e(0) \in \mathbb{S}_{\infty}$ .

Our approach utilizes Proposition 4.1 and the associated state feedback derivation method of Section 4.3.3 for system 4.17 to compute a robustly positively invariant set S', which includes a pre-specified minimum set S<sub>min</sub>, and a state feedback gain *K*. The state feedback gain *K* calculated in this manner not only does stabilize the error system, but it also achieves minimal  $||z_e||_{L_{\infty}, W, S_{\min}}$  with constrained control effort. An outer tube containing trajectories of the uncertain system for all  $x_e(0) \in S'$  and all  $\tilde{\omega} \in W$  is constructed as

$$X'_{N_{h}}(\overline{x},\overline{u}_{N_{h}}(t),\overline{\omega}_{N_{h}}(t)) \coloneqq (\{\overline{x}(t)\} \oplus \mathbb{S}',\{\overline{x}(t+1)\} \oplus \mathbb{S}',..., \qquad (4.20)$$
$$\{\overline{x}(t+N_{h})\} \oplus \mathbb{S}').$$

**Tube-based MPC Problem Formulation** 

The tube-based MPC problem is then defined as a modified version of the original MPC problem 4.10, where the set S' is used to tighten the original MPC constraints 4.10c so that the original MPC constraints are satisfied by the tube  $X'_{N_h}(x, \overline{u}_{N_h}, \overline{\omega}_{N_h})$ . Let S' := { $e \mid 0 \le e \le p, p \in \mathbb{R}^n_{+,s}$ } and  $K \in \mathbb{R}^{n_u \times n}_+$ , respectively, be the robustly positively invariant set and the associated state feedback gain obtained using the method of Section 4.3.3 for the error system 4.17 with  $\tilde{\omega} \in \mathbb{W}$ . At the current state of the nominal system  $\overline{x} := \overline{x}(t)$ , the tube-based MPC problem  $\overline{\mathcal{P}}_{N_h}(t, \overline{x})$  is defined as  $\mathcal{P}_{N_h}(t, x)$  with  $x = \overline{x}$  and with tightened constraints  $0 \le \overline{x}(t + k) \le x_{max} - p$  and  $0 \le \overline{u}(t + k) \le u_{max} - Kp$ ,  $k \in \mathbb{N}_{0:N_h-1}$ ,  $\overline{x}(t + N_h) \in \overline{X}_f$  replacing the original constraints in 4.10, where  $\overline{X}_f$  is a tightened version of the terminal constraint set  $X_f$ . The choice of  $X_f$  and  $\overline{X}_f$  will be discussed in detail in Section 4.4.3.

**MPC Problem Formulation With Improved Tube Base** 

In the tube-based MPC approach, it is also possible to optimize for the base of the tube  $\overline{x} \coloneqq \overline{x}(t)$  to achieve improved performance with respect to the MPC cost function. Let  $\overline{x} \coloneqq \overline{x}(t) \in \mathcal{X}_{N_{h}}(t)$  and  $x \coloneqq x(t) \in \{\overline{x}(t)\} \oplus \mathbb{S}'$  respectively be the current state of the nominal system and that of the uncertain system, where  $\mathcal{X}_{N_{h}}(t)$  is the domain of  $\overline{\mathcal{P}}_{N_{h}}(t,\overline{x})$ . The MPC problem with improved tube base is then defined as

$$\mathcal{P}_{N_{h}}^{*}(t,\boldsymbol{x}): V_{N_{h}}^{*}(t,\boldsymbol{x}) = \min_{\boldsymbol{v}, \overline{\boldsymbol{u}}_{N_{h}}} \{V_{N_{h}}(t,\boldsymbol{x},\boldsymbol{v}, \overline{\boldsymbol{u}}_{N_{h}}(t)) \mid \boldsymbol{x} - \boldsymbol{p} \le \boldsymbol{v} \le \boldsymbol{x}, \qquad (4.21a)$$
$$\overline{\boldsymbol{u}}_{N_{h}}(t) \in \mathcal{U}_{N_{h}}(t,\boldsymbol{v})\},$$

where,

$$W_{N_{h}}(t, \boldsymbol{x}, \boldsymbol{v}, \overline{\boldsymbol{u}}_{N_{h}}(t)) = \sum_{k=0}^{N_{h}-1} \left( \boldsymbol{q}_{z}^{\mathrm{T}}(t+k)\overline{\boldsymbol{z}}(t+k) + \boldsymbol{q}_{u}^{\mathrm{T}}(t+k)\overline{\boldsymbol{u}}(t+k) \right)$$
(4.21b)
$$+ V_{h}(t+N_{h}, \overline{\boldsymbol{x}}(t+N_{h})),$$

$$\mathcal{U}_{N_{\mathsf{h}}}(t,\boldsymbol{\nu}) = \left\{ \boldsymbol{u}_{N_{\mathsf{h}}} \mid 0 \leq \overline{\boldsymbol{x}}(t+k) \leq \boldsymbol{x}_{\max} - \boldsymbol{p}, \ 0 \leq \overline{\boldsymbol{u}}(t+k) \leq \boldsymbol{u}_{\max} - \boldsymbol{K}\boldsymbol{p}, \ (4.21c) \\ \forall \ k \in \mathbb{N}_{0:N_{\mathsf{h}}-1}, \ \overline{\boldsymbol{x}}(t+N_{\mathsf{h}}) \in \overline{\mathbb{X}}_{\mathsf{f}} \right\},$$

$$\overline{\mathbf{x}}(t+k) = \mathbf{x}(t+k; \mathbf{v}, \overline{\mathbf{u}}_{N_{\rm h}}(t), \overline{\mathbf{\omega}}_{N_{\rm h}}(t)), \ k \in \mathbb{N}_{0:N_{\rm h}}, \tag{4.21d}$$

$$\overline{\mathbf{z}}(t+k) = \mathbf{z}(t+k; \boldsymbol{\nu}, \overline{\boldsymbol{u}}_{N_{h}}(t), \overline{\boldsymbol{\omega}}_{N_{h}}(t)), \ k \in \mathbb{N}_{0:N_{h}-1}.$$
(4.21e)

The solution to this new optimization problem, is the improved tube-base  $\overline{x}^*$  and the optimal control sequence  $\overline{u}_{N_h}^*(t, \overline{x}^*) = (\overline{u}^*(t; t, \overline{x}^*), ..., \overline{u}^*(t+N_h-1; t, \overline{x}^*))$ . The constraint  $x - p \le v \le x$  ensures that the first element of the tube 4.20 contains the current state of the uncertain system (i.e.,  $x \in \overline{x}^* \oplus S'$ ). Once again, the MPC control action at the current time step is the first element of the optimal control sequence; that is  $\overline{u}_{mpc}(t) := \overline{u}^*(t; t, \overline{x}^*)$ . The subsequent tube base  $\overline{x}^+$  is given by  $\overline{x}^+ = A\overline{x}^* + B_1(\tilde{\omega} + \overline{\omega}) + B_2 + \overline{u}_{mpc}$ .

#### **Integration of Controllers**

Given sets  $\mathbb{W} := \{ \boldsymbol{\omega} \ge 0 \mid \| \boldsymbol{\Omega}^{-1} \boldsymbol{\omega} \|_{\infty} \le 1, \ \boldsymbol{\Omega} = \text{diag}(\boldsymbol{\omega}_M), \ \boldsymbol{\omega}_M \in \mathbb{R}^{n_{\omega}}_{+,s} \}$  and  $\mathbb{S}_{\min} \subset \mathbb{R}^n_+$ , the set  $\mathbb{S}' := \{ \boldsymbol{e} \mid 0 \le \boldsymbol{e} \le \boldsymbol{p}, \ \boldsymbol{p} \in \mathbb{R}^n_{+,s} \}$  with  $\mathbb{S}_{\min} \subseteq \mathbb{S}'$ , and the associated feedback gain  $\boldsymbol{K}$  minimizing  $\| \boldsymbol{z}_{\mathbf{e}} \|_{L_{\infty},\mathbb{W},\mathbb{S}'}$  are calculated offline. At every time step, the optimization problem 4.21 is then solved and the control input  $\boldsymbol{u}(t) := \overline{\boldsymbol{u}}_{mpc}(t) + \boldsymbol{K}(\boldsymbol{x}(t) - \overline{\boldsymbol{x}}^*(t))$  is applied to the (uncertain) system.

#### **4.4.3.** Recursive Feasibility and Asymptotic Stability

The design parameters need to be chosen such that *asymptotic stability* of the closed-loop system under the control law (4.11) and *recursive feasibility* [10, Chapter 2] of the optimization problem (4.10) are guaranteed. Asymptotic stability of the nominal (i.e., without disturbance) closed-loop system is achieved by choosing

a strictly decreasing terminal cost function  $V_f(x): x \mapsto q_f(x)$  over a control invariant terminal constraint set. Recursive feasibility of the controlled system is a property requiring that  $x(t) \in \mathcal{X}_{N_{h}}(t)$  implies  $x(t+1) \in \mathcal{X}_{N_{h}}(t+1)$  for any  $t \in \mathbb{N}$ . It is well known that recursive feasibility in the presence of an unknown disturbance is achieved if the terminal constraint set  $X_f$  is robustly control invariant [2, 10, 20–22]. Calculating a (robustly) control invariant  $X_f$  in general is a non-trivial task. A classical approach, introduced in [3], gives a maximal robustly controlled invariant set as the fixed-point solution to a recursion over sets. However, only under certain conditions that depend on the system matrices (A, B) and the geometry of admissible state, and control, and disturbance sets [5, 6], the maximal robustly control invariant set can be determined in a finite number of iterations (i.e., in general the fixed-point might not exist). This approach is further adapted in [4] to ensure that the resulting set at every iteration is robust control invariant; hence an arbitrarily precise inner approximation of the maximal robust control set can be calculated. Nonetheless, practical application of such approaches is mostly limited to small-scale system with few states. For example, with polyhedral state and control constraints, each iteration involves operations such as Minkowski differences and polyhedral projections, which become quickly untractable for polyhedra of large dimensions.

We propose a linear-programing-based method for simultaneous calculation of the set  $X_f$  and the terminal cost function  $V_f(\cdot)$  over  $X_f$ . The set  $X_f$  obtained in this manner has the property that it is control invariant when  $\tilde{\boldsymbol{\omega}} = 0$  and is robustly control invariant otherwise. In addition  $V_f(\cdot)$  is strictly decreasing over  $X_f$  when  $\tilde{\boldsymbol{\omega}} = 0$ .

Proposition 4.3. Consider system 4.3 with the external input  $\boldsymbol{\omega}(t) \coloneqq \overline{\boldsymbol{\omega}}(t) + \tilde{\boldsymbol{\omega}}(t)$ , where  $\overline{\boldsymbol{\omega}}(t)$  is known and  $\tilde{\boldsymbol{\omega}}(t) \in \mathbb{W} \coloneqq \{\boldsymbol{\omega} \in \mathbb{R}^{n_{\omega}} \mid \|\boldsymbol{\Omega}^{-1}\boldsymbol{\omega}\|_{\infty} \leq 1, \ \boldsymbol{\Omega} \coloneqq \text{diag}(\boldsymbol{\omega}_M),$  $\boldsymbol{\omega}_M \in \mathbb{R}^{n_{\omega}}_{+,s}\}$  for all  $t \in \mathbb{N}$ . Using the terminal constraint set  $\mathbb{X}_f \coloneqq \{x \in \mathbb{R}^n \mid \|\boldsymbol{P}^{-1}x\|_{\infty} \leq 1, \ \boldsymbol{P} \coloneqq \text{diag}(\boldsymbol{p}), \ \boldsymbol{p} \in \mathbb{R}^{n}_{+,s}\}$ , the MPC problem 4.21 is recursively feasible if there exists a "feed forward" control input  $\overline{\boldsymbol{u}}(t)$ , a feedback gain K(t), and a scalar  $0 \leq \lambda < 1$  satisfying

$$A + B_2 K(t) \ge 0$$
, (4.22a)

$$(\mathbf{A} + \mathbf{B}_2 \mathbf{K}(t) - \lambda \mathbf{I}_n)\mathbf{p} + \mathbf{B}_1 \boldsymbol{\omega}_M < 0, \qquad (4.22b)$$

$$0 \le \mathbf{K}(t)\mathbf{p} \le \mathbf{u}_{\max} - \overline{\mathbf{u}}(t), \qquad (4.22c)$$

$$\begin{bmatrix} \boldsymbol{B}_{2}^{\mathrm{T}} & \boldsymbol{D}_{12}^{\mathrm{T}} \end{bmatrix}^{\mathrm{T}} \overline{\boldsymbol{u}}(t) = -\begin{bmatrix} \boldsymbol{B}_{1}^{\mathrm{T}} & \boldsymbol{D}_{11}^{\mathrm{T}} \end{bmatrix}^{\mathrm{T}} \overline{\boldsymbol{\omega}}(t), \qquad (4.22d)$$

for all  $t \in \mathbb{N}$ .

See Appendix 4.A for the proof.

*Proposition* 4.4. Suppose conditions of Proposition 4.3 are met for the set  $X_f := \{x \in \mathbb{R}^n \mid \|P^{-1}x\|_{\infty} \leq 1, P := \text{diag}(p), p \in \mathbb{R}^n_{+,s}\}$  corresponding to the control law  $u(t, x) := \overline{u}(t) + K(t)x(t)$ , and let  $r(\cdot) := \mathbb{N} \to \mathbb{R}_{+,s}$  be a non-increasing positive bounded function such that  $\lim_{t\to\infty} r(t) > 0$ . Then, with the terminal constraint set  $X_f$  and the terminal cost function  $V_f(t, x) := (t, x) \mapsto r(t) \|P^{-1}x\|_{\infty}$ , the origin is asymptotically stable for the nominal closed-loop system (i.e., with  $\tilde{\omega}(t) = 0$  for all

 $t \in \mathbb{N}$ ) under the MPC control law 4.11 if, in addition to 4.22, it holds that

$$C + D_{12}K(t) \ge 0$$
, (4.23a)

$$(\lambda - 1)r(t) \le -(\boldsymbol{q}_{z}^{\mathrm{T}}(t)(\boldsymbol{C} + \boldsymbol{D}_{12}\boldsymbol{K}(t)) + \boldsymbol{q}_{u}^{\mathrm{T}}(t)\boldsymbol{K}(t))\boldsymbol{p}, \qquad (4.23b)$$

for all  $t \in \mathbb{N}$ .

The proof is provided in Appendix 4.A. Proposition 4.3 and Proposition 4.4 do not automatically result in a finite number of inequalities as they require 4.22 and 4.23 to hold for all  $t \in \mathbb{N}$ . However, with  $\overline{\omega}(\cdot)$  being a periodic function of time such that  $\overline{\omega}(t + T) = \overline{\omega}(t)$  for some  $T \in \mathbb{N}$ , the following propositions, as restatements of Propositions 4.3 and 4.4 for the periodic case, result in a finite number of inequalities.

*Proposition* 4.5. Suppose  $q_z \colon \mathbb{N} \to \mathbb{R}^{n_z}_{+,s}$ ,  $q_u \colon \mathbb{N} \to \mathbb{R}^{n_u}_{+,s}$ , and  $\overline{\omega} \colon \mathbb{N} \to \mathbb{R}^{n_\omega}_{+,s}$  are given *T*-periodic functions of time with  $T \in \mathbb{N}$ , and  $0 \le \lambda < 1$  is a given scalar. Consider the nominal MPC problem 4.10 for the system 4.3 with terminal constraint set  $\mathbb{X}_f \coloneqq \{x \in \mathbb{R}^n \mid \|P^{-1}x\|_{\infty} \le 1\}$  and the terminal cost function  $V_f(x) \colon (t,x) \mapsto r\|P^{-1}x\|_{\infty}$ . Under the MPC control law 4.11, the origin is asymptotically stable for the closed-loop system with  $\omega(t) \coloneqq \overline{\omega}(t)$  and the optimization problem 4.10 is recursively feasible with  $\omega(t) \coloneqq \overline{\omega}(t) + \widetilde{\omega}(t)$  if the following set of linear (in)equalities in variables p,  $Y_0$ , ...,  $Y_{T-1}$ , r,  $\overline{u}(0)$ , ...,  $\overline{u}(T-1)$  is feasible for all  $i \in \mathbb{N}_{0:T-1}$ :

$$\begin{split} \boldsymbol{p} &> 0, \boldsymbol{P} = \operatorname{diag}(\boldsymbol{p}), \\ \boldsymbol{AP} + \boldsymbol{B}_{2}\boldsymbol{Y}_{i} &\geq 0, \\ & \left((\boldsymbol{A} - \lambda \boldsymbol{I}_{n})\boldsymbol{P} + \boldsymbol{B}_{2}\boldsymbol{Y}_{i}\right)\boldsymbol{1}_{n} + \boldsymbol{B}_{1}\boldsymbol{\omega}_{M} < 0, \\ & 0 \leq \boldsymbol{Y}_{i}\boldsymbol{1}_{n} \leq \boldsymbol{u}_{\max} - \overline{\boldsymbol{u}}(i), \\ & \left[\boldsymbol{B}_{2}^{\mathrm{T}} \quad \boldsymbol{D}_{12}^{\mathrm{T}}\right]^{\mathrm{T}}\overline{\boldsymbol{u}}(i) = -\left[\boldsymbol{B}_{1}^{\mathrm{T}} \quad \boldsymbol{D}_{11}^{\mathrm{T}}\right]^{\mathrm{T}}\overline{\boldsymbol{\omega}}(i), \\ & \boldsymbol{CP} + \boldsymbol{D}_{12}\boldsymbol{Y}_{i} \geq 0, \\ & r > 0, \\ & (\lambda - 1)r \leq -\boldsymbol{q}_{2}^{\mathrm{T}}(i)\boldsymbol{CP}\boldsymbol{1}_{n} - \left(\boldsymbol{q}_{2}^{\mathrm{T}}(i)\boldsymbol{D}_{12} + \boldsymbol{q}_{\mathrm{u}}^{\mathrm{T}}(i)\right)\boldsymbol{Y}_{i}\boldsymbol{1}_{n}. \end{split}$$

*Proposition* 4.6. Suppose  $\overline{\boldsymbol{\omega}} \colon \mathbb{N} \to \mathbb{R}_{+,s}^{n_{\omega}}$  is a given *T*-periodic function of time with  $T \in \mathbb{N}$ , and  $\boldsymbol{q}_{z} \colon \mathbb{N} \to \mathbb{R}_{+,s}^{n_{z}}$  and  $\boldsymbol{q}_{u} \colon \mathbb{N} \to \mathbb{R}_{+,s}^{n_{u}}$  are given bounded functions of time, and  $0 \leq \lambda < 1$  is a given scalar. Consider the nominal MPC problem 4.10 for the system 4.3 with terminal constraint set  $\mathbb{X}_{f} \coloneqq \{\boldsymbol{x} \in \mathbb{R}^{n} \mid \|\boldsymbol{P}^{-1}\boldsymbol{x}\|_{\infty} \leq 1\}$  and the terminal cost function  $V_{f}(\boldsymbol{x}) \colon (t, \boldsymbol{x}) \mapsto r \|\boldsymbol{P}^{-1}\boldsymbol{x}\|_{\infty}$ . Under the MPC control law 4.11, the origin is asymptotically stable for the closed-loop with  $\boldsymbol{\omega}(t) \coloneqq \overline{\boldsymbol{\omega}}(t)$  and the optimization problem 4.10 is recursively feasible with  $\boldsymbol{\omega}(t) \coloneqq \overline{\boldsymbol{\omega}}(t) + \tilde{\boldsymbol{\omega}}(t)$  if the following set of linear (in)equalities in variables  $\boldsymbol{p}, \boldsymbol{Y}, r, \overline{\boldsymbol{u}}(0), ..., \overline{\boldsymbol{u}}(T-1)$  is

feasible for all  $i \in \mathbb{N}_{0:T-1}$ :

| $oldsymbol{p} > 0$ , $oldsymbol{P} = diag(oldsymbol{p})$ , | (4.24a) |
|--|---------|
| AD + DV > 0  | (1.24h) |

$$AP + B_2 Y \ge 0, \qquad (4.24b)$$
$$((A - \lambda I_n)P + B_2 Y) 1_n + B_1 \omega_M < 0, \qquad (4.24c)$$

$$0 \le Y \mathbf{1}_n \le u_{\max} - \overline{u}(i), \tag{4.24d}$$

$$\begin{bmatrix} \boldsymbol{B}_{2}^{\mathrm{T}} & \boldsymbol{D}_{12}^{\mathrm{T}} \end{bmatrix}^{\mathrm{T}} \overline{\boldsymbol{u}}(i) = -\begin{bmatrix} \boldsymbol{B}_{1}^{\mathrm{T}} & \boldsymbol{D}_{11}^{\mathrm{T}} \end{bmatrix}^{\mathrm{T}} \overline{\boldsymbol{\omega}}(i), \qquad (4.24e)$$

$$CP + D_{12}Y \ge 0$$
, (4.24f)

$$r > 0$$
 ,

$$(\lambda - 1)r \le - \|\boldsymbol{q}_{z}\|_{L_{\infty}} \mathbf{1}_{n_{z}}^{\mathrm{T}} \boldsymbol{C} \boldsymbol{P} \mathbf{1}_{n} - (\|\boldsymbol{q}_{z}\|_{L_{\infty}} \mathbf{1}_{n_{z}}^{\mathrm{T}} \boldsymbol{D}_{12} + \|\boldsymbol{q}_{u}\|_{L_{\infty}} \mathbf{1}_{n_{u}}^{\mathrm{T}}) \boldsymbol{Y} \mathbf{1}_{n}.$$
(4.24h)

(4.24g)

# 4.5. Model Description of BHS

#### **4.5.1. Evolution of Link Density**

In this section, we present a modeling framework for BHSs that will be used in combination with the methods developed in Sections 4.3 and 4.4 to illustrate the application of those methods in designing a BHS control system. The DCV-based BHS network is symbolically modeled as a directed graph. This symbolic relation implies that not all components of DCV-based baggage handling system are present in its graph representation. Nonetheless, the most important components of the system for the control point of view, including the loading stations, which are entry points of pieces of baggage to the BHS network), the unloading stations, which are departure points of pieces of baggage from the network, the early baggage storage (EBS), which is an automated storage to temporarily store the early baggage, the central DCV storage (CDS), which is the parking location of empty DCVs, and the network junctions, which connect various parts of the network, are present in this model. Within this modeling framework, loading and unloading stations, network junctions, the CDS and the EBS are represented by links of the graph. The nodes of the graph simply re-distribute the inbound DCV flows from their upstream links over their downstream links.

A loading station  $i \in \{1, ..., N_0\}$  is represented by a pair of two distinct links  $(o_i, s_i)$ , where  $o_i$  is the origin link that transports pieces of baggage to the loading station, and where  $s_i$  is the source link that transports empty DCVs to the loading station from the CDS. It is assumed that the pieces of baggage are loaded onto empty DCVs, hence converting empty DCVs to loaded DCVs, at the downstream node of  $(o_i, s_i)$ . The set of all origin links and all source links are respectively denoted by  $O = \{o_1, ..., o_{N_0}\}$  and  $S = \{s_1, ..., s_{N_0}\}$ .

The CDS of the system is represented by a unique external link CDS, where empty DCVs, having left the unloading stations, are stored and from where they are dispatched to the source links.

An unloading station  $i \in \{1, ..., N_D\}$  is represented by a destination link  $d_i$ . Loaded DCVs are unloaded and converted to unloaded DCVs at the downstream node of  $d_i$ . The resulting empty DCVs are then transported to the CDS. The set of

4

all destination links is denoted by  $D = \{d_1, ..., d_{N_D}\}$ .

A network junction  $i \in \{1, ..., N_{net}\}$  is represented by a network link  $n_i$ , which (partially) connects an origin link to a destination link. The set of all network links is  $N = \{n_1, ..., n_{net}\}$ . The early baggage storage (EBS) of the system is represented by the graph link EBS  $\in N$ , where loaded DCVs can be temporarily stored.

Let  $L = 0 \cup S \cup N \cup D \cup \{CDS\}$  denote the set of all links of the graph. For a link  $d \in D$ , let  $P_d \subseteq L$  be the set of links that are on some directed path that include link d. In addition, for a link  $x \in L$ ,  $L_x^-$  and  $L_x^+$  respectively denote the set of incoming and outgoing links of x. For any  $x \in O \cup N$ ,  $d \in D$  and  $y \in N \cup D$ ,  $q_{x \to y,d}$  [DCV/s] is the link flow from link x to link y with destination d given by

$$q_{x \to y,d}(t) = \begin{cases} f_{x \to y,d}^{\dagger}(t), & \text{if } y \in P_d \cap L_x^+ \\ 0, & \text{otherwise.} \end{cases}$$
(4.25)

where  $f_{x \to y,d}^{l}$  [DCV/s] is the flow of loaded DCVs from link x to link y with destination d. Note that this definition implies that only loaded DCVs can flow from  $x \in O \cup N$  to  $y \in N \cup D$  and that the flow of loaded DCVs with destination index d is zero if y is not a downstream link to x or if d cannot be reached via y. For any  $d_1$  and  $d_2 \in D$ ,  $q_{d_1,d_2}^{out}$  [DCV/s] is the rate at which the DCVs destined for  $d_2$  leave destination link  $d_1$ , which is defined as

$$q_{d_1,d_2}^{\text{out}}(t) = \begin{cases} f_{d_1 \to \text{CDS}}^{\dagger}(t), & \text{if } d_1 = d_2 \\ 0, & \text{otherwise.} \end{cases}$$
(4.26)

where  $f_{d_1 \to \text{CDS}}^{!}(t)$  [DCV/s] is the rate at which loaded DCVs in destination link  $d_1$  are unloaded and dispatched to the central DCV storage CDS. This definition implies that only loaded DCVs destined for link  $d_1$  are unloaded in  $d_1$  and then shipped to the central DCV storage. For any  $o \in O$  and  $d \in D$ , the rate ([Bag/s]) at which pieces of baggage destined for d enter origin link o is defined as

$$q_{o,d}^{\text{in}}(t) = \begin{cases} f_{o,d}^{\text{l}}(t), & o \in P_d \\ 0, & \text{otherwise}. \end{cases}$$
(4.27)

where  $f_{o,d}^{\dagger}$  [Bag/s] is the destination-indexed baggage demand at the origin link o. For any  $s \in S$ , the flow [DCV/s] of empty DCVs to the source link s is given as

$$q_s^{\text{in}}(t) = f_{\text{CDS} \to s}^{\text{u}}(t), \qquad (4.28)$$

where  $f_{CDS\rightarrow s}^{u}(t)$  is the link-to-link flow [DCV/s] of empty (unloaded) DCVs from the central DCV storage to the source link *s*. In this modeling framework, as depicted in Fig 4.51, pieces of baggage flow into the origin links, where they are picked up by empty DCVs on the source links. Loaded DCVs then make their way through the network links toward the destination links, possibly after being stored in the EBS. Loaded DCVs are finally unloaded at the destination links, where they are dispatched to the central DCV storage.

Accumulation of loaded DCVs and pieces of baggage on links of the graph is modeled by a destination-index link density [DCV/m], whereas the link density of



Figure 4.51: Graph representation of BHS with loading stations stations  $LS_i$  and their unique upstream links  $o_i$ ,  $i = 1, ..., N_o$ , and with unloading stations stations  $US_i$  and their unique upstream links  $d_i$ ,  $i = 1, ..., N_D$ , and with EBS.

empty DCVs is non-destination-indexed. We assume that empty DCVs enter the source links from the central DCV storage and leave the graph network at the loading stations, where they are entirely *converted* to loaded DCVs by picking up the luggage accumulated on the origin links. Even though origin links technically carry luggage and loaded DCVs flow in the network links, we do not distinguish between luggage densities on origin links and loaded DCVs densities in the rest of the network, using the same notation for both. It is also assumed that loaded DCVs leave the network at destination links, *producing* empty DCVs. Hence, empty DCVs only exist on the central DCV storage and on the source links and their link densities need not to be modeled in the rest of the network.

For any  $o \in O$  and  $d \in D$ , the number of loaded DCVs on the origin link  $o \in O$  with destination  $d \in D$  and the corresponding link density are respectively given by

$$x_{o,d}^{\mathsf{l}}(t+1) = x_{o,d}^{\mathsf{l}}(t) + \Delta t \Big( q_{o,d}^{\mathsf{in}}(t) - F_{o,d}^{\mathsf{out}}(t) \Big),$$
(4.29)

$$\rho_{o,d}^{\mathsf{I}}(t) = \frac{1}{l_o} x_{o,d}^{\mathsf{I}}(t)$$
(4.30)

where  $\Delta t$  is discretization time step and  $F_{o,d}^{out}$  is the total outflow the link given by

$$F_{o,d}^{\text{out}}(t) = \sum_{r \in L_o^+} q_{o \to r,d}(t)$$
 (4.31)

The number of empty (unloaded) DCVs on any source link  $s \in S$  and the associated link density of empty DCVs are respectively given by

$$x_{s}^{u}(t+1) = x_{s}^{u}(t) + \Delta t \Big( q_{s}^{in}(t-t_{s}) - F_{s}^{out}(t) \Big), \qquad (4.32)$$

$$\rho_{s}^{u}(t) = \frac{1}{l_{s}} x_{s}^{u}(t), \qquad (4.33)$$

where the total outflow of the link is given by

$$F_s^{\text{out}}(t) = \sum_{d \in D} F_{o,d}^{\text{out}}(t), \qquad (4.34)$$

with o being the corresponding origin link to the source link s, (i.e., the pair (o, s) belong to the same loading station).

For any  $n \in N$  and  $d \in D$ , the number of loaded DCVs on link n with destination d and the corresponding link densities are respectively described by the following equation:

$$x_{n,d}^{\mathsf{I}}(t+1) = x_{n,d}^{\mathsf{I}}(t) + \Delta t \Big( \sum_{r \in L_n^-} q_{r \to n,d}(t-t_n) - \sum_{r \in L_n^+} q_{n \to r,d}(t) \Big),$$
(4.35)

$$\rho_{n,d}^{\mathsf{I}}(t) = \frac{1}{l_n} x_{n,d}^{\mathsf{I}}(t), \tag{4.36}$$

For any destination links  $d_1 \in D$  and  $d_2 \in D$ , the number of loaded DCVs at  $d_1$  with destination  $d_2$  and the corresponding density of loaded DCVs are given as:

$$x_{d_1,d_2}^{\mathsf{I}}(t+1) = x_{d_1,d_2}^{\mathsf{I}}(t) + \Delta t \Big( \sum_{r \in L_{d_1}^-} q_{r \to d_1,d_2}(t-t_{d_1}) - q_{d_1,d_2}^{\mathsf{out}}(t) \Big),$$
(4.37)

$$\rho_{d_1,d_2}^{\mathsf{I}}(t) = \frac{1}{l_{d_1}} x_{d_1,d_2}^{\mathsf{I}}(t), \tag{4.38}$$

In a similar manner, for the EBS and for any  $d \in D$ , the number of loaded DCVs with destination d stored in the storage and the corresponding density of loeade DCVs are respectively given as

$$x_{\text{EBS},d}^{\mathsf{I}}(t+1) = x_{\text{EBS},d}^{\mathsf{I}}(t) + \Delta t \Big(\sum_{r \in L_{\text{EBS}}^{-}} q_{r \to \text{EBS},d}(t-t_{\text{EBS}}) - \sum_{r \in L_{\text{EBS}}^{+}} q_{\text{EBS} \to r,d}(t)\Big),$$
(4.39)

$$\rho_{\text{EBS},d}^{\text{I}}(t) = \frac{1}{l_{\text{ebs}}} x_{\text{EBS},d}^{\text{I}}(t)$$
(4.40)

The density of empty DCVs in the central DCV storage is described by the following equations:

$$x_{\text{CDS}}^{u}(t+1) = x_{\text{CDS}}^{u}(t) + \Delta t \left( \sum_{d \in D} q_{d,d}^{\text{out}}(t-t_{\text{CDS}}) - q_{s}^{\text{in}}(t) \right),$$
(4.41)

$$\rho_{\rm CDS}^{\rm u}(t) = \frac{1}{l_{\rm CDS}} x_{\rm CDS}^{\rm u}(t) \,, \tag{4.42}$$

Note that (4.31) and (4.34), and (4.41) and (4.37) jointly guarantee that empty DCVs accumulate only in the central DCV storage and on the source links, and that loaded DCVs only accumulate on the rest of the network.

## 4.5.2. Inputs, Outputs and Constraints

The manipulated variables of the model are the DCV flows  $q_{r \to y,d}^{l}$  with  $r \in O \cup N$ ,  $d \in D$ ,  $o \in O \cap P_d$ ,  $y \in N \cup D$ , and  $q_{d_1,d_2}^{l}$  with  $d_1$ ,  $d_2 \in D$ , and  $q_s^{in}$  with  $s \in S$ . Since DCV flows are intrinsically non-negative, we limit the flows to non-negative values in the model. Moreover, we constrain the total outflow capacity of a link as follows:

$$0 \le \sum_{d} \sum_{y} q_{r \to y, d}(t) \le q_r^{\max}, \qquad r \in O \cup N, d \in D, y \in N \cup D$$
(4.43)

$$0 \le \sum_{d_2} q_{d_1, d_2}(t) \le q_{d_1}^{\max}, \qquad d_1 \in D, d_2 \in D$$
(4.44)

$$0 \le \sum_{s} q_s^{\text{in}}(t) \le q_{\text{CDS}}^{\max}, \qquad s \in S.$$
(4.45)

Model outputs are the link densities, which are intrinsically non-negative and bounded. The non-negativity constraint along with capacity constraints are included in the model as below:

$$0 \le \sum_{d \in D} \rho_{o,d}^{\mathsf{l}}(t) \le \rho_o^{\mathsf{l},\max}, \qquad o \in O, d \in D,$$
(4.46)

$$0 \le \sum_{d \in D} \rho_{n,d}^{\mathsf{I}}(t) \le \rho_n^{\mathsf{I},\max}, \qquad n \in N, d \in D$$
(4.47)

$$0 \le \rho_s^{\mathsf{u}}(t) \le \rho_s^{\mathsf{u},\max}, \qquad \qquad s \in S$$
(4.48)

$$0 \le \sum_{d_2 \in D} \rho_{d_1, d_2}^{l}(t) \le \rho_{d_1}^{l, \max}, \qquad d_1 \in D, d_2 \in D$$
(4.49)

$$0 \le \rho_{\text{CDS}}^{\text{u}}(t) \le \rho_{\text{CDS}}^{\text{u,max}},\tag{4.50}$$

$$0 \le \sum_{d \in D} \rho_{\mathsf{EBS},d}^{\mathsf{I}}(t) \le \rho_{\mathsf{EBS}}^{\mathsf{I},\max}.$$
(4.51)

The variable  $0 \le q_{o,d}^{\text{in}}(t) \le q_{o,d}^{\text{in}}$ ,  $(o, d) \in (O \times D)$  is the disturbance input, about which *partial* information is available in terms of its nominal and maximal values.

# 4.6. Case Study

The methods developed in this chapter are now deployed on a BHS network with two loading stations and two unloading stations, which is presented in Figure 4.61. A model in the form of 4.3 with n = 47 states describing DCV queue lengths,  $n_{\rm u} = 106$  control inputs representing link to link flows,  $n_{\rm d} = 848$  states representing delayed samples of control inputs<sup>1</sup>,  $n_{\rm w} = 4$  external inputs representing baggage demand, and  $n_{\rm z} = n$  monitored outputs representing link densities, is developed as

<sup>&</sup>lt;sup>1</sup>With link length of 64 [m] and DCV travel speed of 8 [m/s], a travel time of 8 samples is obtained for a sample time of  $\Delta t = 1$  [s].



Figure 4.61: Graph representation of BHS with loading stations (links 1 and 11), source links 27 and 28, unloading stations (links 18 and 23), EBS link 15, and the CDS (link 26). Solid lines are associated with bags/loaded DCVs and dashed lines correspond to empty DCVs. Each link is assumed to have a length of 64 meters with fixed DCV travel speed of 8 [m/s].

$$\begin{bmatrix} \mathbf{x}(t+1) \\ \mathbf{u}(t) \\ \mathbf{u}(t-1) \\ \vdots \\ \mathbf{u}(t-d+2) \\ \mathbf{u}(t-d+1) \end{bmatrix} = \begin{bmatrix} I_n & \Delta t A_{12} \\ 0 & 0 & \cdots & 0 \\ 0 & I & 0 & \ddots & \vdots \\ \vdots & 0 & \ddots & \ddots & 0 \\ 0 & \vdots & \ddots & I & 0 \\ 0 & 0 & \cdots & 0 & 0 \end{bmatrix} \begin{bmatrix} \mathbf{x}(t) \\ \mathbf{u}(t-d+1) \\ \mathbf{u}(t-d) \end{bmatrix} + \begin{bmatrix} \Delta t B_u \\ 0 \\ \vdots \\ 0 \end{bmatrix} \mathbf{u}(t) \quad (4.52a)$$
$$+ \begin{bmatrix} \Delta t B_\omega \\ 0 \\ \vdots \\ 0 \end{bmatrix} (\overline{\boldsymbol{\omega}}(t) + \tilde{\boldsymbol{\omega}}(t))$$
$$\mathbf{z}(t) = \begin{bmatrix} C_1 & 0 \end{bmatrix} \begin{bmatrix} \mathbf{x}(t) \\ \mathbf{u}(t-1) \\ \vdots \\ \mathbf{u}(t-d) \end{bmatrix}, \quad (4.52b)$$

where  $\Delta t = 1$  [s] is the model sample time and the system matrices  $A_{12} \ge 0$ ,  $B_u$ ,  $B_{\omega} \ge 0$ ,  $C_1 \ge 0$  are appropriately defined based on the modeling framework of Section 4.5.

### **4.6.1.** Configuring the MPC Controller

We impose a maximum link density of  $\rho_{max}^{l} = 0.8$  [DCV/m] on all links, and maximum link-to-link flow of 100 [DCV/min.], which corresponds to  $x_{\text{max}}^{\text{T}} = [481_n^{\text{T}} 1001_{n_d}^{\text{T}}]$ [DCV] and  $u_{\text{max}} = 100 \, 1_{n_{\text{II}}}$  [DCV/min.] in 4.10c. The MPC controller has a prediction horizon of  $N_{\rm h} = 25$  steps with control sample time  $\Delta t = 1$  [s]. The controller employs  $J = q_z^T(t)z(t) + 12q_u^T(t)u(t)$  as the stage cost in the MPC objective function 4.10b and 4.21b, where the weighting function  $q_z \coloneqq \mathbb{N} \to \mathbb{R}^{n_z}_{+,s}$  and  $q_u \coloneqq \mathbb{N} \to \mathbb{R}^{n_u}_{+,s}$ are depicted in Fig. 4.62. The weighting function  $q_{z}(\cdot)$  is designed such that baggage flow arriving at loading stations is first directed to the EBS and is later sent to a destination link as the corresponding unloading station opens. The control inputs weighting function  $q_{\parallel}(\cdot)$  incentivizes DCVs with destination d,  $d \in \{18, 23\}$ to arrive at the corresponding unloading station during its designated time window. In addition, it prevents circular free movements of DCVs in the network by making the loaded DCVs stay in the EBS for as long as possible. The controller assumes the planned baggage profile  $\overline{\omega}(\cdot)$  of Fig. 4.63 and unplanned baggage demand  $\tilde{\boldsymbol{\omega}}(t) \in \mathbb{W} \coloneqq \{\boldsymbol{\omega} \mid 0 \leq \boldsymbol{\omega}^{\mathrm{T}} \leq [4.5 \ 2.5 \ 3.5 \ 4]\}$  for all  $t \in \mathbb{N}$ . The terminal constraint set  $\mathbb{X}_{f} := \{x \geq 0 \mid \|P_{f}^{-1}x\|_{\infty} \leq 1, P_{f} = \text{diag}(p_{f})\}$  and the terminal cost function  $V_{\rm f}(x) \coloneqq r \| P_{\rm f}^{-1} x \|_{\infty}$  are calculated by solving the linear inequalities 4.24 of Proposition 4.6 for system 4.52 as  $p_{\rm f} = 10 \, 1_{n+n_{\rm d}}$  and r = 1250. For tube-based MPC, the solution to the linear program 4.6 of Theorem 4.2 with  $U = 10 I_{n_{11}}$ ,  $M = I_{n+n_{d}}$ , and  $Y_2 = 0$  yields  $p_1 Y_{11}$  and  $\gamma = 0.09$ . The state feedback controller is then obtained as  $K = Y(\text{diag}(p))^{-1}$ , which achieves  $||z_e||_{L_{\infty}, \mathbb{W}, \mathbb{X}_e} < 0.09$  for the error system 4.17 with bounded control effort  $0 \le Kx_e \le 10 \ 1_{n_u}$  for all  $x_e \in \mathbb{X}_e \coloneqq \{x \in \mathbb{R}^{n+n_d} \mid 0 \le x \le p\}$ .

The linear programs for determining the terminal cost and constraint set, and calculating the feedback gain K, and the ones associated with the MPC optimization problems 4.10 and 4.21 are solved using IBM ILOG CPLEX Optimization Studio (version 12.63) [23] connector for Matlab.



Figure 4.62: Weighting functions used in the MPC controller design;  $t_{EBS}(n, d)$  is the travel time from link *n* to destination *d* on the shortest path connecting the two, and  $t_0(d)$  and  $t_c(d)$ , respectively, mark the opening and closing time of destination link *d*. (a): link density weighting function  $q_z(\cdot)$  for links with loaded DCVs. (b): weighting function  $q_u(\cdot)$  for link-to-link flows.



Figure 4.63: Planned and disturbed baggage profile per destination at the loading stations. The opening and closing time of destination  $d \in \{18, 23\}$  are marked by  $t_0(d)$  and  $t_c(d)$ , respectively.



(a) Outflow of DCVs from CDS to the unload- (b) DCV accumulation in the EBS ing stations.

Figure 4.64: MPC control under planned baggage demand.

# 4.6.2. Simulation Results

To compare the performance of MPC, tube-based MPC (TB-MPC), and tube-based MPC with optimized tube base (Optimized TB-MPC) control strategies a series of simulations are conducted based on the morning baggage demand profile of Fig. 4.63, which depicts the nominal baggage demand profile and a disturbed profile. In all simulations, all empty DCVs are initially located at the CDS with no DCV being present on any other link of the BHS network. First, effectiveness of the MPC control strategy under the nominal baggage demand profile is tested. As observed in Fig. 4.64 and Fig. 4.65a, the MPC controller performs as expected in the sense that early baggage demand is directed to EBS, empty DCVs are optimally routed from the CDS to the unloading stations, and the DCV flows arrive at the designated destinations during their respective time window. Next, we consider a series of scenarios that correspond to disturbance sequences  $\tilde{\omega}_{t_{sim}}(i) = (\tilde{\omega}(i; 0), \ \tilde{\omega}(i; 1), \ \ldots, \ \tilde{\omega}(i; t_{sim})), \ i \in \mathbb{N}_{1:7}$ , where  $\tilde{\omega}_{t_{sim}}(i) \in \mathbb{W}^{t_{sim}}$  for all  $i \in \mathbb{N}_{1:7}$ 



(c) TB-MPC: disturbed demand profile scenario V.

(d) Optim. TB-MPC: disturbed demand profile scenario V.

Figure 4.65: MPC control under (disturbed) baggage demand. Flow of DCVs leaving the network at the unloading stations.

and  $t_{sim}$  is the simulation time. A qualitative comparison between the three control strategies is provided in Fig. 4.65 for the sample scenario V. For each scenario, performance of the MPC controller is compared against those of the TB-MPC and Optimized TB-MPC controllers, based on the following two criteria: i) the total relative baggage demand that misses the closing time of its designated destination of the time window defined as

$$\mathsf{loss}(i) = \frac{\sum_{d \in \{18,23\}} \sum_{t=t_c(d)}^{t_{sim}} q_{d,d}^{\mathsf{out}}(i;t)}{\sum_{t=0}^{t_{sim}} 1_{n_w}^{\mathsf{T}} \left( \tilde{\boldsymbol{\omega}}(i;t) + \overline{\boldsymbol{\omega}}(t) \right)}, \ i \in \mathbb{N}_{1:7},$$

and ii) the total closed loop cost defined as

$$J_{\mathsf{cl}}(i) = \sum_{t=0}^{t_{\mathsf{sim}}} \boldsymbol{q}_{\mathsf{z}}^{\mathsf{T}}(t) \boldsymbol{z}(i;t) + 12 \boldsymbol{q}_{\mathsf{u}}^{\mathsf{T}}(t) \boldsymbol{u}(i;t), \ i \in \mathbb{N}_{1:7}$$



(a) Optimal value of predicted cost over the (b) Current closed-loop cost. prediction horizon.

Figure 4.66: Comparison of MPC, TB-MPC and optimized TB-MPC approaches in terms of optimal prediction cost and current cost for the closed-loop system.

The results are summarized in Table 4.61, where we have also listed the relative total baggage demand for the *i*-th scenario, defined as  $\sum_{t=0}^{t_{sim}} 1_{n_w}{}^{\mathrm{T}} \tilde{\boldsymbol{\omega}}(i;t) / \sum_{t=0}^{t_{sim}} 1_{n_w}{}^{\mathrm{T}} \overline{\boldsymbol{\omega}}(t)$ . The total closed loop cost  $J_{cl}(i)$  of TB-MPC and TB-MPC Optimized approaches are normalized by that of the MPC approach for all  $i \in \mathbb{N}_{1:7}$ .

It it observed from Table 4.61 that as we deviate from the nominal scenario, the MPC control strategy performs the worst compared to the tube-based variants in terms of the loss measure. In addition, the difference between closed-loop performance of the tube-based methods and the MPC approach increases in significance for larger deviations from the nominal scenario. It is also evident that optimizing the tube base enhances the performance of the TB-MPC, but the difference is not as significant. The difference between the three methods can also be seen in Fig. 4.66, where the optimal prediction cost at the current state  $V_{N_h}(t, \mathbf{x}(t), \mathbf{u}_{N_h}^*(t))$  and the current closed-loop cost  $J_{cl}(t) = \mathbf{q}_{z}^{T}(t)\mathbf{z}(t) + 12\mathbf{q}_{u}^{T}(t)\mathbf{u}(t)$  are depicted.

# 4.7. Conclusions

For linear discrete time systems subject to an infinity-norm bounded additive disturbance  $\omega \in W$ , we have shown that the simultaneous problem of finding a robustly positively invariant set X and a (constrained) state feedback gain K that minimizes the  $||L||_{\infty}$  norm of the output over this set for all  $\omega \in W$  can be formulated as a linear program when K renders the closed-loop system positive. This solution is then leveraged in a tube-based MPC approach, where the feedback gain K renders the error system positive and the set X is used to characterize a bounding tube for trajectories of the uncertain system and to tighten the nominal MPC state and control input constraints. The feedback gain calculated by this approach has also the advantage that it ensures minimal deviation between the nominal and uncertain trajectories for a given control "budget" in terms of the maximal effort allowed by the feedback controller.

| Scenario     | Norm. demand [-] | Dorformanco critoria                   | Control Strategy |              |               |
|--------------|------------------|--|------------------|--------------|---------------|
|              |                  | Fenomialice citteria                   | MPC              | TB-MPC       | Optim. TB-MPC |
| Nominal      | 1.00             | loss [%]<br>norm. closed-loop cost [-] | 0<br>1.00        | -            | -             |
| Scenario I   | 1.13             | loss [%]<br>norm. closed-loop cost [-] | 0.13<br>1.00     | 0.06<br>0.88 | 0.00<br>0.79  |
| Scenario II  | 1.14             | loss [%]<br>norm. closed-loop cost [-] | 0.32<br>1.00     | 0.16<br>0.87 | 0.15<br>0.77  |
| Scenario III | 1.16             | loss [%]<br>norm. closed-loop cost [-] | 0.15<br>1.00     | 0.10<br>0.83 | 0.00<br>0.77  |
| Scenario IV  | 1.19             | loss [%]<br>norm. closed-loop cost [-] | 0.62<br>1.00     | 0.29<br>0.80 | 0.26<br>0.75  |
| Scenario V   | 1.22             | loss [%]<br>norm. closed-loop cost [-] | 0.69<br>1.00     | 0.30<br>0.75 | 0.28<br>0.72  |
| Scenario VI  | 1.25             | loss [%]<br>norm. closed-loop cost [-] | 3.03<br>1.00     | 0.56<br>0.78 | 0.50<br>0.70  |
| Scenario VII | 1.27             | loss [%]<br>norm. closed-loop cost [-] | 5.81<br>1.00     | 1.02<br>0.69 | 0.96<br>0.61  |

Table 4.61: Overview of simulation results for MPC, TB-MPC, and TB-MPC Optimized control strategies under various disturbance scenarios.

The proposed solution has also been used for calculating an infinity-norm bounded terminal constraint set and an infinity-norm based terminal cost function for the MPC and tube-based MPC approaches via linear programs to ensure recursive feasibility and (robust) asymptotic stability of the closed-loop system.

It has been shown via a case study that the proposed approach can be used to efficiently design (tube-based) MPC approaches for large-scale linear systems as all key components of our approach (i.e., calculating the feedback gain, the terminal constraint set and the terminal cost) and the MPC optimization problem are formulated as linear programs.

Finally, we would like to emphasize that the methods developed in Section 4.3 do not require the open-loop system to be positive; hence they are applicable to the wider class of linear systems.

# References

- [1] Y. Zeinaly and B. De Schutter, *Robustly positively invariant sets for discretetime linear positive systems: Application to a tube-based mpc approach,* (2022), to be submitted to IEEE Transactions on Control Systems Technology.
- [2] F. Blanchini, Set invariance in control, Automatica 35, 1747 (1999).
- [3] D. Bertsekas, Infinite time reachability of state-space regions by using feedback control, IEEE Transactions on Automatic Control 17, 604 (1972).
- [4] M. Rungger and P. Tabuada, Computing robust controlled invariant sets of linear systems, IEEE Transactions on Automatic Control 62, 3665 (2017).
- [5] S. V. Raković, E. C. Kerrigan, D. Q. Mayne, and J. Lygeros, *Reachability analysis of discrete-time systems with disturbances*, IEEE Transactions on Automatic Control **51**, 546 (2006).
- [6] E. C. Kerrigan and J. M. Maciejowski, Invariant sets for constrained nonlinear discrete-time systems with application to feasibility in model predictive control, in Proceedings of the 39th IEEE Conference on Decision and Control, Vol. 5 (2000) pp. 4951–4956.
- [7] A. Tarău, B. De Schutter, and H. Hellendoorn, *Model-based control for route choice in automated baggage handling systems*, IEEE Transactions on Systems, Man, and Cybernetics, Part C: Applications and Reviews 40, 341 (2010).
- [8] Y. Zeinaly, B. De Schutter, and H. Hellendoorn, An integrated model predictive scheme for baggage-handling systems: Routing, line balancing, and emptycart management, IEEE Transactions on Control Systems Technology 23, 1536 (2015).
- [9] Y. Zeinaly, B. De Schutter, and H. Hellendoorn, *Towards a robust multi-level* control approach for baggage handling systems, in 52nd IEEE Conference on Decision and Control (2013) pp. 281–287.
- [10] J. B. Rawlings, D. Q. Mayne, and M. M. Diehl, *Model Predictive Control: Theory, Computation, and Design*, 2nd ed. (Nob Hill Publishing., Madison, Wisconsin, 2018).
- [11] C. E. Garcia, D. M. Prett, and M. Morari, *Model predictive control: Theory and practice–A survey*, Automatica 25, 335 (1989).
- [12] H. Kiendl, J. Adamy, and P. Stelzner, Vector norms as lyapunov functions for linear systems, IEEE Transactions on Automatic Control 37, 839 (1992).
- [13] L. Farina and S. Rinaldi, *Positive Linear Systems: Theory and Applications* (John Wiley & Sons, Inc., 2000).
- [14] A. Berman, M. Neumann, and R. J. Stern, *Nonnegative Matrices in Dynamical Systems* (John Wiley & Sons, Inc., 1989).

- [15] A. Berman and R. J. Plemmons, *Nonnegative Matrices in the Mathematical Sciences* (Academic Press, 1979).
- [16] A. Rantzer, Scalable control of positive systems, European Journal of Control 24, 72 (2015), sI: ECC15.
- [17] C. Briat, Robust stability and stabilization of uncertain linear positive systems via integral linear constraints:  $L_1$ -gain and  $L_{\infty}$ -gain characterization, International Journal of Robust and Nonlinear Control **23**, 1932 (2013).
- [18] M. A. Rami and F. Tadeo, Controller synthesis for positive linear systems with bounded controls, IEEE Transactions on Circuits and Systems II: Express Briefs 54, 151 (2007).
- [19] I. Kolmanovsky and E. G. Gilbert, *Theory and computation of disturbance in-variant sets for discrete-time linear systems*, Mathematical Problems in Engineering 4, 317 (1998).
- [20] E. C. Kerrigan, Robust Constraint Satisfaction: Invariant Sets and Predictive Control, Ph.D. thesis, Control Group, Department of Engineering, University of Cambridge (2001).
- [21] E. C. Kerrigan and J. M. Maciejowski, Robust feasibility in model predictive control: necessary and sufficient conditions, in Proceedings of the 40th IEEE Conference on Decision and Control, Vol. 1 (2001) pp. 728–733.
- [22] E. C. Kerrigan and D. Q. Mayne, Optimal control of constrained, piecewise affine systems with bounded disturbances, in Proceedings of the 41st IEEE Conference on Decision and Control, 2002., Vol. 2 (2002) pp. 1552–1557.
- [23] D. M. Gay, IBM ILOG CPLEX Optimization Studio CPLEX User's Manual, International Business Machines Corporation (2016), Version 12, Release 8.

# Appendix

# **4.A. Proofs**

In this section we provide proofs of the propositions and theorems presented in the chapter. First, we present and prove the following proposition, which can be regarded as an extension of the Theorem 1.3.12 of [15] to system 4.1, that will be consequently used in the proof of Proposition 4.1.

*Proposition* 4.7. The following statements are equivalent for a positive system (4.1) with a set of states  $X := \{x \in \mathbb{R}^n \mid \|P^{-1}x\|_{\infty} \leq 1, P := \text{diag}(p), p \in \mathbb{R}^n_{+,s}\}$ , a disturbance set  $W := \{\omega \in \mathbb{R}^{n_{\omega}} \mid \|\Omega^{-1}\omega\|_{\infty} \leq 1, \Omega := \text{diag}(\omega_M), \omega_M \in \mathbb{R}^{n_{\omega}}_{+,s}\}$ , and a positive scalar  $\lambda \leq 1$ .

- i)  $Ap + B_{\omega}\omega_M < \lambda p$ .
- ii)  $\mathbb{X}^+ \subset \lambda \mathbb{X}$ , where  $\mathbb{X}^+ \coloneqq \{Ax + B_\omega \omega \mid \omega \in \mathbb{W}, x \in \mathbb{X}\}$ .
- iii) The function V : x → ||P<sup>-1</sup>x||<sub>∞</sub> is a Lyapunov function for the system over the set X̄ := ℝ<sup>n</sup> int(X), which is strictly decreasing along system trajectories in X̄; hence, the system state is globally ultimately bounded [2] in X.

*Proof.* (ii  $\Rightarrow$  i): Since  $A \ge 0$  and  $B_1 \ge 0$ , it holds that

$$\begin{aligned} x_{\max}^+ &\coloneqq \max_{x \in \mathbb{X}, \ \omega \in \mathbb{W}} Ax + B_1 \omega = Ap + B_1 \omega_M \\ x_{\min}^+ &\coloneqq \min_{x \in \mathbb{X}, \ \omega \in \mathbb{W}} Ax + B_1 \omega = -(Ap + B_1 \omega_M). \end{aligned}$$

Since, by assumption,  $x^+ \in int(\lambda X)$  for all  $x \in X$  and  $\omega \in W$ , it must then hold that

$$-\lambda \boldsymbol{p} < \boldsymbol{x}_{\max}^{+} < \lambda \boldsymbol{p} \tag{4.53a}$$

$$-\lambda \boldsymbol{p} < \boldsymbol{x}_{\min}^{+} < \lambda \boldsymbol{p}, \tag{4.53b}$$

which is equivalent to  $Ap + B_{\omega}\omega_M < \lambda p$ . Conversely,  $Ap + B_{\omega}\omega_M < \lambda p$  implies 4.53, which, in turn, implies that  $x^+ \in int(\lambda X)$  for all  $x \in X$  and  $\omega \in W$ .

(i  $\Rightarrow$  iii): assuming (i) holds, for any scalar  $c \ge 1$  it follows that  $Acp + B_1\omega_M < \lambda cp$  and that  $x^+ \in \operatorname{int}(\lambda cX)$  for all  $x \in cX$ , and all  $\omega \in W$ . Define  $X(c) \coloneqq \{x \in \mathbb{R} \mid \|P^{-1}x\|_{\infty} = c\}$  as the boundary of cX. For any  $x \in X(c) \subseteq cX$  it then holds that  $\|P^{-1}x\|_{\infty} < c\lambda$  for all  $\omega \in W$ . Hence,  $\|P^{-1}x^+\|_{\infty} - \|P^{-1}x\|_{\infty} < (\lambda - 1)c = (\lambda - 1)\|P^{-1}x\|_{\infty}$  for all  $\omega \in W$ . The proof is complete by noting that  $c \ge 1$  is chosen arbitrarily; so  $x \in \overline{X}$  implies that there exists a  $c \ge 1$  such that  $x \in X(c)$ .

Conversely,  $\|P^{-1}x^+\|_{\infty} < \lambda \|P^{-1}x\|_{\infty}$  for some<sup>2</sup>  $x \in \mathbb{X}(1)$  and a positive  $\lambda \leq 1$ , implies that  $-\lambda p < x^+ < \lambda p$  or, equivalently,  $x^+ \in \operatorname{int}(\lambda \mathbb{X})$  for all  $\omega \in \mathbb{W}$ , and, consequently,  $Ap + B_{\omega}\omega_M < \lambda p$ .

Proof of Proposition 4.1. For the positive system 4.1, equivalence of 4.5a to the set  $\mathbb{X}$  being robustly positively invariant is due to statement (i) of Proposition 4.7 with  $\lambda = 1$ . Note that the set  $\overline{\mathbb{X}}_0 := \left\{ x \in \mathbb{R}^n \mid \| \text{diag}([\| \mathbf{M}(:,1) \|_{\infty} \dots \| \mathbf{M}(:,n) \|_{\infty}]^T \right\}^{-1} x \|_{\infty} < 1$  is the smallest norm-infinity bounded set containing  $\mathbb{X}_0$ . Thus, 4.5c is equivalent to having  $\mathbb{X} := \{ x \mid \| \mathbf{P}^{-1} x \|_{\infty} \le 1 \}$  include  $\mathbb{X}_0$ . Finally, note that for  $\overline{\omega}_k := (\omega_M, \dots, \omega_M) \in \mathbb{W}^{k+1}$ , it follows that

$$z(t; \boldsymbol{p}, \overline{\boldsymbol{\omega}}_{k}) - \boldsymbol{C}\boldsymbol{p} - \boldsymbol{D}_{11}\boldsymbol{\omega}_{M} =$$

$$\begin{cases} 0 & t = 0, \ k \in \mathbb{N} \\ \boldsymbol{C}\left(\sum_{s=0}^{t-1} \boldsymbol{A}^{s}\right)(\boldsymbol{A} - \boldsymbol{I})\boldsymbol{p} + \boldsymbol{C}\left(\sum_{s=0}^{t-1} \boldsymbol{A}^{s}\right)\boldsymbol{B}_{1}\boldsymbol{\omega}_{M} & t \in \mathbb{N}_{0:k}, \ k \in \mathbb{N} \\ < -\boldsymbol{C}\left(\sum_{s=0}^{t-1} \boldsymbol{A}^{s}\right)\boldsymbol{B}_{1}\boldsymbol{\omega}_{M} + \boldsymbol{C}\left(\sum_{s=0}^{t-1} \boldsymbol{A}^{s}\right)\boldsymbol{B}_{1}\boldsymbol{\omega}_{M} = 0. \end{cases}$$

$$(4.54)$$

Hence, it holds that

$$-Cp - D_{11}\omega_M \le z(t; -p, -\overline{\omega}_k) \le z(t; x_0, \omega_k) \le z(t; p, \overline{\omega}_k) \le Cp + D_{11}\omega_M$$

for any  $x_0 \in \mathbb{X}$ ,  $k \in \mathbb{N}$ ,  $\omega_k \in \mathbb{W}^{k+1}$ , and for all  $t \in \mathbb{N}_{0:k}$ , with the equality holding for t = 0,  $x_0 = p$  ( $x_0 = -p$ ), and  $\omega_k = \overline{\omega}_k$  ( $\omega_k = -\overline{\omega}_k$ ). It then follows from the definition of  $||z||_{L_{\infty},\mathbb{W},\mathbb{X}}$  that  $||z||_{L_{\infty},\mathbb{W},\mathbb{X}} = ||Cp + D_{11}\omega_M||_{\infty}$ . Hence,  $Cp + D_{11}\omega_M < 1_{n_z}\gamma$  is equivalent to  $||z||_{L_{\infty},\mathbb{W},\mathbb{X}} < \gamma$ .

*Proof of Theorem 4.2.* The proof is constructed by applying Propositions 4.7 and 4.1 to the closed-loop system 4.4. Letting  $K = (Y_1 - Y_2)P^{-1}$ , constraint 4.6b is equivalent to the closed-loop system 4.4 being positive. Constraints 4.6c and 4.6d are then equivalent to 4.5 of Proposition 4.1, expressed in the closed-loop system matrices. For the closed-loop system, it then follows from Proposition 4.1 that  $\gamma$  is an upper bound to  $||z||_{L_{\infty}, W, X}$ . Equation 4.6e guarantees that the control input constraint  $||U^{-1}Kx||_{\infty} \leq 1$  is satisfied by all vertices of X and, thus, by any  $x \in X$ . Finally, the closed-loop system is globally ultimately bounded in X due to the last statement of Proposition 4.7.

Proof of Proposition 4.3. Define  $u(t, x) := \overline{u}(t) + K(t)x(t)$ , which satisfies the control constraint  $0 \le u(t) \le u_{\max}$  for all  $t \in \mathbb{N}$ . The resulting closed-loop system is

$$\begin{aligned} \mathbf{x}(t+1) &= (\mathbf{A} + \mathbf{B}_2 \mathbf{K}(t))\mathbf{x}(t) + \mathbf{B}_1 \tilde{\boldsymbol{\omega}}(t), \\ \mathbf{z}(t) &= (\mathbf{C} + \mathbf{D}_{12} \mathbf{K}(t))\mathbf{x}(t) + \mathbf{D}_{11} \tilde{\boldsymbol{\omega}}(t), \\ \Delta \mathbf{u}(t) &\coloneqq \mathbf{u}(t, x) - \overline{\mathbf{u}}(t) = \mathbf{K}(t)\mathbf{x}, \end{aligned}$$

<sup>&</sup>lt;sup>2</sup>Note that assuming the Lyapunov forward difference holds for  $x \in X(1)$  does not cause any loss of generality as for any  $y \in X(c)$ , with c being an arbitrary positive scalar,  $c^{-1}y \in X(1)$ .

for all  $t \in \mathbb{N}$ . Since 4.22a renders the closed-loop system positive for all  $t \in \mathbb{N}$ , it follows from 4.22b and Proposition 4.7 that the set  $\mathbb{X}_{f} := \{x \in \mathbb{R}^{n} \mid ||P^{-1}x||_{\infty} \leq 1, P := \text{diag}(p), p \in \mathbb{R}^{n}_{+,s}\}$  is robustly positively invariant for the closed-loop system. Therefore, it holds for system 4.3 that for all  $x(t) \in \mathbb{X}_{f}$ , there exists a feasible control input in the form of  $u(t, x) := \overline{u}(t) + K(t)x(t)$  such that  $x(t+1) \in \mathbb{X}_{f}$  for all  $\tilde{\omega} \in \mathbb{W}$ .

*Proof of Proposition 4.4.* First note that for any  $x(t) \in X_f$ , we have  $||P^{-1}x^+||_{\infty} - ||P^{-1}x||_{\infty} < (\lambda - 1)||P^{-1}x||_{\infty}$  and  $||P^{-1}x||_{p} \ge x$ . Assuming 4.23 holds, it then follows for any  $x(t) \in X_f$  that

$$\begin{split} V_{\mathsf{f}}(t+1, x(t+1)) - V_{\mathsf{f}}(t, x(t)) &= r(t+1) \| P^{-1} x(t+1) \|_{\infty} - r(t) \| P^{-1} x(t) \|_{\infty} < \\ (\lambda - 1) r(t) \| P^{-1} x(t) \|_{\infty} \leq \| P^{-1} x(t) \|_{\infty} (q_{\mathsf{Z}}^{\mathsf{T}}(t) (\mathcal{C} + \mathcal{D}_{12} K(t)) + q_{\mathsf{U}}^{\mathsf{T}}(t) K(t)) p \\ &\leq - (q_{\mathsf{Z}}^{\mathsf{T}}(t) (\mathcal{C} + \mathcal{D}_{12} K(t)) + q_{\mathsf{U}}^{\mathsf{T}}(t) K(t)) x(t) = - (q_{\mathsf{Z}}^{\mathsf{T}}(t) z(t) + q_{\mathsf{U}}^{\mathsf{T}}(t) \Delta u(t)). \end{split}$$

Hence, the time-varying version of "basic stability assumption" [10, Assumption 2.23 (a)] is satsified. In addition the time-varying version of basic stability assumption [10, Assumption 2.33 (b)] is automatically fulfilled as there exists a  $c_1 > 0$  such that  $q_z^{\mathrm{T}}(t)z + q_u^{\mathrm{T}}(t)\Delta u \ge c_1 \mathbf{1}_{n_z+n_u}^{\mathrm{T}} [z^{\mathrm{T}} \Delta u^{\mathrm{T}}]^{\mathrm{T}}$  with any positive bounded functions  $q_z(\cdot) \colon \mathbb{N} \to \mathbb{R}^{n_z}_{+,s}$  and  $q_u(\cdot) \colon \mathbb{N} \to \mathbb{R}^{n_u}_{+,s}$ , and  $V_{\mathrm{f}}(t,x) \le c_2 \|x\|_{\infty}$  with some scalar  $c_2 > 0$  for all  $t \in \mathbb{N}$  and for any  $x \in \mathbb{X}_{\mathrm{f}}$ . Therefore, due to [10, Theorem 2.39], the origin is asymptotically stable in  $\mathcal{X}_{N_{\mathrm{h}}}(t)$  at each  $t \in \mathbb{N}$  for the system  $x^+ = Ax + B_2(K(t) + \overline{u}(t)) + B_1\overline{\omega}(t)$ .

# 5

# Linear Positive Systems May Have a Reachable Subset from the Origin That is Either Polyhedral or Nonpolyhedral

"The beauty of mathematics only shows itself to more patient followers."

Maryam Mirzakhani, 1977 - 2017

Positive systems with positive inputs and positive outputs are used in several branches of engineering, biochemistry, and economics. Both control theory and system theory require the concept of reachability of a time-invariant discrete-time linear positive system. The subset of the state set that is reachable from the origin is therefore of interest. The reachable subset is in general a cone in the positive vector space of the positive real numbers. It is established in this chapter that the reachable subset can be either a polyhedral or a nonpolyhedral cone. For a single-input case, a characterization is provided of when the infinite-time and the finite-time reachable subset are polyhedral. An example is provided for which the reachable subset is nonpolyhedral. Finally, for the case of polyhedral reachable subset(s), a method is provided to verify if a target set can be reached from the origin using positive inputs.

The current chapter is based on [1].

# **5.1. Introduction**

## 5.1.1. Motivation and Scope

In this chapter, the focus is on the reachable subset from the origin of a singleinput time-invariant discrete-time linear positive system. It will be proven that such a reachable subset can be either a polyhedral or a nonpolyhedral cone. A characterization is provided of when this reachable subset is polyhedral.

A positive system may arise in many areas of science and of engineering, such as econometrics [2], bio-chemical reactors [3], compartmental systems [4, 5], and transportation system [6, 7],to name a few. The variables in such systems represent growth rates, concentration levels, mass accumulation, flows, etc. Obviously, variables of this nature can only assume values that are zero or strictly positive.

For problems of control and system theory with positive systems, a solid body of concepts, theorems, and algorithms has been developed. Of particular interest is the theory of linear positive systems [8], which is based on the theory of positive matrices and their geometric equivalent, polyhedral cones, [9-12].

While the theory of linear positive systems overlaps with the theory of linear systems, there are distinct differences between the two. Therefore, several concepts of linear systems cannot be directly generalized to linear positive systems without reformulation. One such property is the notion of reachability and controllability of a linear positive system.

The motivation of the investigation of reachability and controllability of a linear positive system is in 1) their use in control theory as an equivalent condition for the existence of a control law for particular control objectives; and 2) in the theory of realization and of system identification. In a positive system, as it arises in the research areas mentioned above, one may want to know whether from a specified initial state a particular terminal state can be reached by application of a positive input to the system. The state to be reached can be a set of concentrations of chemical substances in bio-reactor or a concentration in a compartment which e.g. a model of tissue in a human being. More generally, one may want to characterize all states of a linear positive system that can be reached from the zero initial state using positive inputs, which is also the object of interest for realization theory of linear positive systems. The choice for the reachable subset from the origin is essential for realization theory. Observability of a linear positive system is then of interest only for states in the reachable set. A characterization of that view of observability does currently not exist in the literature. The condition formulated in the paper [13] is too strong because it is based on the assumption that the reachable set from the origin is the entire positive vector space  $\mathbb{R}^{n}_{+}$ . Therefore, characterizing all states of a linear positive system that can be reached from the zero initial state using positive inputs is the problem to be investigated in this chapter. More details on the problem formulation may be found in Section 5.3.

# 5.1.2. Previous Work

Below the vector space of tuples of the positive real numbers will be referred to as the *positive vector space*; it is formally defined in Section 5.2.

Controllability and reachability of a discrete-time linear positive system has been widely studied and there is a considerable literature. This literature is briefly surveyed below. In most of the literature it is emphasized that the characterization of controllability of a discrete-time linear positive system takes a very different form than that of its counterpart for discrete-time linear systems [14–16]. In addition, while reachability of a linear system may be achieved in a number of steps equal to the state-space dimension, [17], for discrete-time linear positive systems this does not hold. For a linear positive system the number of steps required to reach a certain point in the positive orthant can be larger than the dimension of the system, as noted in [15], where this is illustrated using the model of a pharmacokinetic system.

The concept of reachability used in the literature of discrete-time linear positive systems is whether every state of the positive vector space can be reached from the origin either in finite time or in infinite time. The result is then a characterization of this considered concept of reachability. Publications that are based on that approach include [14, 18–21].

Reachability of a discrete-time linear positive system is characterized using a graph-theoretic approach, and canonical reachable or canonical controllable forms are derived as well in [14, 21]. The authors of [20] have established a link between positive state controllability and positive input controllability of a related system, which is then used to obtain a controllability criterion. A survey of results on controllability and reachability of positive systems is provided in [22, 23]. Controllability results for special classes of 1D and 2D positive systems are provided in [24].

It is worth mentioning that the constrained reachability problem for a discretetime linear system in the presence of disturbance with respect to a target tube or a target set has been widely discussed in the literature [25–28]. Among others [26] investigates this problem by constructing a sequence of target sets. The reachability problem is then transformed into a certain inclusion check on the last target set of this sequence. The authors of [26] also provide an approximate bounding ellipsoid algorithm to calculate the sequence of target sets and the associated input sequence. In [25], constrained reachability with respect to a target set is studied as a special case of reachability with respect to a target sets when these sets are known to be polyhedral. In the above-mentioned literature, checking reachability or controllability of a target set requires one to directly or indirectly construct certain modified target sets in an iterative manner. In addition, it is not known in advance whether a target set can be reached in finite time.

### **5.1.3.** Contribution of This Work

The contribution of this chapter to control and system theory is described next. Attention is restricted to a time-invariant discrete-time linear positive system. The problem for a continuous-time linear positive system is different. The results are mostly for a single-input system. The object of interest is the reachable subset from the origin state in either finite time or in infinite time. The problem is to characterize this reachable subset, in particular to determine whether the reachable subset is either a polyhedral cone or a nonpolyhedral cone. This problem is of interest to both control theory and to realization theory.

The problem considered in this chapter differs from the reachability or controllability problems treated in the literature. In the literature, the problem whether any state of the positive vector space can be reached by use of a positive input from the zero initial state has been investigated and a corresponding characterization of this concept has been provided. In this chapter the focus is on the characterization of the reachability subset which will often be a strict subset of the positive vector space. Moreover, it will be investigated whether the reachable subset is a polyhedral cone or a nonpolyhedral cone. In the existing literature the reachable subset has to equal the positive vector space which is a polyhedral cone. Surprisingly, as presented in this chapter, there exists an example of a linear positive system of which the reachable subset from the origin is a nonpolyhedral cone in the positive vector space. A consequence of this is that the reachable subset has to be investigated for the following cases: for a prespecified finite time, for an arbitrary finite time, and for infinite time. It will also be shown that the reachable subset can in general not be determined in a number of steps that equals the dimension of the state set but that the number of steps can be strictly larger than the dimension of the state set.

The specific contributions of the chapter are then as follows. A characterization of when the infinite-time reachable subset is a polyhedral cone, is provided in Theorem 5.13. A related result regarding the geometry of reachable sets for discrete-time linear systems (not necessarily positive) with positive control inputs can be found in [29]. A characterization of when the finite-time reachable subset from the origin is a polyhedral cone, is provided in Theorem 5.15. An example of linear positive system for which the reachable subset is nonpolyhedral is provided in Example 4. Results for the problem of when the reachable set contains a particular cone of terminal states are summarized in Proposition 5.18 and in Proposition 5.19.

The structure of the chapter is described next. Section 5.2 presents necessary background knowledge on positive matrices and positive systems. It also reports key terminology of controllability and reachability and links this to linear positive systems while highlighting existing view of the characterization of controllability and reachability of linear positive systems in the literature. Section 5.3 presents the approach of this chapter and the problem formulation. The characterization of the infinite-time reachable set as a polyhedral cone is provided in Section 5.4. The characterization of the finite-time reachable set as a polyhedral cone is provided in Section 5.5. Numerical verifiable conditions for the polyhedrality of the reachset in terms of the spectrum of the system matrix are provided also in those sections. Section 5.6 provides results on how to determine reachability for a specified control objective in the form of a subset of the positive vector space of a linear positive system.

# **5.2.** Preliminaries

**5.2.1. Positive Real Numbers, Positive Matrices, and Cones** The reader is assumed to be familiar with the integers, the real numbers, and vector spaces. Denote the set of the integers by  $\mathbb{Z}$ , the set of strictly-positive integers by  $\mathbb{Z}_+ = \{1, 2, ...\}$ , and the set of the natural numbers by  $\mathbb{N} = \{0, 1, 2, ...\}$ . For  $n \in \mathbb{Z}_+$ denote  $\mathbb{Z}_n = \{1, 2, ..., n\}$ .

The *real numbers* are denoted by  $\mathbb{R}$ , the set of the *positive real numbers* or the *positive numbers* by  $\mathbb{R}_+ = [0, \infty)$ , and the set of the *strictly-positive real numbers* by  $\mathbb{R}_{s+} = (0, \infty) \in \mathbb{R}_+$ . The term *positive real numbers* is preferred by the authors over the term *nonnegative real numbers* which occurs in the literature. The term *positive real numbers* is used in the book [30, p. 19].

Define the *positive vector space* of tuples of the positive real numbers as the tuple  $(\mathbb{R}_+, \mathbb{R}^n_+)$  with the algebraic operations described next. The set of the positive real numbers is closed with respect to addition and to multiplication. There does not exist an additive inverse while in the subset  $(0, \infty)$  there always exists a multiplicative inverse. The set of positive vectors  $\mathbb{R}^n_+$  is closed with respect to addition but there does not exist an additive inverse in this set. The vector of all-ones in  $\mathbb{R}^n_+$  is denoted by  $\mathbb{1}_n$ . When used without a subscript  $\mathbb{1}$  is a vector of appropriate dimension of which all elements are equal one.

For an integer  $m \in \mathbb{Z}_+$  and a set of positive vectors  $a_1, a_2, ..., a_m, a_i \in \mathbb{R}^n_+$  define in the positive vector space the set

$$\operatorname{conv}([a_1 \ \dots \ a_m]) = \{ x \in \mathbb{R}^n_+ \mid x = \sum_{i=1}^m \lambda_i a_i, \ \lambda_i \ge 0, \ i = 1, \dots, m, \ \sum_{i=1}^m \lambda_i = 1 \}$$
(5.1)

as the *convex polytope* generated by  $a_i$ , i = 1, ..., m.

Define in the vector space of the real numbers  $\mathbb{R}^n$  the *open ball* with center  $x \in \mathbb{R}^n$  and with radius  $r \in (0, \infty)$  as the set

$$B(\mathbf{x}, r) = \{ \mathbf{y} \in \mathbb{R}^n | \| \mathbf{y} - \mathbf{x} \|_2 < r \}.$$
(5.2)

The norm on  $\mathbb{R}^n$  is the Euclidean norm,  $||x||_2 = (\sum_{i=1}^n x_i^2)^{1/2}$ . This norm is also used on  $\mathbb{R}^n_+$ . An open ball in the positive vector space  $\mathbb{R}^n_+$  is defined in a similar manner with  $y \in \mathbb{R}$  replaced by  $y \in \mathbb{R}^n_+$  in (5.2).

A *positive matrix* A of size  $n \times m$  for  $n, m \in \mathbb{Z}_+$  is a matrix of which each element  $A_{i,j} = A_{ij}$  belongs to the positive real numbers  $\mathbb{R}_+$ . The set of such matrices is denoted by  $\mathbb{R}^{n \times m}_+$ .

The geometric view point of positive vectors is formulated in terms of rays and of cones as defined next. A *ray* is a half line  $Y \subset \mathbb{R}^n_+$  for  $n \in \mathbb{Z}_+$  described by a direction vector  $x \in \mathbb{R}^n_+ \setminus \{0\}$  such that for all  $c \in \mathbb{R}_+$ , *Y* contains all elements of the form  $c \cdot x \in Y$ . Equivalently,

$$\exists n \in \mathbb{Z}_+, \exists x \in \mathbb{R}^n_+ \setminus \{0\}, \quad C(x) = \{c \cdot x \in \mathbb{R}^n_+ \mid \forall c \in \mathbb{R}_+\}.$$

Below  $c \cdot x$  will be denoted by c x.

A cone is a nonempty subset  $C \subseteq \mathbb{R}^n_+$  such that (1) if  $x \in C$  and  $c \in \mathbb{R}_+$  then  $c \ x \in C$ ; and (2) if  $x, \ y \in C$  then  $x + y \in C$ . It follows that  $0 \in C$  for any cone C. By definition, a cone always includes the zero element of the positive vector space. That zero element is called the *apex* of the cone. Cones with an apex not at zero of the positive vector space are not used in this chapter.

A cone *C* is called a *polyhedral cone* if there exists an integer  $m \in \mathbb{Z}_+$  and a set of positive vectors  $a_1, a_2, ..., a_m \in C \subseteq \mathbb{R}^n_+$  such that, for any  $x \in C$  there exists positive real numbers  $y_i \in \mathbb{R}_+$  for i = 1, ..., m,  $x = \sum_{i=1}^m y_i a_i$ . Equivalently, *C* is a polyhedral cone if

$$\exists m \in \mathbb{Z}_{+}, \exists a_{1}, ..., a_{m} \in C, \text{ such that,}$$

$$C = \{x \in \mathbb{R}^{n}_{+} | \exists y \in \mathbb{R}^{m}_{+} \text{ such that } x = Ay\}, \text{ where,}$$

$$A = \begin{bmatrix} a_{1} & a_{2} & ... & a_{m} \end{bmatrix} \in \mathbb{R}^{n \times m}_{+}, y = \begin{bmatrix} y_{1} \\ y_{2} \\ \vdots \\ y_{m} \end{bmatrix}.$$

In the representation used above, the cone will also be denoted by

$$C = \operatorname{cone}([a_1 \dots a_m]) = \operatorname{cone}(A)$$
(5.3)

for the positive matrix  $A \in \mathbb{R}^{n \times m}_+$  with the understanding that the cone is generated by the columns of the matrix A. Moreover, with little abuse of the notation for  $A \in \mathbb{R}^{n \times m}_+$  and  $X \in \mathbb{R}^{n \times p}_+$ , the cone generated by stacking up the m + p columns of the matrices A and X will be denoted by  $C = \operatorname{cone}([A X])$ .

A cone is called a *nonpolyhedral cone* if it is not polyhedral. This implies that there does not exist a finite number  $m \in \mathbb{Z}_+$  as in the above definition. The term *round cone* could also be used in this case. An example of a round cone is the well known ice cream cone which may be found in [9, Ex. 1.2.2].

An example of a polyhedral cone is given by

$$C = \{ x \in \mathbb{R}^{4}_{+} | \exists y \in \mathbb{R}^{4}_{+} \text{ such that } x = Ay \},$$
  
with  $A = \begin{bmatrix} 1 & 0 & 0 & 1 \\ 1 & 1 & 0 & 0 \\ 0 & 1 & 1 & 0 \\ 0 & 0 & 1 & 1 \end{bmatrix}.$ 

A boundary ray of a cone *C* is a ray of the cone that lies on the boundary of the cone. A ray lies on the boundary of a cone if for every  $\epsilon \in (0, 1)$  sufficiently small and for every element *x* of the ray, the ball  $B(x, \epsilon)$  includes an element outside the cone.

It is called an *extreme (boundary) ray* of the cone if it cannot be written as the strict convex combination of two different rays. Thus  $x \in C$  is an extreme ray if there do not exist vectors y,  $z \in C$  that are boundary rays and a scalar  $c \in (0, 1)$  such that  $x = c \ y + (1 - c) \ z$ . In the above example, each of the columns of the matrix A is an extremal ray of cone cone(A).

More technical concepts and results regarding positive matrices may be found in Appendix 5.A because these are well known and not a contribution of this chapter.

The reader may find additional information on positive real numbers, positive matrices, and cones in the books [9, 31, 32].

#### **5.2.2. Linear Positive Systems**

**Definition 5.1.** Define a *discrete-time linear positive system* with *system matrix A* and *input matrix B* by the representation

$$\begin{aligned} \mathbf{x}(t+1) &= \mathbf{A}\mathbf{x}(t) + \mathbf{B}\mathbf{u}(t), \ t_0 \in \mathbb{N}, \ \mathbf{x}(t_0) = \mathbf{x}_0, \\ \mathbf{A} \in \mathbb{R}^{n \times n}_+, \ \mathbf{B} \in \mathbb{R}^{n \times m}_+, \ t \in T = \{t_0, \ t_0 + 1, \ t_0 + 2, ...\}, \\ \mathbf{x}_0 \in \mathbb{R}^n_+, \ \mathbf{u} : T \to \mathbb{R}^m_+, \ \mathbf{x} : T \to \mathbb{R}^n_+. \end{aligned}$$
 (5.4)

An explicit expression for the state function of a discrete-time linear positive system is well known and provided by the formula

$$\mathbf{x}(t) = \mathbf{A}^{t-t_0} \mathbf{x}_0 + \sum_{s=t_0}^{t-t_0-1} \mathbf{A}^s \mathbf{B} \mathbf{u}(t-1-s), \ \forall \ t \in T,$$
 (5.5)

$$(t_0, x_0) \stackrel{u(t_0:t-1)}{\mapsto} (t, x(t)), \text{ where}$$
(5.6)  
$$u(t_0:t-1) = (u(t_0), u(t_0+1), \dots, u(t-1)).$$

For a *time-invariant discrete-time linear positive system* we may assume  $t_0 = 0$  in Definition 5.1 and in the explicit solution (5.5) as the time axis can be shifted to the zero time without affecting the trajectories.

Definition 5.1 requires that the mathematical objects of the definition exist. An alternative definition, which may be found in the literature, defines a linear positive system as a linear system with as state space  $X = \mathbb{R}^n$  and requires that for any initial state  $x_0 \in \mathbb{R}^n_+$  and any positive input function  $u : T \to \mathbb{R}^m_+$ , the resulting state function x is such that for all  $t \in T$ ,  $x(t) \in \mathbb{R}^n_+$ . It can then be proven that this alternative definition leads to the condition that the matrices A and B are positive matrices. Thus the alternative definition leads back to the form of Definition 5.1.

Books on positive systems or books with chapters on positive systems include [8, 16, 24].

#### 5.2.3. Terminology of Controllability and Reachability

The literature of control and system theory is not standardized in regard to the terms controllability and reachability. The authors have chosen to use in this chapter the terms as introduced by R.E. Kalman in Chapter 2 of the book [33, Def. 2.13, Def. 2.14, p. 32]. Almost the same definitions may be found in [34, Def. 3.1.1]. Related papers of Kalman on controllability are [35, 36].

Consider the discrete-time linear system with the representation (5.4) and the corresponding solution (5.5). Associate with this system the *initial tuple*  $(t_0, x_0) \in T \times \mathbb{R}^n_+$  consisting of the initial time  $t_0$  and the initial state  $x_0$  where  $t_0$  will often be

taken to be zero,  $t_0 = 0$ , and the *terminal tuple*  $(t_1, x_1)$  consisting of the terminal time  $t_1$  and the terminal state  $x_1$  where  $t_1 \in T$  and  $x_1 = x(t_1)$ . The solution displayed above is then denoted as the *transition* 

$$(t_0, \boldsymbol{x}_0) \stackrel{\boldsymbol{u}(t_0:t_1-1)}{\mapsto} (t_1, \boldsymbol{x}(t_1)).$$

In system theory one often distinguishes between reachability and controllability: for reachability one considers an initial tuple consisting of an initial time and an initial state as fixed and one has to determine which tuples of a terminal time and a terminal state can be reached by the use of a positive input; for controllability one considers a terminal time and terminal state as fixed and one has to determine from which tuples of an initial time and an initial state one can reach the selected terminal state at the terminal time by the use of a positive input.

In case of a time-invariant system the concepts of reachability and of controllability do not depend on the initial time because the time axis can be shifted to the zero time without affecting the trajectories.

**Definition 5.2.** Consider a linear positive system as defined in Definition 5.1.

(a) Fix an initial tuple  $(t_0, x_0) \in T \times \mathbb{R}^n_+$ . The terminal tuple  $(t_1, x_1) \in T \times \mathbb{R}^n_+$  is called *reachable* from the initial tuple (i.e., can be *reached* from the initial tuple), if there exists a positive input  $u : \{t_0, t_0 + 1, ..., t_1 - 1\} \to \mathbb{R}^m_+$  such that the transition  $(t_0, x_0) \xrightarrow{u(t_0:t_1-1)} (t_1, x(t_1)) = (t_1, x_1)$  exists for this system. (R.E. Kalman states this for  $x_0 = 0$ .) The terminal tuple is called *reachable from the origin* if it is reachable from the initial tuple  $(t_0, 0) \in T \times \mathbb{R}^n_+$ .

Define the *reachable set* from  $(t_0, x_0) \in T \times \mathbb{R}^n_+$  as

$$\operatorname{Reachset}(t_0, \mathbf{x}_0) = \left\{ \begin{array}{l} \mathbf{x}_1 \in \mathbb{R}^n_+ | \exists t_1 \in T, \exists \mathbf{u} : \{t_0, \dots, t_1 - 1\} \to \mathbb{R}^m_+, \\ \operatorname{such that} (t_0, \mathbf{x}_0) \xrightarrow{\mathbf{u}(t_0; t_1 - 1)} (t_1, x_1) \end{array} \right\}.$$

(b) Fix a terminal tuple  $(t_1, x_1) \in T \times \mathbb{R}^n_+$ . The initial tuple  $(t_0, x_0) \in T \times \mathbb{R}^n_+$  is called *controllable* to the terminal tuple (i.e., can be *controlled* to the terminal tuple) if there exists an input  $u : \{t_0, t_0 + 1, ..., t_1 - 1\} \to \mathbb{R}^m_+$  such that the transition  $(t_0, x_0) \xrightarrow{u(t_0:t_1-1)} (t_1, x_1)$  exists for this system. (R.E. Kalman requires that the terminal state  $x_1 = 0$ .) The initial tuple is called *controllable to the origin* if it is controllable to the terminal tuple  $(t_1, 0) \in T \times \mathbb{R}^n_+$ .

Define the *controllable set* to the terminal tuple  $(t_1, x_1) \in T \times \mathbb{R}^n_+$  as

$$\mathsf{Conset}(t_1, \mathbf{x}_1) = \begin{cases} \mathbf{x}_0 \in \mathbb{R}^n_+ | \exists t_0 \in T, \exists \mathbf{u} : \{t_0, \dots, t_1 - 1\} \to \mathbb{R}^m_+, \\ \text{such that } (t_0, \mathbf{x}_0) \xrightarrow{\mathbf{u}(t_0: t_1 - 1)} (t_1, \mathbf{x}_1) \end{cases} \end{cases}.$$

For linear systems, not necessarily a linear positive system, the following result holds.

*Lemma* 5.3. [34, Lemma 3.1.5] Consider a time-invariant discrete-time linear system (not necessarily a linear positive system). The system is a reachable system on the interval  $\{t_0, ..., t_1\}$ , if and only if it is reachable from the origin on the same interval.

The above result does not hold for linear positive systems as the following example shows.

Example 1. Consider the time-invariant linear positive system

$$\mathbf{x}(t+1) = \begin{pmatrix} 1/2 & 0 \\ 0 & 1 \end{pmatrix} \mathbf{x}(t) + \begin{pmatrix} 2 & 0 \\ 0 & 1 \end{pmatrix} \mathbf{u}(t), \ \mathbf{x}(0) = \mathbf{x}_0.$$

Then the reachable set from the origin is the full positive vector space  $\mathbb{R}^2_+$ . If  $x_0 = (1, 1)^T$  then the reachable set from that initial state equals

 $X(\mathbf{x}_0) = \{\mathbf{x} \in \mathbb{R}^2_+ | x_1 \ge 0.5 x_{0,1} = 0.5, x_2 \ge x_{0,2} = 1\}.$ 

Hence the state  $x_1 = (0.4, 0.4)^T$  can never be reached from  $x_0$  using positive inputs. Thus reachability from the origin and from an arbitrary initial state of the positive vector space are different concepts for linear positive systems.

From the above example it is clear that the reachable set from the origin and the reachable set from an arbitrary initial state are different objects. In this chapter attention is restricted to the reachable set from the origin.

# 5.2.4. Existing Results on Reachability and Controllability of Linear Positive Systems

The existing view of the characterization of controllability and reachability as known in the literature, is discussed below. In most papers of the literature, the characterization of controllability or of reachability of a linear positive system is based on the following definition.

**Definition 5.4.** [16, p. 74, Def. 7]. A linear positive system is said to be completely reachable if all states  $x \ge 0$  are reachable in finite time from the origin, that is, if  $X_r = \mathbb{R}^n_+$ , where  $X_r$  denotes the cone of all reachable states in finite time using a positive input.

The underlying idea behind Definition 5.4 probably originates from making an analogy to reachability of linear systems. This definition is based on the assumption that the state space equals  $X = \mathbb{R}^n$ . Note that in Definition 5.4  $X_r \subseteq \mathbb{R}^n_+$  by definition, hence the equality  $X_r = \mathbb{R}^n_+$  holds if in addition  $\mathbb{R}^n_+ \subseteq X_r$ . The following theorem states a necessary and sufficient condition for reachability with respect to Definition 5.4 for the single-input case.

**Theorem 5.5.** [16, Th. 27]. A discrete-time linear positive system with a singleinput is completely reachable if it is possible to reorder its state variables in such a way that the input u directly influences only  $x_1$ , and  $x_i$  directly influences  $x_{i+1}$  for i = 1, 2, ..., n - 1.


Figure 5.21: Example 2. The shaded area, associated with K, represents the region of interest for which controllability needs to be checked.

Additional results may be found in [16, Ch. 8]. The criterion for complete reachability of a linear positive system with multiple inputs based on Definition 5.4 is more involved, but it is required that the controllability matrix of the corresponding linear system, [**B** AB ...  $A^k$ B], includes a monomial submatrix of dimension *n*, for some  $k \in \mathbb{N}_+$  [14, 15, 19, 21, 22]. Such conditions are often too strong to be satisfied by most practical linear positive systems.

For several examples of linear positive systems, complete reachability as of Definition 5.4 is not required. For example in economic systems, one would be interested to know whether a certain growth rate can be achieved, which corresponds to checking whether a certain extremal ray of a cone inside the positive vector space is reachable. In bio-chemical reactors, it may be of interest to know whether a set of desired mass concentrations can be reached by applying a particular input (for example, a flow of materials).

An example follows that illustrates the concept of reachability stated above.

Example 2. Consider the discrete-time time-invariant linear positive system

$$x(t+1) = Ax(t) + bu(t), x(0) = x_0,$$

with

$$\boldsymbol{A} = \begin{bmatrix} 4 & 4 \\ 11 & 2 \end{bmatrix}, \ \boldsymbol{b} = \begin{bmatrix} 2 \\ 1 \end{bmatrix}, \ \boldsymbol{x}_0 = 0.$$

It is of interest to determine whether the states in the cone  $K \subset \mathbb{R}^2_+$ , defined by (5.7) and illustrated by Figure 5.21, can be reached in finite time:

$$K : \begin{cases} 3x_1 - 2x_2 \ge 0, \\ 3x_2 - 2x_1 \ge 0, \\ x_1 \ge 0, x_2 \ge 0. \end{cases}$$
(5.7)

Since  $K \subset \mathbb{R}^2_+$ , in order to answer this question using the classical approach, one needs to check the reachability of  $\mathbb{R}^2_+$ , which is very conservative considering the fact that *K* occupies only a small portion of  $\mathbb{R}^2_+$ . It can be verified that

$$\begin{bmatrix} \boldsymbol{b} & \boldsymbol{A}\boldsymbol{b} & \dots & \boldsymbol{A}^{k}\boldsymbol{b} \end{bmatrix} = \begin{bmatrix} 2 & 12 & \cdots \\ 1 & 24 & \cdots \end{bmatrix}$$

does not include a monomial submatrix of dimension 2 for any  $k \in \mathbb{N}_+$ . Therefore, the conditions of Theorem 5.5 do not hold and we cannot deduce anything about the reachability of *K*. Nevertheless, invoking Theorem 5.15 and using the results of Section 5.6, it turns out that K is reachable from the origin in a finite number of steps.

# **5.3. Approach of This Chapter**

The chapter changes the focus of reachability of a linear positive system. In the classical literature the system is reachable from the origin if the reach set from the origin equals the entire positive vector space  $\mathbb{R}^{n}_{+}$ .

In this chapter, the approach is to determine the reachable set from the origin, in either finite time or in infinite time, as defined below. The reachable set is then the main object of study. In this chapter, there is no requirement that the reachable set from the origin equals the positive vector space  $\mathbb{R}^n_+$ .

In the late 1960s and the 1970s the geometric view point gained momentum in control and system theory. This viewpoint was developed by W.M. Wonham, [37], for time-invariant linear control systems using the concept of a linear subspace of a vector space. The geometric approach to control of nonlinear control systems was described in the book [38]. Later this led to the development of control theory in differential-geometric structures, [39, 40], and in algebraic-geometric structures such as rings [41].

In the geometric approach to control systems the main concept is the reachable set from the origin. In the context of observability, it is the kernel of the output map, but that will not be treated in this chapter. For linear positive systems, the main geometric concept is a cone in the positive vector space  $\mathbb{R}^n_+$ . This geometric object allows the use of abstract algebra for theory and algorithms. Therefore, in this chapter the geometric approach to linear positive systems is used.

Based on this new view point, the system theoretic problem under study is: Characterize the reachable set from the origin of a linear positive system. The reachable set from the origin is by definition a cone in the positive vector space. A question is then: Is the reachable set from the origin a polyhedral cone or a nonpolyhedral cone?

*Remark* 3. The above formulation has been for decades the approach to reachability in system theory. The reachable set from the origin is defined as stated above. The reachable set in general may be a strict subset of the ambient space in which it is situated. The reader may want to look at the definitions of the reachable subset for discrete-time polynomial systems, [41], for continuous-time polynomial systems, [42], rational systems, [43], and infinite-dimensional linear systems, [44].

#### 5.3.1. Concepts

The reachable set and its role in the problem of reachability and of controllability of linear positive systems have been already discussed in the literature [14, 15, 19, 21, 22]. Below the concept inspired by [15] is used. Recall that only reachability from the origin, the zero initial state, is considered and that the system is restricted to

have an input with only one component. Recall the formula of the state transition of a time-invariant discrete-time linear positive system as

$$\begin{aligned} \mathbf{x}(t) &= \sum_{s=0}^{t-1} A^s \ \mathbf{b} \ u(t-1-s) \\ &= \left[ \begin{array}{cc} \mathbf{b} & A\mathbf{b} & \dots & A^{t-1}\mathbf{b} \end{array} \right] \left[ \begin{array}{c} u(t-1) \\ u(t-2) \\ \vdots \\ u(0) \end{array} \right] \end{aligned}$$
 (5.8)

with conmat<sub>k</sub>(A, b) =  $\begin{bmatrix} b & Ab & \dots & A^{k-1}b \end{bmatrix}$  being the *controllability matrix of in*dex k.

It is useful to have notation for the infinite reachable set and to contrast that with the finite reachable set, which is the purpose of the following definition.

**Definition 5.6.** Consider a single-input time-invariant discrete-time linear positive system with representation

$$x(t+1) = Ax(t) + bu(t), x(0) = 0.$$
 (5.9)

Define the following subsets of the state space: the *k*-step reachable subset from the origin, the finite-time reachable subset from the origin, and the infinite-time reachable subset from the origin, respectively as the sets,

$$\mathsf{Reachset}_{k}(A, b) = \left\{ x \in \mathbb{R}^{n}_{+} | \exists u : \mathbb{N}^{k-1} \to \mathbb{R}_{+}, (0, 0) \stackrel{u}{\mapsto} (k, x) \right\}, \forall k \in \mathbb{Z}_{+}, (5.10)$$

$$\operatorname{Reachset}_{f}(A, b) = \bigcup_{k=0}^{\infty} \operatorname{Reachset}_{k}(A, b), \qquad (5.11)$$

$$\operatorname{Reachset}_{\infty}(A, b) = \operatorname{Reachset}_{f}(A, b).$$
(5.12)

Here, the notation  $\overline{S}$  denotes the closure of the set *S* with respect to the Euclidean topology.

The reachable subsets defined above are subsets of the state set. To simplify the terminology, in the remainder of the chapter these sets are referred to as the *reachable set from the origin* or as the *reachable set*, without the use of the term *subset*.

Once a reachable set has been defined, there is no need for the concept of complete reachability.

*Proposition* 5.7. The k-step reachable subset, the finite-time reachable subset, and the infinite-time reachable subset of Definition 5.6, each from the zero initial state, equal respectively the expressions

$$\operatorname{Reachset}_{k}(A, b) = \operatorname{cone}(\operatorname{conmat}_{k}(A, b)), \qquad (5.13)$$

$$\operatorname{Reachset}_{f}(A, b) = \operatorname{cone}([b \ Ab \ A^{2}b \ \dots]), \qquad (5.14)$$

Reachset<sub>$$\infty$$</sub>(*A*, *b*) = Reachset<sub>f</sub>(*A*, *b*), where (5.15)

conmat<sub>k</sub>(A, b) = [b Ab 
$$A^2b \dots A^{k-1}b$$
] (5.16)

*Proof.* The proof is skipped as it can be derived in a straightforward manner. The reader is referred to [22, 45] for similar proofs. The proof could also be deduced from the corresponding definition in [15].

#### **5.3.2.** Problem Formulation

Having characterized the infinite-time and the finite-time reachable sets from the origin, the main questions of this chapter are discussed next.

*Problem* 5.8. For a single-input time-invariant linear positive system, the problems to be addressed in this chapter are:

- (a) Is the finite-time reachable set from the origin Reachset<sub>f</sub>(*A*, *b*) a polyhedral cone or a nonpolyhedral cone?
- (b) Is the infinite-time reachable set from the origin  $\text{Reachset}_{\infty}(A, b)$  a polyhedral cone or a nonpolyhedral cone?
- (c) If the control objective is specified as a cone in the positive vector space or as a subset of that space, is that control objective subset then contained in the reachable set from the origin?

Note that the *k*-time reachable set is by definition always a polyhedral set.

# 5.4. When Is the Infinite-Time Reachable Set a Polyhedral Set?

In this section, we investigate the polyhedrality of  $\text{Reachset}_{\infty}(A, b)$ , and characterize this in terms of a necessary and sufficient conditions on the system matrix A.

The reader is expected to have knowledge of concepts and of results of positive linear algebra as summarized in Appendix 5.A. The notations used below may be found in Appendix 5.A.

As summarized in Appendix 5.A, a positive matrix which is nonzero and of dimension  $n \ge 2$  is either irreducible or can be fully reduced. The analysis of the matrix  $A^k$  for  $k \in \mathbb{Z}_+$  or for its limit,  $\lim_{k\to\infty} A^k$ , can then be carried out (1) for irreducible positive matrices and, (2) for fully reduced matrices. Below the case of an irreducible system matrix A is carried out. The case of a fully reduced positive matrix is then relatively simple based on the results for the irreducible case [9].

For the remainder of this section, the reader should keep in mind the restriction to an irreducible positive matrix  $A \in \mathbb{R}^{n \times n}_+$ .

*Proposition* 5.9. Consider the linear positive system given in (5.4). Assume that  $A \in \mathbb{R}^{n \times n}_+$  is irreducible with cyclicity index  $1 \le h \le n$  and  $b \in \mathbb{R}^n_+$ . Then, the infinite-time reachable set from the origin,  $\text{Reachset}_{\infty}(A, b)$ , is polyhedral if and only if there exists a  $k^* \in \mathbb{Z}_+$  such that

$$A^{k^*}b \in \operatorname{cone}([b \ Ab \ \dots \ A^{k^*-1}b \ A_{f,0}b \ \dots \ A_{f,h-1}b]),$$
 (5.17)

where matrices  $A_{f,i}$  are introduced in Definition 5.24.

99

*Proof.* The result is almost obvious by geometric considerations except for the presence of the set of vectors  $\{A_{f_0}b, \dots, A_{f_{h-1}}b\}$ .

Sufficiency: We will show that

$$C = \operatorname{cone}\left( [\boldsymbol{b} \ \boldsymbol{A} \boldsymbol{b} \ \dots \ \boldsymbol{A}^{k^*-1} \boldsymbol{b} \ \boldsymbol{A}_{\mathsf{f},0} \boldsymbol{b} \ \dots \ \boldsymbol{A}_{\mathsf{f},h-1} \boldsymbol{b} ] \right)$$

is *A*-invariant. Let  $x = \sum_{i=0}^{k^*-1} c_i A^i b + \sum_{i=0}^{h-1} c_{f,i} A_{f,i} b$  for arbitrary positive coefficients  $c \in \mathbb{R}^{k^*}_+$  and  $c_f \in \mathbb{R}^{h}_+$ . We then have

$$Ax = \sum_{i=0}^{k^*-1} c_i A^{i+1} b + \sum_{i=0}^{h-1} c_{f,i} A A_{f,i} b.$$
 (5.18)

Using (5.17), and noting that (see Definition 5.24)

$$AA_{f,i} = A_{f,i+1}, \quad i = 0, ..., h-2$$

$$AA_{f,h-1} = (\rho(A))^h A_{f,0},$$
(5.19)

(5.18) can be expressed as  $Ax = \sum_{i=0}^{k^*-1} c'_i A^i b + \sum_{i=0}^{h-1} c'_{f,i} A_{f,i} b$  for some  $c' \in \mathbb{R}^{k^*}_+$ and some  $c'_{f,i} \in \mathbb{R}^h_+$ . This proves that  $Ax \in C$  for any  $x \in C$ . Hence, the system trajectory (5.8) remains in C and Reachset<sub> $\infty$ </sub>(A, b) = C is polyhedral.

Necessity: Let  $x_{\infty} = \lim_{k \to \infty} \frac{A^k b}{(\rho(A))^k}$ . Note that  $x_{\infty}$  is characterized by the set of *h* vectors  $A_{f,0}b, \dots, A_{f,h-1}b$  [15, Th. 2](also see proof of Lemma 5.25.) In

fact, Lemma 5.25 states that  $x_{\infty} \in \operatorname{cone}([A_{f,0}b \dots A_{f,h-1}b])$ . By the definition of Reachset<sub> $\infty$ </sub>(A, b) as the closure of Reachset<sub>f</sub>(A, b), and by the above explanation of  $x_{\infty}$ , the extremal rays of the polyhedral

Reachset<sub> $\infty$ </sub>(*A*, *b*) belong to the sequence { $A^k b \in \mathbb{R}^n_+$ ,  $k \in \mathbb{N}$ } or are extremal rays of the cone, cone( $[A_{f,0}b \dots A_{f,h-1}b]$ ). Again, by the assumption that Reachset<sub> $\infty$ </sub>(A, b) is polyhedral, there exists a finite  $k^* \in \mathbb{Z}_+$  such that  $\square$ 

$$A^{\kappa} b \in \text{cone}([b \ ... \ A^{\kappa-1}b \ A_{f,0}b \ ... \ A_{f,h-1}b]).$$

It is clear that if (5.17) is established for an integer  $k^* \in \mathbb{Z}_+$ , it will hold for any  $k \ge k^*$ . The smallest integer  $k^* \in \mathbb{Z}_+$  satisfying (5.17) is called the *vertex number* and denoted by  $k_{vert}^{\infty}$  of the reachable set Reachset<sub> $\infty$ </sub>(*A*, *b*). Following the steps of the proof of Proposition 5.9, we can put forward the following corollary.

*Corollary* 5.10. Given  $A \in \mathbb{R}^{n \times n}_+$  irreducible with cyclicity index  $h \in \{1, ..., n\}$  and  $\boldsymbol{b} \in \mathbb{R}^{n}_{+}$ , the following statements are equivalent:

- (a) Reachset<sub> $\infty$ </sub>(A, b) is polyhedral.
- (b) There exists an integer  $k_{vert}^{\infty} \in \mathbb{Z}_+$  such that cone([**b** Ab ...  $A^{k-1}b A_{f,0}b$  ...  $A_{f,h-1}b$ ]) is A-invariant for  $k \ge k_{vert}^{\infty}$ .
- (c) There exists an integer  $k_{\text{vert}}^{\infty} \in \mathbb{Z}_+$  such that for all  $k \ge k_{\text{vert}}^{\infty}$ , the matrix equation

$$AM = MX$$
, has a solution  $X \in \mathbb{R}^{(k+h) \times (k+h)}_+$ , where  
 $M = \begin{bmatrix} b & Ab & \dots & A^{k-1}b & A_{f,0}b & \dots & A_{f,h-1}b \end{bmatrix}$ .

**Definition 5.11.** A square positive matrix  $A \in \mathbb{R}^{n \times n}_+$  is said to have a *positive recursion* if the following holds:

$$\exists m \in \mathbb{N}, \ \exists c_i \in \mathbb{R}_+ \text{ for } i = 0, \dots, m-1 \text{ such that}$$
(5.20)  
$$A^m = \sum_{i=0}^{m-1} c_i A^i,$$

or, equivalently, if

$$g(\lambda) = \lambda^m - \sum_{i=0}^{m-1} c_i \lambda^i = 0, \quad \forall \lambda \in \operatorname{spec}(A).$$

In terms of the characteristic polynomial of A,  $p_A$ , the existence of a positive recursion implies that  $g = p_A Q$ , where Q is a polynomial of degree q with  $0 \le q \le m$ . It is then immediate that

$$m = n + q \ge n. \tag{5.21}$$

Before presenting our main results on polyhedrality of reachable subsets, we report a key theorem ([46, Th. 5]). In the following Q denotes the set of all real polynomials of the form  $c_n x^n - \sum_{i=0}^{n-1} c_i x^i$ , where  $n \ge 1$ ,  $c_n > 0$ , and  $c_i \ge 0$  for all i.

**Theorem 5.12.** [46, Th. 5] Let  $\{a_1, ..., a_k\}$  be given complex numbers, and let P(x) be the polynomial  $x^k - a_1 x^{k-l} - \cdots - a_k$ . Then conditions (A), (B) and (C) below are equivalent:

- (A) Any infinite sequence  $(u_n)_{n\geq 0}$  of complex numbers which satisfies the recursion  $u_{n+k} = a_1 u_{n+k-1} + a_2 u_{n+k-2} + \dots + a_k u_n$  for  $n \geq 0$ , also satisfies a recursion with positive coefficients.
- (B) The polynomial P(x) divides a polynomial in Q.
- (C) In case the polynomial P(x) has a positive root r, then all conditions (1)-(4) below are satisfied:
  - (C1)  $r \ge |\alpha|$  for any root  $\alpha$  of P(x).
  - (C2) if  $\alpha = r$  for some root  $\alpha$  of P(x), then  $\alpha/r$  is a root of unity.
  - (C3) all roots P(x) with absolute value r are simple.
  - (C4) if  $P(r) = P(r\epsilon) = 0$ , where  $\epsilon^k = 1$  with  $k \ge 1$  minimal, then P(x) has no roots of the form  $s\omega$  where 0 < s < r and  $\omega^k = 1$ .

We are now in the position to state a characterization of Proposition 5.9 in terms of spec(A), hence, providing numerically verifiable conditions as to when (5.17) holds.

**Theorem 5.13** (Polyhedrality of Reachset<sub> $\infty$ </sub>(*A*, *b*)). *Given an irreducible matrix A*  $\in \mathbb{R}^{n \times n}_+$  and  $b \in \mathbb{R}^n_+$ , the following statements are equivalent:

(a) The infinite-time reachable subset is polyhedral, hence there exists an integer  $k^* \in \mathbb{Z}_+$  such that

 $\operatorname{Reachset}_{\infty}(A, b) = \operatorname{cone}([\operatorname{conmat}_{k^*}(A, b) A_{f,0}b \dots A_{f,h-1}b]).$ 

Denote the lowest integer for which the above equality holds by  $k_{vert}^{\infty} \in \mathbb{Z}_+$ .

- (b) The matrix  $A_2$  defined in Definition 5.22, satisfies a positive recursion.
- (c) If there exists a positive  $\lambda_r \in \text{spec}(A_2)$ , then the following conditions all hold:
  - (c1)  $\lambda_r = \rho(\mathbf{A}_2)$ .
  - (c2) For any  $\lambda \in \sigma^{\rho}(A_2)$ ,  $\lambda = \rho(A_2)\exp(\phi_{\lambda}2\pi i)$ , where  $\phi_{\lambda} \in \mathbb{Q}$  is a rational number.
  - (c3)  $\sigma^{\rho}(A_2)$ , defined in Definition 5.22, includes only simple eigenvalues.
  - (c4) Given  $M \in \mathbb{Z}_+$  by Lemma 5.23, no  $\lambda^- \in \sigma^-(A_2)$  has a polar angle which is an integer multiple of  $2\pi/Mh$ .

Note that the condition (a) of Theorem 5.13 involves the determination of the integer  $k^*$ , which is in principle a test with an infinite number of steps. Similarly, condition (b) is a test with an infinite number of steps. However, condition (c) of the theorem is a finite test though it requires the exact eigenvalues.

*Proof.* (a) $\Rightarrow$ (b) $\Rightarrow$ (c): Since Reachset<sub> $\infty$ </sub>(*A*, *b*) is polyhedral, according to Corollary 5.10, there is a sufficiently large  $k \ge n - h$  such that the equation

$$A[b \ Ab \ \dots \ A^{k-1}b \ A_{f,0} \ \dots \ A_{f,h-1}] = [b \ Ab \ \dots \ A^{k-1}b \ A_{f,0}b \ \dots \ A_{f,h-1}b]X$$

has a solution  $X \ge 0$ . It can be easily verified using (5.17)-(5.19) that

$$\boldsymbol{X} = \begin{bmatrix} \boldsymbol{X}_{1} & \boldsymbol{0} \\ \boldsymbol{X}_{3} & \boldsymbol{X}_{2} \end{bmatrix}, \quad \boldsymbol{X}_{1} = \begin{bmatrix} \boldsymbol{0} & \boldsymbol{0} & \cdots & \boldsymbol{0} & \boldsymbol{\alpha}_{0} \\ \boldsymbol{1} & \boldsymbol{0} & \cdots & \boldsymbol{0} & \boldsymbol{\alpha}_{1} \\ \boldsymbol{0} & \boldsymbol{1} & \boldsymbol{0} & \boldsymbol{\alpha}_{2} \\ \vdots & \ddots & \vdots \\ \boldsymbol{0} & \cdots & \boldsymbol{0} & \boldsymbol{1} & \boldsymbol{\alpha}_{k-1} \end{bmatrix}, \quad (5.22)$$
$$\boldsymbol{X}_{2} = \begin{bmatrix} \boldsymbol{0} & \boldsymbol{0} & \cdots & \boldsymbol{0} & \boldsymbol{\beta}_{0} \\ \boldsymbol{1} & \boldsymbol{0} & \cdots & \boldsymbol{0} & \boldsymbol{0} \\ \boldsymbol{1} & \boldsymbol{0} & \cdots & \boldsymbol{0} & \boldsymbol{0} \\ \vdots & \ddots & \vdots \\ \boldsymbol{0} & \cdots & \boldsymbol{0} & \boldsymbol{1} & \boldsymbol{0} \end{bmatrix}, \quad \boldsymbol{X}_{3} = \begin{bmatrix} \boldsymbol{0} & \boldsymbol{0} & \cdots & \boldsymbol{0} & \boldsymbol{\beta}_{0} \\ \boldsymbol{0} & \boldsymbol{0} & \cdots & \boldsymbol{0} & \boldsymbol{\beta}_{1} \\ \boldsymbol{0} & \boldsymbol{0} & \boldsymbol{0} & \boldsymbol{\beta}_{2} \\ \vdots & \ddots & \vdots \\ \boldsymbol{0} & \cdots & \boldsymbol{0} & \boldsymbol{0} & \boldsymbol{\beta}_{h-1} \end{bmatrix}. \quad (5.23)$$

constitutes a solution, where  $X_1 \in \mathbb{R}_+^{k \times k}$ ,  $X_2 \in \mathbb{R}_+^{h \times h}$ , and  $X_3 \in \mathbb{R}_+^{h \times k}$ . Let  $p_{X_1}(\lambda) = \det(\lambda I - X_1)$  and  $p_{X_2}(\lambda) = \det(\lambda I - X_2)$ . Since by assumption,  $k \ge n - h$  and rank(conmat<sub>n</sub>(A, b)) = n, due to [47, Lemma 3.10],  $p_A(\lambda)$  divides  $p_X(\lambda) = p_{X_1}(\lambda)$  $p_{X_2}(\lambda) = (\lambda^h - \rho^h(A))(\lambda^k - \alpha_{k-1}\lambda^{k-1} - \dots - \alpha_0)$ . Since A is irreducible with cyclicity index h,  $p_A(\lambda)$  can be expressed as  $p_A(\lambda) = p_{A_1}(\lambda)p_{A_2}(\lambda) = (\lambda^h - \rho^h(A))p_{A_2}(\lambda)$ . Therefore,  $p_{A_2}(\lambda)$  divides  $p_{X_2}(\lambda)$ , which, due to statements (A) and (B) of Theorem 5.12, proves that  $A_2$  has a positive recursion of the form  $A_2^{k^*} - \gamma_{k^*-1}A_2^{k^*-1} - \cdots - \gamma_0 I = 0$  for some  $n - h \le k^* \le k$  and for some  $\gamma \in \mathbb{R}_+^{k^*}$ . Assume  $A_2$  has a positive eigenvalue. Since  $A_2$  satisfies a positive recursion, the statements (C1-C4) in (C) of Theorem 5.12 hold for  $p_{A_2}(\lambda)$ . It is straightforward to check that this implies that (c1)-(c4) holds<sup>1</sup>.

(c)⇒(b)⇒(a): Assume  $A_2$  has a positive eigenvalue. We need to prove that statements (c1)-(c4) imply a positive recursion for  $A_2$  of the form  $A_2^{k^*} - \alpha_{k^*-1}A_2^{k^*-1} - \cdots - \alpha_0 I = 0$ , for  $k^* \ge n - h$  and  $\alpha \in \mathbb{R}_+^{k^*}$ , and that, in turn, implies polyhedrality of the infinite-time reachable subset.

First we show that the statements (c1)-(c4) imply the statements (C1)-(C4) of Theorem 5.12. The statement  $\lambda_r \in \sigma^{\rho}(A_2)$  implies (C1) of Theorem 5.12. The requirement of all  $\lambda \in \sigma^{\rho}(A_2)$  having a rational polar phase implies (C2). The requirement of all  $\lambda \in \sigma^{\rho}(A_2)$  being simple implies (C3), and (C4) is implied from  $\sigma^-(A_2)$  including no eigenvalue with polar phase  $2\pi m/Mh$  for any  $m \in \mathbb{Z}$  [9, Theorem 2.2.20]. Next, invoking the equivalence between (C) and (B) of Theorem 5.12 for  $p_{A_2}(\lambda)$ , one can observe that there is a polynomial  $Q(\lambda)$  of positive degree such that

$$g(\lambda) = p_{A_2}(\lambda)Q(\lambda) = \lambda^{k^*} - \alpha_{k^*-1}\lambda^{k^*-1} - \dots - \alpha_0 = 0,$$
 (5.24)

for  $k^* \ge n - h$  and  $\alpha \in \mathbb{R}^{k^*}_+$ . It follows from (5.20) that  $A_2$  has a positive recursion, which results in (b).

Given (b), there exists a polynomial  $g(\lambda)$  of degree  $k^* \ge n - h$  satisfying (5.24), from which one concludes that  $p_A(\lambda) = p_{A_1}(\lambda)p_{A_2}(\lambda)$  divides  $h(\lambda) = p_{A_1}(\lambda)g(\lambda) = (\lambda^h - \rho^h(A))(\lambda^{k^*} - \alpha_{k^*-1}\lambda^{k^*-1} - \dots - \alpha_0)$ . Now consider the equation AM = MXwith  $M = [b \ Ab \ \dots \ A^{k^*-1}b \ A_{f,0}b \ \dots \ A_{f,h-1}b]$ , where  $X \in \mathbb{R}^{(n+k^*)\times(n+k^*)}$  is an unknown matrix. Since commat<sub>k\*</sub>(A, b) is full rank by assumption and  $k^* \ge n - h$ , M is of full rank as well. Then, it is known from [47, Lemma 10] that  $p_A(\lambda)$  divides  $p_X(\lambda)$ . Hence, we can choose X such that  $p_X(\lambda) = h(\lambda)$ . A possible choice of X, having substituted  $k^*$  for k, is then given by (5.22)-(5.23). It is clear from (5.22)-(5.23) that X admits a positive solution. Based on Corollary 5.10, this implies that Reachset<sub> $\infty$ </sub>(A, b) is polyhedral.

*Remark* 4. For a polyhedral Reachset<sub> $\infty$ </sub>(*A*, *b*) the following can be observed:

- (a) Due to (5.21) and from the second part of the proof of Theorem 5.13 the vertex number of Reachset<sub> $\infty$ </sub>(*A*, *b*),  $k_{vert}^{\infty}$ , is at least n h, which implies that Reachset<sub> $\infty$ </sub>(*A*, *b*) has at least *n* generators. It has exactly *n* generators (i.e., it is simplicial) if and only if the characteristic polynomial  $p_{A_2}$  of  $A_2$  has non-positive coefficients.
- (b) In the view of Lemma 5.25, Reachset<sub> $\infty$ </sub>(*A*, *b*) can be expressed as Reachset<sub> $\infty$ </sub>(*A*, *b*) = cone([*b Ab* ... *A*<sup>*k*-1</sup>*b v*<sub>*f*,0</sub> ... *v*<sub>*f*,*h*-1</sub>]), where *v*<sub>*f*,0</sub>, ..., *v*<sub>*f*,*h*-1</sub> are the *h* distinct positive eigenvectors of *A*<sup>*h*</sup> associated with the eigenvalue  $\rho^h(A)$ .

<sup>&</sup>lt;sup>1</sup>Condition  $\lambda_r \in \sigma^{\rho}(A_2)$  follows from (C1) of Theorem 5.12, and conditions (c2) and (c3) are, respectively, a direct result of (C2) and (C3). Finally, (c4) is implied from (C4) using Lemma 5.23.

*Example* 3 (polyhedral Reachset $\infty(A, b)$ ). Consider discrete-time linear time-invariant positive system of (5.4) with system matrices

$$\boldsymbol{A} = \begin{bmatrix} 0.9727 & 0 & 0.0263 \\ 0.0388 & 0.1273 & 0.2156 \\ 0 & 3.4497 & 0 \end{bmatrix}, \ \boldsymbol{b} = \begin{bmatrix} 0 \\ 1 \\ 1 \end{bmatrix}$$

where *A* is primitive, i.e., is irreducible with cyclicity index h = 1. We have spec(*A*) = {1, 0.9, -0.8}. We can assume  $A_1 = 1$ , and  $A_2 = \text{diag}(0.9, -0.8)$ . Using Theorem 5.13, it is immediate that conditions (c1) and (c2) hold as  $\lambda = 0.9$  is a simple eigenvalue of  $A_2$ , which equals the spectral radius of  $A_2$ . Condition (c1) holds as well since the polar angle of  $\lambda = -0.8$  is not an integer multiple of the polar angle of  $\lambda = 0.9$ . Hence, it can be concluded that the infinite-time reachable subset Reachset<sub> $\infty$ </sub>(*A*, *b*) is polyhedral. We can also conclude that  $A_2$  has a positive recursion, which is readily verified as  $p_{A_2}(\lambda) = \lambda^2 - 0.1\lambda - 0.72$ . Example 5.41 illustrates the growth of Reachset<sub>*k*</sub>(*A*, *b*). It can be observed that Reachset<sub>f</sub>(*A*, *b*) is not polyhedral since the cone keeps growing for increasing values of *k*. Its closure is, however, polyhedral as shown in Figure 5.41d.

*Example* 4 (non-polyhedral Reachset<sub> $\infty$ </sub>(*A*, *b*)). Consider the time-invariant discrete-time linear positive system of (5.4) with system matrices

$$\boldsymbol{A} = \begin{bmatrix} 0 & 1 & 0 \\ 1 & 0 & 0.5 \\ 0 & 0.4 & 1 \end{bmatrix}, \ \boldsymbol{b} = \begin{bmatrix} 0 \\ 1 \\ 0 \end{bmatrix},$$

where *A* has cyclicity index h = 1 with spec(*A*) = {-1.05, 0.7116, 1.3383}. One can assume  $A_1 = 1.3383$  and  $A_2 = \text{diag}(-1.05, 0.7116)$ . It is immediate that condition (c1) of Theorem 5.13 is not satisfied as  $0.7116 \neq \rho(A_2)$ . Therefore, Reachset<sub> $\infty$ </sub>(*A*, *b*) is not polyhedral. This is illustrated by Figure 5.42d, from which it is clear that Reachset<sub> $\infty$ </sub>(*A*, *b*) is approaching a round cone as introduced in Section 5.2.



Figure 5.41: (a), (b), (c): The growth of the reachability cone Reachset<sub>k</sub>(*A*, *b*) of Example 3 for different values of *k*, where generators of the cone are marked by asterisks, and the Frobenius eigenvector is marked by a red dot. (d): The growth of the reachable cone mapped on the 3-dimensional simplex  $S = \{x \in \mathbb{R}^3_+ | \mathbb{I}^T x = 1\}.$ 

# 5.5. When Is the Finite-Time Reachable Subsets a Polyhedral Set?

The polyhedrality of the finite-time reachability set from the origin,  $\text{Reachset}_f(A, b)$ , will be proven to be a special case of polyhedrality of  $\text{Reachset}_{\infty}(A, b)$  but with stricter requirements.

In this section we investigate the polyhedrality of the finite-time reachable set from the origin, Reachset<sub>f</sub>(A, b). Consider a linear positive system with an irreducible system matrix  $A \in \mathbb{R}^{n \times n}_+$  with the cyclicity index  $h \in \{1, ..., n\}$ . It follows from Proposition 5.9 that the finite-time reachable set from the origin

Reachset<sub>f</sub>(A, b) is polyhedral if and only if there exists a positive integer  $k^* \in \mathbb{Z}_+$ 

104



k = 6 (red and blue regions), k = 10 (red, blue and green regions). Reachset $\infty(A, b)$  approaches a "round cone".

Figure 5.42: (a), (b), (c): The growth of the reachability cone Reachset<sub>k</sub>(*A*, *b*) of Example 4 for different values of *k*, where generators of the cone are marked by asterisks, and the Frobenius eigenvector is marked by a red dot. (d): The growth of the reachable cone mapped on the 3-dimensional simplex  $S = \{x \in \mathbb{R}^3_+ | \mathbb{1}^T x = 1\}.$ 

such that

$$\operatorname{Reachset}_{k^*+1}(A, b) \subseteq \operatorname{Reachset}_{k^*}(A, b), \qquad (5.25)$$

$$\Leftrightarrow A^{k^*} \boldsymbol{b} \in \text{Reachset}_{k^*}(\boldsymbol{A}, \boldsymbol{b}). \tag{5.26}$$

The smallest  $k^*$  for which (5.26) holds is referred to as the *vertex number*,  $k_{vert}$ , of Reachset<sub>f</sub>(A, b). Note that (5.26) also implies that

$$\operatorname{cone}\left(\left[A_{f,0}\boldsymbol{b} \ \dots \ A_{f,h-1}\boldsymbol{b}\right]\right) \subseteq \operatorname{Reachset}_{k_{\operatorname{vert}}}(\boldsymbol{A},\boldsymbol{b}), \tag{5.27}$$

which is clearly a restriction on (5.17).

*Corollary* 5.14. For an irreducible  $A \in \mathbb{R}^{n \times n}_+$  with cyclicity index  $1 \le h \le n$  and for  $b \in \mathbb{R}_+$ , equivalence of the following statements follows directly from the above argument:

- (a) Reachset<sub>f</sub>(A, b) is polyhedral.
- (b) There exists an integer  $k_{\text{vert}} \in \mathbb{Z}_+$  such that  $\operatorname{cone}([\boldsymbol{b} A \boldsymbol{b} \dots A^k \boldsymbol{b}])$  is *A*-invariant for any  $k \ge k_{\text{vert}}$ .
- (c) There exists an integer  $k_{\text{vert}} \in \mathbb{Z}_+$  such that for the matrix equation

$$A[b Ab ... A^{k-1}b] = [b Ab ... A^{k-1}b]X,$$

there exists a solution  $X \in \mathbb{R}^{(k) \times (k)}_+$ , with  $k \ge k_{\text{vert}}$ .

(d) Based on (5.27) and Lemma 5.25, there exists an integer  $k_{\text{vert}} \in \mathbb{Z}_+$  such that for any  $k \ge k_{\text{vert}}$ , cone $([v_{f,0} \dots v_{f,h-1}]) \subseteq \text{Reachset}_k(A, b)$ .

The following theorem provides necessary and sufficient conditions on spec(A) for polyhedrality of  $\text{Reachset}_f(A, b)$ . These conditions turn out to be a conservative version of those of Theorem 5.13.

**Theorem 5.15** (Polyhedrality of Reachset<sub>f</sub>(A, b)). Let  $A \in \mathbb{R}^{n \times n}_+$  be irreducible with index of cyclicity  $h \in \{1, ..., n\}$  and  $b \in \mathbb{R}^n_+$ . Then the following statements are equivalent:

- (a) The finite-time controllable subset is polyhedral and hence there exists an integer  $k^* \in \mathbb{Z}_+, k^* \ge k_{\text{vert}}$  such that  $\text{Reachset}_f(A, b) = \text{Reachset}_{k^*}(A, b)$ .
- (b) A has a positive recursion.
- (c) The matrix  $A_2$ , defined in Definition 5.22, does not have any positive eigenvalue.

*Proof.* (a)  $\Rightarrow$  (b)  $\Rightarrow$  (c): Based on Corollary 5.14 with  $k \ge n$  we obtain

 $A(\operatorname{conmat}_k(A, b)) = (\operatorname{conmat}_k(A, b))X,$ 

where  $X \in \mathbb{R}^{k \times k}_+$  is given by

$$\boldsymbol{X} = \begin{bmatrix} 0 & 0 & \cdots & 0 & \alpha_0 \\ 1 & 0 & \cdots & 0 & \alpha_1 \\ 0 & 1 & & 0 & \alpha_2 \\ \vdots & \ddots & & \vdots \\ 0 & \cdots & 0 & 1 & \alpha_{k-1} \end{bmatrix}.$$

Since, by assumption, conmat<sub>n</sub>(*A*, *b*) is of full rank and  $k \ge n$ , there exists [47, Lemma 3.10] a polynomial  $Q(\lambda)$  of positive degree such that  $p_A(\lambda)Q(\lambda) = p_X(\lambda) = \lambda^k - \alpha_{k-1}\lambda^{k-1} - \cdots - \alpha_1\lambda - \alpha_0$ , which, in the view of Definition 5.11, proves that *A* has a positive recursion. Noting that (b) is equivalent to condition (B) of Theorem 5.12 ([46, Th. 5]), all conditions (C1)-(C4) are then fulfilled. In particular, (C4) holds as conditions (C1)-(C3) are already satisfied for a positive irreducible matrix due to the Perron-Frobenius theorem [9, Th. 2.1.4, 2.2.20]. Condition (C4) requires that no eigenvalue  $\lambda^- \in \sigma^-(A)$  has a polar angle of  $2\pi k/h$  for k = 0, ..., h - 1. Since spec(*A*) is invariant under a polar rotation of  $2\pi m/h$  for

any  $m \in \mathbb{Z}$ , no  $\lambda^- \in \sigma^-(A)$  is then positive. Noting that for an irreducible matrix,  $(\sigma^{\rho}(A) \setminus \{\rho(A)\}) \cap \mathbb{R}_{s+} = \emptyset$  and that spec $(A_2) = (\sigma^-(A) \cup \sigma^{\rho}(A) \setminus \{\rho(A)\})$ , one concludes that  $A_2$  has no positive eigenvalue.

(c) ⇒ (b) ⇒ (a): Given (c), we have spec( $A_2$ )∩ $\mathbb{R}_{s+} = \emptyset$ . For an irreducible matrix it holds that  $(\sigma^{\rho}(A) \setminus \{\rho(A)\}) \cap \mathbb{R}_{s+} = \emptyset$ . Since spec( $A_2$ ) =  $\sigma^{-}(A) \cup (\sigma^{\rho}(A) \setminus \{\rho(A)\})$ , it follows that  $\sigma^{-}(A) \cap \mathbb{R}_{s+} = \emptyset$ , from which it can be immediately concluded that  $\nexists \lambda \in \sigma^{-}(A)$ ,  $\lambda = |\lambda| \exp(i2\pi m/h)$  for any  $m \in \mathbb{Z}$ . Hence, we establised that (C4) of Theorem 5.12 ([46, Th. 5]) holds for  $p_A(\lambda)$ . Moreover, statements (C1)-(C3) hold as well for  $p_A$  as A is irreducible. Therefore, due to (B) of Theorem 5.12, there exists a polynomial Q of positive degree, such that  $p_A(\lambda)Q(\lambda) = \lambda^{k^*} - \alpha_{k^*-1}\lambda^{k^*-1} - \cdots - \alpha_1\lambda - \alpha_0$ , where  $k^* \ge n$  and  $\alpha_i \ge 0$ ,  $i = 0, 1, \dots, k^* - 1$ . This proves that A has a positive recursion based on Definition 5.11. Then, (a) immediately follows as  $A^{k^*}b = \sum_{i=0}^{k^*-1} \alpha_i A^i b$ .

*Remark* 5. Note that since deg $(Q(\lambda)) \ge 0$ ,  $k_{\text{vert}}$  of Reachset<sub>f</sub>(A, b) is at least n, and it equals n if and only if  $p_A(\lambda) = \lambda^n - \alpha_{n-1}\lambda^{n-1} - \cdots - \alpha_1\lambda - \alpha_0$  with  $\alpha_i \ge 0$ , i = 0, ..., n - 1. Hence Reachset<sub>f</sub>(A, b) is a simplicial cone (i.e., has n generators) if and only if the characteristic polynomial of A has non-positive coefficients. One such matrix is a cyclic matrix with cyclicity index h = n as  $p_A(\lambda) = \lambda^n - \rho^n(A)$ .

Comparing Theorem 5.13 to Theorem 5.15 reveals that the latter is a restricted version of the former. For example, condition (b) of Theorem 5.13 requires a part of A (i.e.,  $A_2$ ) to have a positive recursion while that of Theorem 5.15 requires the entire A to have a positive recursion.

*Example* 5 (polyhedral Reachset<sub>f</sub>(A, b)). Consider the time-invariant discrete-time linear positive system of (5.4) with system matrices

| <i>A</i> = | ΓO      | 1.6333 | 1.1049 | 0 1    | , <b>b</b> = | 707 |   |
|------------|---------|--------|--------|--------|--------------|-----|---|
|            | 23.5667 | 6.0944 | 0      | 0      |              | 0   |   |
|            | 0       | 0      | 1.1225 | 1.0672 |              | 1   | , |
|            | 0       | 1.6611 | 0      | 0.7830 |              | 1   |   |

where *A* is irreducible with cyclicity index h = 1. It can be verified that spec(*A*) =  $\{10, -4, 1 + 1i, 1 - 1i\}$ . One can recognize that no eigenvalue of  $A_2 = \text{diag}(-4, 1 + i, 1 - i)$  is positive. Therefore, condition (c) of Theorem 5.15 holds and it follows that *A* has a positive recursion. In fact, it can be verified that in this case it holds that  $A^6 = 166.7569I_4 + 16.1434A + 39.7036A^4 + 6.0262A^5$ , where  $I_4$  denotes the identity matrix of dimension  $4 \times 4$ . In addition, we can conclude that Reachset<sub>f</sub>(*A*, *b*) is polyhedral with  $k_{\text{vert}} = 6$ . This is illustrated by Figure 5.51, where it is observed that Reachset<sub>k</sub>(*A*, *b*) stops growing for  $k \ge 6$ , i.e., Reachset<sub>k</sub>(*A*, *b*) = Reachset<sub>6</sub>(*A*, *b*), with  $C_{\text{lim}}$  introduced in Definition 5.24. Note that in this particular example, since h = 1, we have  $C_{\text{lim}} = \text{cone}(A_{f,0}b) = \{cv_f | c \in \mathbb{R}_{s+}\}$ , where  $v_f$  is the Frobenius eigenvector of  $A^h$ .

*Remark* 6 (Concluding Remark on Theorems 5.13 and 5.15). Theorems 5.13 and 5.15 emphasize the equivalence between the three statements; but this does not



Figure 5.51: Example 5: growth of the reachability cone mapped on the 3-dimensional simplex  $S = \{x \in \mathbb{R}^3_+ | \mathbb{1}^T x = 1\}$ ; the generators of the cone and the Frobenius eigenvector are, respectively, marked by asterisks and a dot.

imply that all cases are directly verifiable. In fact, it is very difficult to verify statement (b) directly especially since  $k_{vert}^{\infty}$  and  $k_{vert}$  are not known a priori. In practice, statement (a) is practically what one is interested in, and (c) provides numerically verifiable conditions. Statement (b) serves the dual purpose of facilitating the proof and providing insight into otherwise-very-abstract statement (a) and statement (c) by relating them to the matrix having a (partial) positive recursion. Moreover, the characterization (b) will be useful for a different algebraic characterization which is to be developed.

#### 5.5.1. Special Case

So far it has been assumed that  $\operatorname{rank}(\operatorname{conmat}_n(A, b)) = n$ . Based on this assumption, the polyhedrality of the finite-time reachable set only depends on the spectrum of A. In addition,  $k_{\operatorname{vert}} \ge n$  for  $\operatorname{Reachset}_{f}(A, b)$ . We now point out that in the absence of such an assumption,  $\operatorname{Reachset}_{f}(A, b)$  can depend on the structure of b and that the vertex number can be less than n. In particular, it will be shown that  $k_{\operatorname{vert}} = h$  if  $b \in \mathbb{R}^{n}_{+}$  is of a particular structure.

**Theorem 5.16.** Let  $A \in \mathbb{R}^{n \times n}_+$  be irreducible with cyclicity index h with  $0 \le h \le n-1$ . Then, Reachset<sub>f</sub> $(A, b) = \text{cone}(\text{conmat}_h(A, b))$  if  $b \in \text{cone}([v_{f,0} \dots v_{f,h-1}])$ , where  $v_{f,i}$ ,  $i = 0, \dots, h-1$  are the h positive eigenvectors of  $A^h$ .

*Proof.* Assume  $\boldsymbol{b} = \sum_{i=0}^{h-1} c_i \boldsymbol{v}_{f,i}$  for some  $\boldsymbol{c} \in \mathbb{R}^h_+$ . Then, since

$$\boldsymbol{A}^{h}\boldsymbol{b} = \sum_{i=0}^{h-1} c_{i}\rho^{h}(\boldsymbol{A})\boldsymbol{v}_{f,i} = \rho^{h}(\boldsymbol{A})\boldsymbol{b},$$

it is immediate to see that  $A(conmat_h(A, b)) = (conmat_h(A, b))X$  has a positive

solution

$$\boldsymbol{X} = \begin{bmatrix} 0 & 0 & \cdots & 0 & \rho^h(\boldsymbol{A}) \\ 1 & 0 & \cdots & 0 & 0 \\ 0 & 1 & & 0 & 0 \\ \vdots & & \ddots & & \vdots \\ 0 & \cdots & 0 & 1 & 0 \end{bmatrix},$$

which, in the view of Corollary 5.14, completes the proof.

For *A* primitive (i.e., h = 1), this results in the obvious case of Reachset<sub>f</sub>(*A*, *b*) being a ray along the Frobenius eigenvector  $v_f$  of *A* when  $b = cv_f$  for any  $c \ge 0$ .

# 5.6. Does the Reachable Set Contain a Pre-specified Set?

A direct consequence of polyhedrality of infinite- or finite-time reachable subset discussed in Section 5.4 and Section 5.5 is that it enables us to determine whether a given subset of the positive vector space is reachable from the origin. Given a cone  $C_{obj} \subseteq \mathbb{R}^n_+$  of control objectives or a subset of  $\mathbb{R}^n_+$ , the problem considered here is to investigate whether  $C_{obj}$  is contained in Reachset<sub>f</sub>(A, b) or in Reachset<sub> $\infty$ </sub>(A, b). Of particular interest is when  $C_{obj} \subset \mathbb{R}^n_+$  is a polyhedral cone or a polytope. Note that if the control objective cone  $C_{obj}$  is not polyhedral then once can outer approximate it by a polyhedral cone  $C_{out} \subseteq \mathbb{R}^n_+$  such that  $C_{obj} \subset C_{out}$ .

Here, it is assumed that the reachability cone or its closure is polyhedral and that its corresponding vertex number or an upper bound of it is known. Note that the authors are not aware of any method to directly compute an upper bound for  $k_{\text{vert}}$  or for  $k_{\text{vert}}^{\infty}$ . Nonetheless, such an upper bound could be imposed by length of the control sequence that can be practically applied. Let  $N \in \mathbb{Z}_+$  denote an upper bound to  $k_{\text{vert}}^{\infty}$ , or, where applicable, an upper bound to  $k_{\text{vert}}$ . Hence  $\text{Reachset}_{\infty}(A, b) = \text{cone}([b \dots A^{N-1}b \ v_{f,0} \dots \ v_{f,h-1}])$  and/or  $\text{Reachset}_{f}(A, b) = \text{cone}([b \dots A^{N-1}b])$ .

*Proposition* 5.17. Let  $C_{obj} = cone([\mathbf{p}_1 \dots \mathbf{p}_m])$  or  $C_{obj} = conv([\mathbf{p}_1 \dots \mathbf{p}_m])$ , where  $\mathbf{p}_i \in \mathbb{R}^n_+$ , i = 1, ..., m. Then

(a) C<sub>obj</sub> is reachable in finite time if and only if

 $\forall p \in \{p_1, \dots, p_m\}, p \in \text{Reachset}_f(A, b).$ 

(b) C<sub>obj</sub> is reachable in infinite time (to be called *almost reachable*) if and only if

$$\forall p \in \{p_1, \dots, p_m\}, p \in \text{Reachset}_{\infty}(A, b), \text{ and} \\ \exists p' \in \{p_1, \dots, p_m\} \text{ such that } p' \notin \text{Reachset}_{f}(A, b).$$

*Proof.* The proof is obvious from Definition 5.14 and considering the fact that a cone can be expressed as a positive combination of its generators.

It is obvious from Proposition 5.17, that checking for reachability from the origin involves checking the following condition for each  $i \in \{1, ..., m\}$ :

$$\exists x_i \in \{ z | Mz = p_i, z \in \mathbb{R}^N_+ \}, \tag{5.28}$$

where  $M \in \mathbb{R}^{n \times N}_{+}$ . Depending on the problem being investigated, either  $M = [b \dots A^{N-1}b v_{f,0} \dots v_{f,h-1}] \text{ or } M = [b \dots A^{N-1}b].$ 

In general, since  $N \ge n$  (see Remark 4 and Remark 5), (5.28) defines an underdetermined system of equations. It is known that the positive solution of (5.28)is not unique in general [48, 49], and that uniqueness is guaranteed when the solution is sufficiently sparse [48]. The author of [50] characterizes necessary and sufficient conditions on the polytope P = conv(M) for uniqueness of the solution, and he proves that a unique solution exists if and only if P is k-neighborly <sup>2</sup>. In [49, 53], an equivalent condition is presented in terms of the null space of M. In this regard, this problem relates to the sparse measurement problem, where the aim is to reconstruct a positive sparse vector from lower-dimensional linear measurements [54]. The results in this field do not directly apply here as the necessary sparsity condition is usually not met. In addition, we are not interested in finding the sparsest solution of (5.28), which is normally an NP-hard problem [48].

Consider for  $n \in \mathbb{Z}_+$  the positive matrix  $A \in \mathbb{R}^{n \times n}_+$ . Let  $N \in \mathbb{Z}_+$  with N > n be an upper bound of  $k_{\text{vert}}$  or an upper bound of  $k_{\text{vert}}^{\infty}$ . Denote by C(N, n) size of the set of all *n*-subsets of  $\mathbb{Z}_N = \{1, ..., N\}$ . Let the index set  $\mathcal{I}_i$  be an *n*-subset (i.e.,  $|\mathcal{I}_j| = n$ ) of  $\mathbb{Z}_N$  for j = 1, 2, ..., C(N, n) such that  $\bigcup_{j=1}^{C(N,n)} \mathcal{I}_j = \mathbb{Z}_N$  and  $\mathcal{I}_j \neq \mathcal{I}_k$ ,  $j, k = 1, 2, ..., C(N, n), j \neq k.$ 

Let  $I_{\mathcal{I}_i}$  denote the matrix with n columns, where the columns are chosen from the columns of  $I_N$  (i.e., the identity matrix of dimension N) according to the index set  $\mathcal{I}_i$  and let  $C_{obj} = \operatorname{cone}([\boldsymbol{p}_1 \dots \boldsymbol{p}_m])$ .

Proposition 5.18. Consider the above defined objects. Then, for any  $i \in \{1, ..., m\}$ , equation (5.28) has a solution  $x_i$  if and only if,

$$\boldsymbol{X}^{i} = \left\{ \boldsymbol{x}_{j}^{i} | \boldsymbol{x}_{j}^{i} = \boldsymbol{I}_{\mathcal{I}_{j}} (\boldsymbol{M} \boldsymbol{I}_{\mathcal{I}_{j}})^{-1} \boldsymbol{p}_{i}, \ \boldsymbol{x}_{j}^{i} \in \mathbb{R}^{N}_{+}, \ j = 1, \dots, C(N, n) \right\},$$
(5.29)

is a non-empty set.

*Proof.* From our assumption we have  $p_i \in \text{cone}(M)$ . Since N > n, due to Carathéodory theorem [55],  $p_i$  also lies in at least one simplicial cone generated by n columns of *M*. Let  $\mathcal{J}^i \subset \{1, ..., N\}$  with  $|\mathcal{J}^i| = n$  be an index set composed of the indices of the columns generating this simplicial cone, and let  $M_{J^i}$  denote the columns of Mcorresponding to  $\mathcal{J}^i$ . We can then write  $p_i \in \operatorname{cone}(M_{\mathcal{J}^i})$ , which can be expressed as  $MI_{\mathcal{J}^i} z^i = p_i$  having a solution  $z^i \in \mathbb{R}^n_+$ . Since *M* has full row rank and  $I_{\mathcal{J}^i}$  is of full column rank, one obtains  $z^i = (MI_{\mathcal{I}^i})^{-1} p_i$ . Finally, we obtain a solution  $x_i^i \in \mathbb{R}^N_+$ , where  $\mathbf{x}_{i}^{i} = \mathbf{I}_{\mathcal{I}^{i}} \mathbf{z}^{i} = \mathbf{I}_{\mathcal{I}^{i}} (\mathbf{M} \mathbf{I}_{\mathcal{I}^{i}})^{-1} \mathbf{p}_{i}$ .

The converse is proved in a straightforward manner by noticing that every  $z \in X^i$ satisfies (5.28). 

<sup>&</sup>lt;sup>2</sup>A k-neighborly polytope is a convex polytope in which every set of k or fewer vertices forms a face [51, 52].

*Remark* 7. Let  $X^i = \{x_1^i, ..., x_{q_i}^i\}$  for some  $q_i \in \mathbb{Z}_+$ . It is then clear from the proof of Proposition 5.18 that the set of solutions of (5.28) is the convex hull of  $X^i$ , i.e., we have for (5.28) that  $x_i \in \text{conv}(X^i)$ .

Note that even though Proposition 5.18 provides a method to determine whether  $C_{obj} \subseteq \operatorname{cone}(M)$  by checking inclusion of  $C_{obj}$  in any simplicial subcone of  $\operatorname{cone}(M)$ , the computational complexity of this method can be prohibitive as the check must be conducted for all C(N, n) simplicial subcones in the worst case. A more practical approach is then presented by the following proposition.

Proposition 5.19. Let

Define the following optimization problem for each  $i \in \{1, ..., m\}$ :

$$\min_{x_i} x_i^{\mathrm{T}} \mathbb{1}$$
(5.30)  
subject to  $Mx_i = p_i$ , and  $x_i \ge 0$ .

We then have the following:

- (a) The optimization problem (5.30) with  $M = M_{\infty}$  has an optimal solution  $x_i^* \in \mathbb{R}^N_+$  if and only if (5.28) has a solution with  $M = M_{\infty}$ .
- (b) The optimization problem (5.30) with  $M = M_f$  has an optimal solution  $x_i^* \in \mathbb{R}^N_+$  if and only if (5.28) has a solution with  $M = M_f$ .

*Proof.* If (5.28) has a solution, the set  $X^i$  in (5.29) is non-empty. As mentioned in Remark 7, the feasible set of (5.30) is  $conv(X^i)$ . Therefore, the convex optimization problem with linear penalty function converges to the minimum 1-norm solution in the feasible set. The converse is obvious.

*Example* 6. We conclude this section with an example illustrating the application of Proposition 5.19. Consider the system matrices of Example 5. Let  $C_{obj}$  be the polytope given by

$$C_{\text{obj}} = \Big\{ \boldsymbol{p} \in \mathbb{R}^4_+ \Big| \boldsymbol{p} = \sum_{i=1}^4 \lambda_i \boldsymbol{p}_i, \lambda_i \ge 0, \sum_{i=1}^4 \lambda_i = 1 \Big\},\$$

where

$$p_1 = [1, 3, 1, 1]^{\mathsf{I}}, p_2 = [1, 3, 4, 3]^{\mathsf{I}},$$
  
 $p_3 = [1, 2, 2, 1]^{\mathsf{T}}, p_4 = [1, 1, 2, 1]^{\mathsf{T}}.$ 

We will now check whether the system initially at rest can be steered to any point in  $C_{obj}$  in finite time. From Example 5, it is known that  $k_{vert} = 6$ . Thus taking M =

 $[b \ Ab \ ... \ A^5b]$ , we solve the linear programming problem (5.30) using the Dual-Simplex algorithm implemented in the Matlab Optimization Toolbox. The optimal solutions are obtained as

 $x_1^* = [0.1209, 0.3735, 0, 0.0078, 0, 0.0001]^T,$   $x_2^* = [2.3460, 0.6165, 0.0876, 0, 0.0003, 0]^T,$   $x_3^* = [0.2989, 0.6982, 0.0473, 0, 0.0003, 0]^T,$  $x_4^* = [0.2517, 0.7798, 0.0071, 0, 0.0003, 0]^T.$ 

Hence, the vertices of  $C_{obj}$  can be reached from the origin in a finite number of steps using positive inputs, which are determined by the solution vectors  $x_i^*$ . Moreover, since  $k_{vert} = 6$ , every vertex of  $C_{obj}$  can be reached in at most 6 steps from the origin. Since  $C_{obj}$  is the convex hull of its vertices, we can conclude that any point  $p = \sum_{i=1}^{4} \lambda_i p_i \in C_{obj}$  can be reached from the origin in at most 6 steps using the input sequence  $u^* = \sum_{i=1}^{4} \lambda_i x_i^*$ .

## **5.7.** Conclusions and Future Work

The main contribution of the chapter is the result that the reachable set from the origin of a linear positive system can be either a polyhedral cone or a nonpolyhedral cone depending on the system matrices. Among other applications, this has direct consequences for the realization problem, where the choice for the reachable subset from the origin is essential as observability of a linear positive system is then of interest only for states in the reachable set.

For a single-input case, necessary and sufficient conditions for polyhedrality of the reachable set from the origin and its closure are provided. These conditions are expressed in terms of characteristics of eigenvalues of the system. Finally, the chapter presents a method to determine for a positive linear system whether a given target set in the positive orthant can be reached from the origin.

There are several technical issues to be studied. Is it possible to determine in a finite number of steps for a positive matrix whether there exists a positive recursion for it?

In this chapter, we have focused on the single input case, where  $b \in \mathbb{R}^n_+$ . The problem of characterizing the reachability set from the origin for the multi-input case is an interesting problem because the results developed here are not directly applicable. The main issue, as noted in [47], is that the direct sum of two non-polyhedral cones may still result in a polyhedral cone. Therefore, one cannot apply the results of this chapter to a set of systems  $(A, b_i)$  separately, with  $b_i$  being a column of B.

Finally, it is also of interest to investigate the geometry of the reachable set when the controllability matrix is not of full rank. As far as the authors of this chapter know, this is still an open issue.

## References

- Y. Zeinaly, J. H. van Schuppen, and B. De Schutter, *Linear positive systems may have a reachable subset from the origin that is either polyhedral or non-polyhedral*, SIAM Journal on Matrix Analysis and Applications **41**, 279 (2020).
- [2] W. Leontief, *Input-Output Economics*, 2nd ed. (Oxford University Press, Oxford, 1986) p. 436.
- [3] J. Jacquez, *Compartmental analysis in biology and medicine, 2nd Ed.* (The University of Michigan Press, Ann Arbor, 1985) p. 560.
- [4] W. M. Haddad, V. Chellaboina, and Q. Hui, *Nonnegative and compartmental dynamical systems* (Princeton University Press, Princeton, 2010) p. 616.
- [5] J.M. van den Hof, Positive linear observers for linear compartmental systems, SIAM J. Control & Opt. 36, 590 (1998).
- [6] R. Shorten, F. Wirth, and D. Leith, A positive system model of TCP-like congestion control: Asymptotic results, IEEE/ACM Transactions on Networking 14, 616 (2006).
- [7] Y. Zeinaly, B. De Schutter, and H. Hellendoorn, An integrated model predictive scheme for baggage-handling systems: Routing, line balancing, and emptycart management, IEEE Trans. Control Systems Technology 23, 1536 (2015).
- [8] A. Berman, M. Neumann, and R. J. Stern, *Nonnegative Matrices in Dynamic Systems*, 1st ed. (Wiley-Interscience, New York, USA, 1989) p. 167.
- [9] A. Berman and R. J. Plemmons, *Nonnegative Matrices in the Mathematical Sciences*, 1st ed. (Academic Press, 1979) p. 340.
- [10] C. Davis, Theory of positive dependence, Amer. J. Math. 76, 733 (1954).
- [11] D. Gale, Convex polyhedral cones and linear inequalities, in Activity analysis of production and allocation, 13, edited by T. Koopmans (Wiley & Sons, New York, 1951) pp. 287–297.
- [12] J.S. Van der Graft, Spectral properties of matrices which have invariant cones, SIAM J. Appl. Math. 16, 1208 (1968).
- [13] H. M. Härdin and J. H. van Schuppen, Observers for linear positive systems, Linear Algebra & its Applications 425, 571 (2007).
- [14] R. Bru, S. Romero, and E. Sánchez, Canonical forms for positive discrete-time linear control systems, Linear Algebra & its Applications 310, 49 (2000).
- [15] P. G. Coxson and H. Shapiro, *Positive input reachability and controllability of positive systems*, Linear Algebra & its Applications **94**, 35 (1987).

- [16] L. Farina and S. Rinaldi, *Positive Linear Systems: Theory and Applications*, Pure and Applied Mathematics (John Wiley & Sons, Inc., New York, 2011) p. 318.
- [17] D. G. Luenberger, Introduction to Dynamic Systems Theory, Models and Applications (J. Wiley & Sons, New York, 1991) p. 464.
- [18] Z. Bartosiewicz, Linear positive control systems on time scales; controllability, Mathematics of Control, Signals, and Systems 25, 327 (2013).
- [19] M. Fanti, B. Maione, and B. Turchiano, *Controllability of multi-input positive discrete-time systems*, International Journal of Control **51**, 1295 (1990).
- [20] C. Guiver, D. Hodgson, and S. Townsley, *Positive state controllability of positive linear systems*, Systems & Control Letters 65, 23 (2014).
- [21] M. Valcher, Controllability and reachability criteria for discrete time positive systems, International Journal of Control 65, 511 (1996).
- [22] L. Caccetta and V. Rumchev, A survey of reachability and controllability for positive linear systems, Annals of Operations Research 98, 101 (2000).
- [23] T. Kaczorek, Some Recent Developments in Positive and Compartmental Systems, in Proc. SPIE (Photonics Applications in Astronomy, Communications, Industry, and High-Energy Physics Experiments II), Vol. 5484, edited by R. S. Romaniuk (SPIE, Bellingham, WA, 2004) pp. 1–18.
- [24] T. Kaczorek, *Positive 1D and 2D Systems*, 1st ed., Communications and Control Engineering (Springer, Berlin, 2002) p. 431.
- [25] D. Bertsekas and I. Rhodes, *Recursive state estimation for a set-membership description of uncertainty*, IEEE Transactions on Automatic Control **16**, 117 (1971).
- [26] J. Glover and F. Schweppe, Control of linear dynamic systems with set constrained disturbances, IEEE Transactions on Automatic Control 16, 411 (1971).
- [27] P. olof Gutman and M. Cwikel, Admissible sets and feedback control for discrete-time linear dynamical systems with bounded controls and states, IEEE Transactions on Automatic Control **31**, 373 (1986).
- [28] S. Keerthi and E. Gilbert, Computation of minimum-time feedback control laws for discrete-time systems with state-control constraints, IEEE Transactions on Automatic Control 32, 432 (1987).
- [29] L. Benvenuti and L. Farina, The geometry of the reachability set for linear discrete-time systems with positive controls, SIAM Journal on Matrix Analysis and Applications 28, 306 (2006).

- [30] J. Dieudonné, Foundations of Modern Analysis (Academic Press, New York, 1969) p. 412.
- [31] R. Brualdi and H. Ryser, Combinatorial Matrix Theory (Cambridge University Press, Cambridge, 1991) p. 367.
- [32] H. Minc, Nonnegative Matrices, 1st ed. (John Wiley & Sons, New York, 1988) p. 206.
- [33] R. Kalman, P. Falb, and M. Arbib, *Topics in Mathematical Systems Theory* (McGraw-Hill Book Co., New York, 1969) p. 358.
- [34] E. D. Sontag, Mathematical Control Theory: Deterministic Finite Dimensional Systems, 2nd ed., Text in Applied Mathematics, Vol. 6 (Springer, New York, NY, 1998) p. 532.
- [35] R. E. Kalman, A new approach to linear filtering and prediction problems, ASME Journal of Basic Engineering 82, 35 (1960).
- [36] R. E. Kalman, Mathematical description of linear dynamical systems, Journal of the Society for Industrial and Applied Mathematics Series A Control 1, 152 (1963).
- [37] W. M. Wonham, Linear Multivariable Control: A Geometric Approach, 3rd ed. (Springer-Verlag, Berlin, 1979) p. 350.
- [38] E. Lee and L. Markus, *Foundations of Optimal Control Theory* (Wiley, New York, 1967) p. 576.
- [39] A. Isidori, Nonlinear Control Systems, 3rd ed. (Springer, London, 1995) p. 549.
- [40] A. Isidori, *Nonlinear Control Systems II*, 1st ed., Communications and Control Engineering Series (Springer-Verlag, London, 1999) p. 293.
- [41] E. D. Sontag, *Polynomial Response Maps*, 1st ed., Lecture Notes in Control and Information Sciences, Vol. 13 (Springer-Verlag, Berlin, 1979) p. 168.
- [42] Z. Bartosiewicz, *Minimal polynomial realizations*, Mathematics of Control, Signals and Systems 1, 227 (1988).
- [43] J. Němcová and J. van Schuppen, Realization theory for rational systems: Minimal rational realizations, Acta Applicandae Mathematicae 110, 605 (2010).
- [44] R. Curtain and H. Zwart, An Introduction to Infinite-Dimensional Linear Systems Theory, 1st ed., Texts in Applied Mathematics No. 21 (Springer Verlag, Berlin, 1995) p. 698.
- [45] P. G. Coxson, L. C. Larson, and H. Schneider, *Monomial patterns in the se-quence A<sup>k</sup>b*, Linear Algebra & its Applications **94**, 89 (1987).

- [46] M. Roitman and Z. Rubinstein, On linear recursions with nonnegative coefficients, Linear Algebra & its Applications 167, 151 (1992).
- [47] L. Benvenuti and L. Farina, The geometry of the reachable set for linear discrete-time systems with positive controls, SIAM Journal on Matrix Analysis and Applications 28, 306 (2006).
- [48] D. L. Donoho and J. Tanner, Sparse nonnegative solution of underdetermined linear equations by linear programming, Proceedings of the National Academy of Sciences **102**, 9446 (2005).
- [49] M. Wang, W. Xu, and A. Tang, A unique "nonnegative" solution to an underdetermined system: From vectors to matrices, IEEE Transactions on Signal Processing 59, 1007 (2011).
- [50] D. L. Donoho, High-dimensional centrally symmetric polytopes with neighborliness proportional to dimension, Discrete & Computational Geometry 35, 617 (2006).
- [51] G. M. Ziegler, *Lectures on Polytopes*, 1st ed., Graduate Texts in Mathematics No. 152 (Springer, Berlin, 1995) p. 370.
- [52] B. Grünbaum, *Convex Polytopes*, 2nd ed., Pure and Applied Mathematics No. 16 (Wiley & Sons, London, 1967) p. 471.
- [53] D. L. Donoho and J. Tanner, Counting the faces of randomly-projected hypercubes and orthants, with applications, Discrete & Computational Geometry 43, 522 (2010).
- [54] M. A. Khajehnejad, A. G. Dimakis, W. Xu, and B. Hassibi, *Sparse recovery of nonnegative signals with minimal expansion*, IEEE Transactions on Signal Processing 59, 196 (2011).
- [55] I. Bárány and R. Karasev, Notes about the Carathéodory number, Discrete & Computational Geometry 48, 783 (2012).
- [56] R. A. Horn and C. R. Johnson, *Matrix Analysis*, 2nd ed. (Cambridge University Press, Cambridge, UK, 2012) p. 662.
- [57] L. Benvenuti and L. Farina, *Revisiting the linear recursions with nonnegative coefficients problem*, Linear Algebra and its Applications **535**, 191 (2017).

# Appendix

# **5.A.** Positive Matrices

The reader finds in this appendix a summary of the theory of positive matrices including concepts and decompositions as far as is necessary for the understanding of this chapter. This theory is well known and therefore not stated in the body of the chapter.

## **Decompositions of Positive Matrices**

As is well known in the theory of positive matrices, such matrices can be either reducible or irreducible as defined next. See the books [9, 56] for the definitions.

**Definition 5.20.** Consider a positive matrix  $A \in \mathbb{R}^{n \times n}_+$  for  $n \in \mathbb{Z}_+$ . Call this matrix *reducible* if

$$\exists P \in \mathbb{R}_{+}^{n \times n}, \text{ a permutation matrix,} \exists n_{1}, n_{2} \in \mathbb{Z}_{+}, \exists A_{11} \in \mathbb{R}^{n_{1} \times n_{1}}, A_{12} \in \mathbb{R}^{n_{1} \times n_{2}}, A_{22} \in \mathbb{R}^{n_{2} \times n_{2}}, \text{ such that } n = n_{1} + n_{2} \text{ and} A = P \begin{pmatrix} A_{11} & A_{12} \\ 0 & A_{22} \end{pmatrix} P^{T}.$$
(5.31)

Call the matrix A *irreducible* if (1)  $A \neq 0$  and (2) A is not reducible.

Call the matrix *A* fully reduced if either n = 1 or there exists a transformation by a permutation matrix *P* so that  $PAP^T$  has a decomposition in upper-block-diagonal form with only irreducible submatrices on the block-diagonal. Thus the lower-block-diagonal matrices are all zero. The particular form of a fully-reduced positive matrix is thus

$$A = P \begin{pmatrix} A_{11} & A_{12} & A_{13} & \dots & A_{1,n-1} & A_{1,n} \\ 0 & A_{22} & A_{23} & \dots & A_{2,n-1} & A_{2,n} \\ 0 & 0 & A_{33} & \dots & A_{3,n-1} & A_{3,n} \\ \vdots & \vdots & \ddots & \vdots & \vdots \\ 0 & 0 & \dots & A_{n-1,n-1} & A_{n-1,n} \\ 0 & 0 & \dots & 0 & A_{n,n} \end{pmatrix} P^{T},$$
(5.32)

where  $P \in \mathbb{R}^{n \times n}_+$  is a permutation matrix and the matrices on the block-diagonal of (5.32) are all irreducible positive matrices.

**Decompositions of Positive Matrices Based on Eigenvalues** Recall that for a matrix  $A \in \mathbb{R}^{n \times n}$  the *spectrum* is defined as the set of its eigenvalues and the *spectral radius* is defined as  $\rho(A) = \max_{\lambda \in \text{SpeC}(A)} |\lambda|$ . It follows from [9, Th. 1.3.2] that every positive matrix  $A \in \mathbb{R}^{n \times n}_+$  has at least one eigenvalue which equals its spectral radius.

**Definition 5.21.** [9, Def. 2.2.26]. Define for an integer  $n \in \mathbb{Z}_+$  and an irreducible positive matrix  $A \in \mathbb{R}^{n \times n}_+$ , the *index of cyclicity* of A as the number  $h \in \mathbb{Z}_+$  such that h equals the maximum number of distinct eigenvalues of A which are in modulus equal to the spectral radius  $\rho(A)$ . In mathematical notation:

$$h = \max\{k \in \mathbb{Z}_n \mid \forall i \in \mathbb{Z}_k, |\lambda_i(A)| = \rho(A)\}.$$
(5.33)

It follows from the comment above the previous definition that  $h \ge 1$ . If  $h \ge 2$  then one says that the matrix A is cyclic of index h.

**Definition 5.22.** Consider an integer  $n \in \mathbb{Z}_+$  and an irreducible matrix  $A \in \mathbb{R}^{n \times n}_+$ . Partition the set of eigenvalues into the following two subsets:  $\sigma^{\rho}(A)$ , which is the spectrum of A on the circle centered at origin with radius  $\rho(A)$ , and  $\sigma^-(A)$ , which is the spectrum of A strictly inside the disc centered at origin with radius  $\rho(A)$ . Hence,

$$\sigma^{\rho}(A) = \{\lambda \in \operatorname{spec}(A) \mid |\lambda(A)| = \rho(A)\},$$

$$\sigma^{-}(A) = \{\lambda \in \operatorname{spec}(A) \mid |\lambda(A)| < \rho(A)\}$$
(5.34)

with

spec(A) = 
$$\sigma^{\rho}(A) \cup \sigma^{-}(A), \ \sigma^{\rho}(A) \cap \sigma^{-}(A) = \emptyset;$$
  
 $n_1 = |\sigma^{\rho}(A)| = h, \ n_2 = |\sigma^{-}(A)| = n - n_1 = n - h$ 

In addition, there exists a nonsingular matrix  $S \in \mathbb{R}^{n \times n}$  such that the matrix  $S^{-1}AS$  is block-diagonal with,

$$S^{-1}AS = \text{block-diag}(A_1, A_2),$$
(5.35)  
$$A_1 \in \mathbb{R}^{n_1 \times n_1}, A_2 \in \mathbb{R}^{n_2 \times n_2}, \text{ spec}(A_1) = \sigma^{\rho}(A), \text{ spec}(A_2) = \sigma^{-}(A).$$

Finally, define the sets  $\sigma^{\rho}(A_2)$  and  $\sigma^{-}(A_2)$  in a similar manner with A being replaced by  $A_2$  in (5.34) and define the set  $\sigma^{0}(A) \subseteq \sigma^{\rho}(A_2)$  as the set of all eigenvalues of  $A_2$  whose whose polar angle is a rational multiple of  $2\pi$ .

The notation  $|\sigma^{\rho}(A)|$  denotes the number of elements of the indicated set. That the decomposition (5.35) is indeed a partition follows from Perron-Frobenius theorem [9, Th. 2.1.4, 2.2.20] and from the concept of spectral radius as the maximal value of the absolute values of all eigenvalues. In general, the matrices  $A_1$  and  $A_2$ depend on *S*. However, the relations (5.35) hold for any such *S*. When the matrices  $A_1$  and  $A_2$  are used in the body of the chapter, then these are characterized by their spectra. Also note that in contrary to  $\sigma^{\rho}(A)$ ,  $\sigma^{\rho}(A_2)$  can be empty set.

Next, we present the following lemma about the existence of a subset of eigenvalues that are among the  $(M \ h)$ -th root of unity for some  $M \in \mathbb{Z}_+$ . This lemma is used in the sequel for deriving the conditions on spec(A) for A to have a positive recursion.

*Lemma* 5.23. Consider the objects of Definition 5.22.

Then, there exists a minimal integer  $M \in \mathbb{Z}_+$  such that

$$\sigma_0(\mathbf{A}) \subseteq \left\{ \lambda \in \operatorname{spec}(\mathbf{A}_2) \middle| \lambda = \rho(\mathbf{A}_2) \exp(\frac{2\pi k}{Mh}i), k = 0, \dots, Mh - 1 \right\},$$
(5.36)

or, equivalently, there exists a minimal integer  $M \in \mathbb{Z}_+$  such that the eigenvalues of  $A_2/\rho(A_2)$  with unit modulus whose arguments are a rational multiple of  $2\pi$  are among the (M h)-th roots of unity.

*Proof of Lemma 5.23.* Let  $\delta^0$  be a set of  $n_{\delta^0} \in \mathbb{Z}_+$  members of  $\sigma^0$  with the property that the difference between the polar angle of no two members of  $\delta^0$  is an integer multiple of  $2\pi/h$ , or formally we define  $\delta^0 = \{\lambda_1, ..., \lambda_{n_{\delta^0}} \in \sigma^0 | \arg(\lambda_i) - \arg(\lambda_j) \neq 2z\pi/h, i \neq j, z \in \mathbb{Z}\}$ . For  $\lambda_j \in \delta^0$ ,  $j = 1, ..., n_{\delta^0}$ , let  $\arg(\lambda_j) = \frac{2\pi p_j}{q_j}$ . Define the sets  $\sigma_j^0 \subset \sigma^0$  for  $j = 1, ..., n_{\delta^0}$  as

$$\sigma_j^0 = \left\{ \lambda \in \operatorname{spec}(A_2) \middle| \lambda = \rho(A_2) \exp\left( (k/h + p_j/q_j) 2\pi i \right), \ k = 0, \dots, h-1 \right\},$$

or equivalently using the notation  $s_{j,k} \equiv kq_j + hp_j \pmod{hq_j}$ ,

$$\sigma_j^0 = \Big\{ \lambda \in \operatorname{spec}(A_2) \Big| \lambda = \rho(A_2) \exp\left(\frac{s_{j,k}}{hq_j} 2\pi i\right), \ k = 0, \dots, h-1 \Big\}.$$

It is clear that  $\sigma_1^0, ..., \sigma_{n_{\delta^0}}^0$  are mutually disjoint. In addition, since the eigenvalues of A are invariant under polar rotation of  $2k\pi/h$  for any  $k \in \mathbb{Z}$ , we have  $\sigma^0 = \bigcup_{j=1}^{n_{\delta^0}} \sigma_j^0$ . Noting that  $0 \le s_{j,k} \le hq_j - 1$  for k = 0, ..., h - 1 and for  $j = 1, ..., n_{\delta^0}$ , one observes that  $\sigma_0$  has the form proposed in (5.36) by choosing  $M = \operatorname{lcm}(q_1, ..., q_{n_{\delta^0}})$ .

It follows from [9, Th. 2.2.20] that if the matrix  $A \in \mathbb{R}^{n \times n}_+$  is irreducible and if A is of index of cyclicity  $h \ge 2$  then there exists a permutation matrix  $P \in \mathbb{R}^{n \times n}$  and matrices  $\{A_{i,i+1} \in \mathbb{R}^{n_i \times n_{i+1}}_+, i = 0, 1, ..., h - 1 \pmod{h}\}$  such that,

$$\sum_{i=0}^{h-1} n_i = n,$$

$$A = P \begin{pmatrix} 0 & A_{1,2} & 0 & \dots & 0 & 0 \\ 0 & 0 & A_{2,3} & \dots & 0 & 0 \\ \vdots & 0 & \ddots & A_{h-2,h-1} & 0 \\ 0 & 0 & 0 & \dots & 0 & A_{h-1,h} \\ A_{h,1} & 0 & 0 & \dots & 0 & 0 \end{pmatrix} P^T, \quad (5.37)$$
with groups disconstructed blocks

with square diagonal blocks.

One then says that the positive matrix *A* is *cogredient* to the block matrix of equation (5.37); [9, Def. 2.1.2].

The irreducible positive matrix  $A \in \mathbb{R}^{n \times n}_+$  is called *primitive* if its trace is strictly positive; see [9, Def. 2.1.8, Cor. 2.2.28].

It follows from the proof of [9, Th. 2.2.30] that if the matrix  $A \in \mathbb{R}^{n \times n}_+$  is irreducible and if A is of index of cyclicity  $h \ge 2$  then there exists a permutation matrix  $P \in \mathbb{R}^{n \times n}$  such that

$$A^{h} = P\begin{pmatrix} C_{1,1} & 0 & 0 & \dots & 0 & 0 \\ 0 & C_{2,2} & 0 & \dots & 0 & 0 \\ \vdots & \dots & \vdots & \ddots & 0 & 0 \\ 0 & 0 & 0 & \dots & C_{h-1,h-1} & 0 \\ 0 & 0 & 0 & \dots & 0 & C_{h,h} \end{pmatrix} P^{T},$$
  
$$\forall i \in \mathbb{Z}_{h}, \ C_{i,i} \in \mathbb{R}^{n_{i} \times n_{i}}_{+} \text{ are primitive matrices with } \rho(C_{i,i}) = \rho^{h}(A);$$
  
$$\sum_{i=1}^{h} n_{i} = n.$$

Sources for the above theory are not only [9] but also the book [31, Ch. 3].

#### **Limits of Powers of Positive Matrices**

It follows from Theorem [9, Th. 2.4.1] that for a primitive irreducible matrix  $A \in \mathbb{R}^{n \times n}_+$ , the following limit exists:

$$\lim_{k\to\infty}\left(\frac{A}{\rho(A)}\right)^k\in\mathbb{R}^{n\times n}_+.$$

Next the above results can be combined. Consider an irreducible matrix  $A \in \mathbb{R}^{n \times n}_+$ . Assume that the index of cyclity of A is such that  $h \ge 2$ . It then follows from the above that  $A^h$  is cogredient to a block diagonal matrix with on the diagonal primitive irreducible matrices. From the above existence of the limit then follows that,

$$A^{h} = P\begin{pmatrix} C_{1,1} & 0 & 0 & \dots & 0 & 0 \\ 0 & C_{2,2} & 0 & \dots & 0 & 0 \\ \vdots & \dots & \vdots & \ddots & 0 & 0 \\ 0 & 0 & 0 & \dots & C_{h-1,h-1} & 0 \\ 0 & 0 & 0 & \dots & 0 & C_{h,h} \end{pmatrix} P^{T},$$
  
$$\lim_{k \to \infty} \left(\frac{A^{h}}{\rho^{h}(A)}\right)^{k} = P \operatorname{block-diag}\left(\lim_{k \to \infty} (C_{1,1}/\rho^{k}(C_{1,1}), \dots, \lim_{k \to \infty} (C_{h,h}/\rho^{k}(C_{h,h})\right) P^{T}$$
$$= P \operatorname{block-diag}(C_{\infty,1,1}, \dots, C_{\infty,h,h}) P^{T} \in \mathbb{R}^{n \times n}_{+}.$$

Next, we introduce the following lemma characterizing the limit behavior of conmat<sub>k</sub>(A, b) as  $k \to \infty$ , which is used for characterizing the infinite-time reachable subset Reachset<sub> $\infty$ </sub>(A, b).

**Definition 5.24.** Let the positive matrix  $A \in \mathbb{R}^{n \times n}_+$  be irreducible with index of cyclicity h with  $1 \le h \le n$  and let  $b \in \mathbb{R}^m_+$ . Define the matrices and the *limit cone* 

according to

$$\forall i \in \{0, \dots, h-1\}, \quad \boldsymbol{A}_{f,i} = \lim_{k \to \infty} \left( \left( \frac{\boldsymbol{A}}{\rho(\boldsymbol{A})} \right)^h \right)^k \boldsymbol{A}^i,$$
$$C_{\lim} = \operatorname{cone}([\boldsymbol{A}_{f,0}\boldsymbol{b} \ \dots \ \boldsymbol{A}_{f,h-1}\boldsymbol{b}]).$$
$$\forall i \in \{0, \dots, h-1\}, \quad \boldsymbol{A}_{f,i} = \lim_{k \to \infty} \left( \left( \frac{\boldsymbol{A}}{\rho(\boldsymbol{A})} \right)^h \right)^k \boldsymbol{A}^i,$$
$$C_{\lim} = \operatorname{cone}([\boldsymbol{A}_{f,0}\boldsymbol{b} \ \dots \ \boldsymbol{A}_{f,h-1}\boldsymbol{b}]),$$

Define for i = 0, ..., h - 1 the positive eigen vectors  $v_{f,i} \in \mathbb{R}^n_+$  of the h distinct eigenvalues of the matrix  $A^h$  associated with the Perron root of  $\rho^h(A)$ ; thus,

$$A^h v_{f,i} = \rho^h(A) v_{f,i}$$

Lemma 5.25. Consider the objects of Definition 5.24. Then the limit cone satisfies

$$\mathcal{C}_{\lim} \subseteq \operatorname{cone}([v_{f,0} \dots v_{f,h-1}]).$$

*Proof of Lemma 5.25.* Since *A* is irreducible, there exists a monomial matrix  $S \in \mathbb{R}^{n \times n}_+$  [9] such that

$$\hat{A} = S^{\mathrm{T}} A S = \begin{bmatrix} 0_{n_{1}} & A_{1} & 0 & \dots & 0 \\ \vdots & \ddots & \ddots & \ddots & \vdots \\ \vdots & \ddots & \ddots & 0 \\ 0 & \dots & \dots & 0_{n_{h-1}} & A_{h-1} \\ A_{h} & 0 & \dots & \dots & 0_{n_{h}} \end{bmatrix}, \ \hat{B} = S^{\mathrm{T}} b$$

where  $0_{n_i} \in \mathbb{R}^{n_i \times n_i}$ ,  $i \in \mathbb{N}$  are square blocks with  $\sum_{i=1}^n n_i = n$ , and where  $A_i$  has no

zero rows or columns with  $L_1 = \prod_{i=1}^{n} A_i$  being an irreducible matrix. Then we have

$$\hat{A}^{h} = \text{diag}(L_1, ..., L_h)$$
, where  $L_k = \prod_{i=k}^{h} A_i \prod_{j=1}^{\text{mod}(h+k-1,h)} A_j$  is a primitive matrix of dimension

 $n_k \times n_k$  with Perron root  $\rho^h(A)$ . Define the matrix  $\hat{A}_{f,i} = \lim_{p \to \infty} \frac{A^{ph}}{\rho^{ph}} \hat{A}^i$  for i = 0, ..., h - 1. Since  $L_i$ , i = 1, ..., h is primitive, it follows from [9] that

$$\hat{A}_{f,0} = \begin{bmatrix} \boldsymbol{x}_1^1 & \dots & \boldsymbol{x}_1^{n_1} & 0 & 0 & 0 & 0 & \dots & 0 \\ 0 & \dots & 0 & \boldsymbol{x}_2^1 & \dots & \boldsymbol{x}_2^{n_2} & 0 & \dots & 0 \\ 0 & \dots & 0 & 0 & \dots & 0 & 0 & \dots & 0 \\ \vdots & \vdots \\ 0 & 0 & 0 & 0 & 0 & 0 & \boldsymbol{x}_h^1 & \dots & \boldsymbol{x}_h^{n_h} \end{bmatrix},$$

where  $x_i^k = c_i^k x_i$  with  $c_i^k$ ,  $k = 1, ..., n_i$ , being some positive scalars and with  $x_i \in \mathbb{R}_{s+}^{n_i \times n_i}$  being the Frobenius eigenvector of  $L_i$ . Note that due to the block structure of  $\hat{A}$ ,  $\hat{A}_{f,i}$  retains the same structure as  $\hat{A}_{f,0}$  up to a scaled permutation of its columns for i = 1, ..., h - 1. Hence, we have  $\hat{A}_{f,i}\hat{b} \in \text{cone}(C)$ , where

$$\boldsymbol{C} = \begin{bmatrix} \boldsymbol{x}_1 & 0 & \dots & 0 \\ 0 & \boldsymbol{x}_2 & \dots & 0 \\ 0 & 0 & \dots & 0 \\ \vdots & \vdots & \dots & \vdots \\ 0 & \dots & 0 & \boldsymbol{x}_h \end{bmatrix}.$$

In the original coordinates, we have  $A_{f,i}b \in \operatorname{cone}(SC)$ . Clearly, since the columns of C are the positive eigenvectors of  $\hat{A}^h$  and since S is monomial, we have  $SC = [v_{f,0} \dots v_{f,h-1}]$ , where  $v_{f,i} \in \mathbb{R}^{n \times n}_+$  is the (i + 1)-th positive eigenvector of  $A^h$  for  $i = 0, \dots, h-1$ . This proves that  $\operatorname{cone}([A_{f,0}b \dots A_{f,h-1}b]) \subseteq \operatorname{cone}([v_{f,0} \dots v_{f,h-1}])$ .  $\Box$ 

## **5.B.** Note on Theorem 5.12

After the publication of our work [1], upon which the current chapter is based, it was brought to the authors' attention that the formulation of Theorem 5.12 [46, Th. 5] (specifically statement C4) is incorrect. Theorem 5.12 concerns the characterization of polynomials admitting a nonnegative recursion that is used in the proofs of Theorems 5.13 and 5.15 of the current chapter. A counter example to Theorem 5.12 is provided by L. Farina and L. Benvenuti in [57], where they also state the correct formulation of the given theorem. The correct formulation of the theorem is reported below for convenience.

**Theorem 5.26.** [57, Th. 1] Let  $\{a_1, ..., a_k\}$  be given complex numbers, and let P(x) be the polynomial  $x^k - a_1 x^{k-l} - \cdots - a_k$ . Then conditions (A), (B) and (C) below are equivalent:

- (A) Any infinite sequence  $(u_n)_{n\geq 0}$  of complex numbers which satisfies the recursion  $u_{n+k} = a_1 u_{n+k-1} + a_2 u_{n+k-2} + \dots + a_k u_n$  for  $n \geq 0$ , also satisfies a recursion with positive coefficients.
- (B) The polynomial P(x) divides a polynomial in Q.
- (C) In case the polynomial *P*(*x*) has a positive root *r*, then all conditions (1)-(4) below are satisfied:
  - (C1)  $r \ge |\alpha|$  for any root  $\alpha$  of P(x).
  - (C2) if  $\alpha = r$  for some root  $\alpha$  of P(x), then  $\alpha/r$  is a root of unity.
  - (C3) all roots P(x) with absolute value r are simple.
  - (C4) if *d* is the minimal integer such that  $\epsilon^d = 1$  for all roots of unity  $\epsilon$  which satisfy  $P(r\epsilon) = 0$ , then P(x) has no roots of the form  $s\omega$  where 0 < s < r and  $\omega^d = 1$ .

The difference between the two formulations of statement C4 is that in Theorem 5.12, due to M. Roitman and Z. Rubinstein [46], statement C4 can hold with different values of integer k for different groups of roots with absolute value r, while in Theorem 5.26, due to L. Benvenuti and L. Farina [57], statement C4 must hold with the same value of the integer k (m in their own words) for all roots with absolute value r. It is proved in [57] that the latter formulation of C4 must be used to conclude equivalence between statements A, B, and C of the theorem, which we need for Theorems 5.13 and 5.15.

We investigated whether this change has any consequences for Theorems 5.13 and 5.15, and concluded that conditions laid out by these theorems are not affected by using Theorem 5.26 as the basis for our proofs. Note that the proof of Theorem 5.15 is based on C4 of Theorem 5.12 with k = h. In this case k cannot take any other value than h as A has exactly h eigenvalues of the form

$$\rho(\mathbf{A})\exp\left(\frac{2\pi m}{h}i\right), \ m=0,\ldots,h-1.$$

In the case of Theorem 5.13, C4 of Theorem 5.12 must hold with the smallest integer M (see Lemma 5.23) such that

$$\sigma^{\rho}(\boldsymbol{A}_{2}) \subseteq \left\{ \lambda \middle| \lambda = \rho(\boldsymbol{A}_{2}) \exp\left(\frac{2\pi k}{Mh}i\right), k = 0, \dots, Mh - 1 \right\},\$$

where  $\sigma^{\rho}(A_2)$  is the set of *all* eigenvalues of  $A_2$  with modulus  $\rho(A_2)$ . Therefore, our proof is based on a *single* value for k (k = M) for *all* eigenvalues of  $A_2$  with modulus  $\rho(A_2)$ .

# 6

# Conclusions and Further Research

"I would rather have questions that can't be answered than answers that can't be questioned."

Richard Phillips Feynman, 1918 – 1988

# **6.1.** Concluding Remarks

This thesis has addressed a range of topics on model-based control of baggage handling systems and control of discrete-time linear positive systems, where the results derived for control of positive systems have been subsequently used to develop a scalable tube-based model predictive control (MPC) scheme for large-scale baggage handling systems (BHSs). In short, the thesis has made contributions along the following main directions:

- 1) Development of a flow-based modeling framework for large-scale logistic systems such as BHSs, and scalable MPC-based control strategies for optimal operation of such systems subject to disturbances.
- 2) Computation of a robustly positively invariant hypercube for discrete-time linear positive systems subject to additive constraints together with computation of the state feedback controller that minimizes the output  $L_{\infty}$  norm of the output over the hypercube.
- Characterization of reachable subsets for discrete-time linear positive systems, and formulation of the reachability problem of a polyhedral subset of the positive orthant.

We will now elaborate on the contributions and the results.

## 6.1.1. Robust Model-based Control of Baggage Handling Systems

A modeling and control design framework for large-scale BHSs has been proposed that is based on approximating movements of pieces of baggage by continuous flows, rather than describing their individual movements. This was necessary to arrive at models suitable for developing model-based control methods discussed in the thesis. The developed models were then used within the MPC framework as MPC enables one to incorporate (partial) knowledge of the disturbance (i.e., the baggage demand), capacity and flow constraints, and a measure of performance in an integrated control design approach.

We have shown that the original nonlinear programming (NLP)-based MPC formulation, which is not scalable to large instances of BHSs or to longer prediction horizons, can be expressed as a linear programming (LP)-based MPC problem or a iterative linear programming (ILP)-based MPC problem, which leads to tractable optimization problems for large BHSs. Especially ILP-based MPC has been shown to achieve accurate predictions in relation to NLP-based MPC, hence, making it possible to use longer prediction horizons. Therefore, this approach enables one to fully benefit from knowledge of the future demand profiles (due to the use of large prediction horizons), while scaling to large BHSs. The performance of the developed MPC schemes has been illustrated by a simulation-based case study. It has been shown that, with fully predictable baggage demand at the loading stations, the ILP-based MPC approach outperforms the current state-of-the-art heuristics-based approach at an affordable computational cost. In addition, the LP-based MPC approach also outperforms the current state-of-the-art approach for short prediction horizons while offering the cheapest solution in terms of the required computational resources.

The MPC solution has then been extended to the cases where only a partial knowledge of the future baggage demand is known. The uncertainty in the baggage demand has been modeled by expressing the baggage demand as the sum of a nominal fully predictable component and a relatively smaller zero-mean random component. To provide robustness against the additive uncertainty, a tube-based control scheme has been employed, which is a low-complexity robust MPC approach consisting of a top-level MPC controller for the nominal system and a bottom-level state feedback controller for the error system. The error system describes the difference between the uncertain system (i.e., the actual system) and the nominal system. We have investigated two different approaches, as the subject of two separate chapters of this thesis, to the design of the state feedback controller along with the appropriately modified version of the nominal constraints so that the nominal MPC problem is still feasible in presence of the error dynamics. Such modifications commonly entail restricting the nominal MPC state and control constraints by the minimal robust invariant set and its image on the control subspace, the computation of which is very difficult in practice for large-scale systems. To make our approach applicable to large-scale systems, we have proposed a method to calculate the smallest robustly positively invariant ellipsoids and hypercubes that include the minimal robust invariant set, and we use these sets to tighten the nominal MPC

#### constraints.

For the first approach, a feedback controller minimizing the induced  $L_2$  gain of the closed-loop error system from the disturbance input to the error output has been developed. The problem of searching for the state feedback gain as well as the associated robustly positively invariant ellipsoid has been formulated as an optimization problem with a linear objective function subject to linear matrix inequalities (LMIs). Furthermore, to impose state and control constraints on the error system, another formulation of this problem has been developed that allows for specification of control and state constraints via suitable additional LMIs.

To develop the second approach, we have first developed some theoretical background for linear positive systems subject to additive disturbance, and we have shown that, for positive systems, the existence of a robustly positively invariant set is characterized by a set of linear inequalities. This result is then exploited to design the state feedback controller for the error system such that the resulting closed-loop system is positive and optimal in the sense of the  $L_{\infty}$  norm of the disturbance-driven output. We have formulated the joint problem of searching for such feedback gain and the corresponding robustly positively invariant hypercube as a linear program. We have also extended the solution to include state and control constraints for the closed-loop error system, and a minimum size for the desired robustly positively invariant hypercube.

Systems of BHS nature are often subject to capacity constraints, which are represented by upper and lower bound on the states and the control inputs. The conservatism introduced by ellipsoidal approximation of such constraints, as needed in the first approach, may lead to an infeasible set of LMIs for large-scale systems. Further to this point, linear programs can generally be solved more efficiently than linear semi-definite optimization problems subject to LMIs. Hence, in contrast to the first approach, which is only suitable for small-scale problems, the latter approach to feedback design scales well for large-scale systems. In addition, the second approach allows us to use the  $L_{\infty}$  norm of the output as the criterion for feedback gain design, which provides a better measure of the worst-case performance of the system than the  $L_2$ -gain of the system for systems that are subject to additive disturbances characterized by their minimum and maximum values. Finally, the effectiveness of the latter state feedback design approach has been illustrated in a large-scale BHS case study, where a tube-based MPC controller is designed to deal with structured uncertainty of the baggage demand. In this case study, we have also shown how recursive feasibility and asymptotic stability of the nominal system can be guaranteed by the appropriate choice of the terminal cost function and the terminal constraint set.

Models and methods developed in Chapters 2- 4 can in principle be applied to vehicle-based sorting systems, where the feed sequence can be controlled and there is sufficient buffer capacity. However, their applicability to heavily loaded conveyor-based sorting systems may be limited since such systems cannot be heavily buffered and there is usually no control over the feed sequence.

## 6.1.2. Reachability Problem for Discrete-time Linear Positive Systems

For discrete-time linear positive systems, we have shown that the reachable subset from the origin using nonnegative controls and its closure can be either a polyhedral cone (i.e., generated by a finite number of extremal rays) or a nonpolyhedral cone depending on system matrices. Whether the reachable subset or its closure is polyhedral is closely linked with the system matrix A, or an appropriately decomposed version of it, admitting a positive recursion; that is there exist a positive integer mand non-negative real coefficients  $c_i$ , i = 0, ..., m - 1 such that  $A^m = \sum_{i=0}^{m-1} c_i A^i$ . We have derived necessary and sufficient conditions in terms of the spectrum of the system matrix A for polyhedrality of the reachable subset and its closure.

Polyhedrality of the reachable subset or its closure helps to define an alternative formulation for reachability of discrete-time linear positive systems, the one which concerns whether a subset of the positive orthant can be reached from the origin using a finite sequence of nonnegative controls or can be arbitrarily closely approached from the origin using a sufficiently long nonnegative control sequence. This definition of the reachability problem is more useful from an application point of view, and is in contrast to the classical definition, which is concerned with the reachability of the entire positive orthant. This is owed to the fact that in practice one is often interested in reachability of a certain "target" subset of the positive orthant rather than the entire positive orthant.

For cases where the reachable subset or its closure is a polyhedral cone, we have presented a method to determine whether a given target subset of the positive orthant can be reached from the origin. In case the reachable subset is polyhedral, the developed method can also be used to calculate the finite nonnegative control sequence driving the system from the origin to any point in the target set. The applicability of our proposed approach is illustrated by several examples.

# **6.2. Recommendations for Further Research**

# 6.2.1. Extensions of Models and Methods Developed for BHSs

The models developed in Chapter 2 for BHSs can in principle be extended to cover other applications of a similar nature such as logistic systems and material handling systems. In this regard it is worthwhile to explore whether alternative ways of modeling density-dependent flow travel times through the network, for example by including the effect of network congestion on travel times, yield models of higher accuracy. Furthermore, in addition to robustness against variations in the demand profile, which was the subject of Chapters 3 and 4, fault tolerance and robustness against model parameters should be integrated in the model-based control design process so that the closed-loop system is robust against model mismatch. Among others, model mismatch can arise from broken components or capacity drop due to closure of a part of the network. Finally, this work has considered a centralized control design with full information of the system. For large-scale systems such as BHSs the centralized controller can be both unreliable, as the control system has a

single point of failure, and computationally expensive. Hence, one is motivated to investigate distributed or decentralized robust MPC solutions for such systems.

In this research work, we have addressed high-level model-based control design of large-scale logistic systems. In practice, the decisions made by the high-level controller in terms of the link-to-link transfer rate of material need to be translated to the number of carts, boxes, DCVs, etc. that need to be dispatched from one link to another in a given period of time. The effect of the conversion of the flow variables, generated by the high-level controller, to the discrete actions of lower level controllers should be taken into account when assessing the performance of the control system. To this end, one must develop a corresponding quantity-based *plant* model to asses the performance of the MPC controller designed for the flowbased model. This amounts to employing an elaborate plant model and a simplified control model. In this way, the effect of *realization of actuation* (i.e., translating flows to movements of DCVs) on the performance of the control system can be studied.

The models developed for BHSs in this thesis, and possibly their extensions, can be used in feasibility studies of new BHSs. During the pre-design process and as a part of cost and benefit analysis, extensive simulations based on BHS models can be used to determine maximum system throughput for a given capacity, expected time delays due to baggage transfer, robustness of the system in the event of faults, etc. In addition, for existing BHSs, such model-based simulations can help optimize the system operation along user-defined Key Performance Indicators (KPIs) to maximize utilization of the existing infrastructure as efficiently as possible. The use of modelbased simulations as an instrumental part of planing new BHSs or for upgrading existing ones can ultimately lead to sustainable BHSs that are not only efficiently designed, but are also well predictable in terms of maintenance, future costs, and expected profits.

In this thesis, we have focused on modern automated BHSs with respect to modeling, control design, and performance analysis. In doing so, we have omitted the human factor, which has a significant impact on the overall quality of baggage handling service. In fact, the bottleneck in baggage handling is often in the manual handling part. Think of baggage not being supplied in time or removed in timely manner at the unloading stations. Delays made in manual processes can easily propagate and adversely impact the overall quality of the baggage handling service. A very interesting research avenue is, therefore, to analyze the baggage handling process in big airports not in terms of technical capabilities of automated BHSs, nor the control algorithms, but in terms of the human factors. In this regard, one can develop (stochastic) models to determine the effect of human-induced delays on the performance of the BHSs, and include those models in overall control design for BHSs such that the designed control strategy is ultimately sufficiently robust against human-induced delays.

## 6.2.2. Extensions of the Control Theory of Linear Positive Systems

The work in this thesis on polyhedrality of finite-time and infinite-time reachable subsets for linear positive systems only concernedd single-input systems with irreducible system matrix A. To establish a full picture of the reachable subsets, extension of these results to the multiple-input case needs to be investigated. Further generalization of such results should allow to consider systems with a reducible system matrix. In addition, in absence of any structure for the input vector b, the conditions derived in Chapter 5 depend only on the spectrum of the system matrix A. It is worthwhile to investigate whether an assumed structure for B would lead to different conditions for polyhedrality of the reachable subsets. Finally, to obtain a complete picture of the controllability problem when a subset of the positive orthant is considered as the target set, it is necessary to consider the case where the initial state is not the origin, but an arbitrary point in the positive orthant, and to derive the conditions under which the finite-time controllable subsets are polyhedral.

In Chapter 5, we have addressed the problem of determining whether the reachable subset is a polyhedral cone or a nonpolyhedral cone. In addition to control theory, this problem is of interest to the theory of realization and of system identification. The choice for the reachable subset from the origin is essential for realization theory as observability of a linear positive system is then of interest only for states in the reachable set. A characterization of that view of observability does currently not exist in the literature. The condition formulated in [1] is too strong because it is based on the assumption that the reachable set from the origin is the entire positive vector space  $\mathbb{R}^n_+$ . Developing necessary and sufficient conditions for the described view of observability (i.e., observability of only a subset of the positive orthant) will then pave the way towards a better understanding of the realization problem for linear positive systems.

## 6.2.3. Other Application Areas of Control Theory of Linear Positive Systems

Recursive feasibility of the optimization problem and asymptotic stability of the closed-loop system are key properties of any MPC design. Although these properties can be essentially achieved by choosing a sufficiently large prediction horizon, enforcing them explicitly by choosing an appropriate terminal constraint set and a terminal cost function is essential when the prediction horizon cannot be chosen arbitrarily large due to computational concerns. The common solution approach of calculating a (robustly) positively invariant subspace for linear systems requires expensive iterative calculation, which quickly becomes prohibitive for systems with a large number of states and inputs. The results of Chapter 4 on calculating (robustly) positively invariant sets for linear positive systems and the associated state feedback control law defined over those sets can be further developed to calculate appropriate terminal constraint set and a terminal cost function for MPC of linear systems. This approach may be of particular interest in the case of parametric MPC, where the control sequence over a prediction horizon *N* is parameterized as  $u_N(t) = (K(t)x(t), ..., K(t+N-1)x(t+N-1))$ , state and control constraints are

expressed by ∞-norm bounded sets, and the MPC objective function is of the form  $V_N(\mathbf{x}) = \sum_{i \in \mathbb{N}_{0:N-1}} \|\mathbf{P}^{-1}\mathbf{x}(i)\|_{\infty} + \|\mathbf{U}^{-1}\mathbf{u}(i)\|_{\infty} + \|\mathbf{Q}^{-1}\mathbf{x}(N)\|_{\infty}$ , with  $\mathbf{P} > 0$ ,  $\mathbf{U} > 0$ , and  $\mathbf{Q} > 0$  being diagonal matrices. The methods developed in Chapter 4 may then be extended to find a (robustly) positively invariant hypercube and the corresponding Lyapunov function that can be used as the terminal constraint set and the terminal cost function to guarantee the recursive feasibility of the MPC problem and asymptotic stability of the closed-loop system. This approach is expected to yield a linear programming formulation for both calculating the (robustly) invariant hypercube and calculating the optimal control sequence. In addition, developing similar results as those of Chapter 4 for the case where the disturbance is characterized by a 1-norm bounded set of the form  $\mathbb{W} = \{ \boldsymbol{\omega} \mid \| \boldsymbol{\Omega}^{-1} \boldsymbol{\omega} \|_1 \leq \boldsymbol{\omega}_{\max} \}$ , with a diagonal  $\Omega > 0$ , should be also considered. In this case, one should investigate whether the problem of searching for a feedback controller that minimizes the  $L_1$ -norm of the disturbance driven output and the associated robustly positively invariant set can be expressed as a linear program. In addition, the geometry of such set and the associated Lyapunov function defined over such set is of interest. In a similar manner to what we discussed above, such results could also be useful for the MPC application, where the aim is to design a receding horizon finite horizon controller minimizing a suitably defined  $L_1$ -norm based cost function subject to  $L_1$ -norm based constraints.

Application of methods developed based on the theory of positive dynamical systems, such as those developed in Chapter 4, is not restricted to the classical examples of positive systems such BHSs, logistic systems, economic systems, bioreactors, pharmacokinetic systems, etc. In fact, the result of Chapter 4 on control synthesis and other similar results for control of linear positive systems only require the closed-loop system to be positive, and neither the control input, the disturbance input, nor the initial state is required to be nonnegative. Hence, other major application fields such as motion control, robotics, process control, etc., where the open-loop system is not necessarily positive, can also benefit from the lower-complexity control design solutions and analysis methods developed for linear positive systems as long as the feedback control is designed in such a manner that the closed-loop system positive. Exploring the relevance and application of methods developed for linear positive systems to a broader application domain is therefore a topic worth more attention.

Many of the well-known control design problems take a simpler form for linear positive systems. For example, it has been shown in [2–4] that the KYP lemma for linear positive systems can be expressed in terms of a diagonal matrix variable rather than a general symmetric matrix, which drastically simplifies the  $H_{\infty}$ -design problem. Similarly, in [5–8], it has been shown that stability analysis of an interconnection of systems based on  $L_1$ -gain and  $L_{\infty}$ -gain criteria and the corresponding controller synthesis problems can be expressed as linear programs. In addition, we have shown in Chapter 4 that the joint problem of computing the smallest robustly invariant hypercube containing the minimal robust invariant subset and computing the corresponding state feedback controller minimizing the  $L_{\infty}$ -norm of the disturbance driven output over the said hypercube can be expressed as a linear problem.
While the mentioned works make contributions to analysis of feedback systems and controller synthesis for linear positive systems, potential application of the theory nonnegative matrices and positive dynamical systems in other domains of control theory such as optimal and receding horizon observer design problems, (distributed) parameter estimation problems, and joint state and parameter estimation problems has been mostly unexplored. This is an interesting research area, which could yield promising results as such problems may also benefit from simpler solution forms due to the properties of linear positive systems.

#### References

- H. M. Härdin and J. H. van Schuppen, Observers for linear positive systems, Linear Algebra and its Applications 425, 571 (2007).
- [2] T. Tanaka and C. Langbort, The bounded real lemma for internally positive systems and h-infinity structured static state feedback, IEEE Transactions on Automatic Control 56, 2218 (2011).
- [3] F. Najson, On the Kalman–Yakubovich–Popov lemma for discrete-time positive linear systems: a novel simple proof and some related results, International Journal of Control 86, 1813 (2013).
- [4] A. Rantzer, On the Kalman-Yakubovich-Popov lemma for positive systems, IEEE Transactions on Automatic Control 61, 1346 (2016).
- [5] Y. Ebihara, D. Peaucelle, and D. Arzelier, Optimal L<sub>1</sub>-controller synthesis for positive systems and its robustness properties, in 2012 American Control Conference (ACC) (2012) pp. 5992–5997.
- [6] C. Briat, Robust stability and stabilization of uncertain linear positive systems via integral linear constraints:  $L_1$ -gain and  $L_{\infty}$ -gain characterization, International Journal of Robust and Nonlinear Control **23**, 1932 (2013).
- [7] A. Rantzer, Scalable control of positive systems, European Journal of Control 24, 72 (2015), sI: ECC15.
- [8] J. Shen and J. Lam, Static output-feedback stabilization with optimal L<sub>1</sub>-gain for positive linear systems, Automatica 63, 248 (2016).

## Summary

Large-scale baggage handling systems, or large-scale logistic system, for that matter, pose interesting challenges to model-based control design. These challenges concern computational complexity, scalability, and robustness of the proposed solutions. This thesis tackles these issues in a collection of papers organized in two overlapping parts. The first part concerns modeling and Model Predictive Control (MPC) design of large-scale baggage handling systems (BHSs), where a modeling framework for BHSs is proposed that is subsequently used to develop an MPC scheme for control of large-scale BHSs. The MPC controller optimizes for the timely arrival of pieces of baggage at their destination within the BHS network under capacity constraints while minimizing the overall cost of transporting pieces of baggage. Several formulations for the resulting constrained optimization problem are proposed, and they are compared with each other in terms of closed-loop performance and computational complexity. It is shown, via simulation studies, that the proposed solutions can outperform a heuristics-based approach commonly used for control of BHSs while scaling well to larger BHS network instances.

In its second part, the thesis focuses on robustness of control design in the face of a partially known disturbance input (i.e., input baggage demand), and especially on developing a scalable tube-based MPC scheme. For this purpose, considering the BHS model essentially as a linear positive system, a linear-programming-based approach is proposed for the joint calculation of a robustly positively invariant subset and a constrained state feedback controller that minimizes the disturbance-driven  $L_{\infty}$  norm of the output over this set. A tube-based MPC control scheme is finally developed by coupling the state feedback controller with a nominal MPC controller, guaranteeing recursive feasibility and asymptotic stability. It is shown via simulation studies that the proposed tube-based approach is effective against unpredictable disturbances. In addition, since the design of both the nominal MPC controller and the state feedback controller involves only linear programs, the proposed tubebased approach scales well to BHS networks of larger size.

Linear positive systems are of interest in several branches of engineering, logistics, biochemistry, and economics. As a spin-off topic and inspired by the applications of the theory of linear positive systems to modeling and control design of systems in the mentioned domains, the third part of the thesis focuses on the reachability analysis of discrete-time linear positive systems. More specifically, we revisit the problem of characterizing the subset of the state space that is reachable from the origin for discrete-time linear positive systems. This problem is of interest in topics such as optimal control of linear positive systems and realization theory of linear positive systems. It is established in this thesis that the reachable subset can be either a polyhedral or a nonpolyhedral cone. For the single-input case, a characterization is provided of when the infinite-time and the finite-time reachable subsets are polyhedral. Finally, for the case of polyhedral reachable subsets, a method, based on solving a set of linear equations, is provided to verify whether a target set can be reached from the origin using positive inputs.

-

# **List of Publications**

- 7. Y. Zeinaly, B. De Schutter, *Robustly Positively Invariant Sets for Discrete-time Linear Positive Systems: Application to a Tube-based MPC Approach*, To be submitted to IEEE Transactions on Control Systems Technology, 2022.
- Y. Zeinaly, J. H. van Schuppen, B. De Schutter, *Linear Positive Systems May Have a Reachable Subset from the Origin That is Either Polyhedral or Nonpolyhedral*, SIAM Journal on Matrix Analysis and Applications **41**, 1 (2020), pp. 279–307.
- Y. Zeinaly, B. De Schutter, H. Hellendoorn, An Integrated Model Predictive Scheme for Baggage Handling Systems: Routing, Line Balancing, and Empty-Cart Management, IEEE Transactions on Control Systems Technology 23, 4 (2015), pp. 1536–1545.
- Y. Zeinaly, B. De Schutter, H. Hellendoorn, *Towards a Robust Multi-level Control Approach for Baggage Handling Systems*, in 52nd IEEE Conference on Decision and Control (CDC), (2013), pp. 281–287.
- Y. Zeinaly, B. De Schutter, H. Hellendoorn, An MPC Scheme For Routing Problem In Baggage Handling Systems: A Linear Programming Approach, in 10th IEEE International Conference On Networking, Sensing and Control (ICNSC), (2013), pp. 786–791.
- Y. Zeinaly, B. De Schutter, H. Hellendoorn, A Model Predictive Approach For Baggage Handling Systems, in 16th International IEEE Conference on Intelligent Transportation Systems (ITSC), (2013), pp. 687–693.
- Y. Zeinaly, B. De Schutter, H. Hellendoorn, A Model Predictive Control Approach For The Line Balancing In Baggage Handling Systems, IFAC Proceedings Volumes 45, 24 (2012), pp. 215–220.

## **Curriculum Vitæ**

Yashar Zeinaly was born in April 1985 in Urmia, Iran. In August 2007, he obtained his B.Sc. degree in electrical engineering with a specialization in control engineering from Khajeh Nasir Toosi University of Technology, Tehran. Iran. In 2008, he was admitted to the master program Systems, Control & Mechatronics at Chalmers University of Technology, Gothenburg, Sweden. He was a visiting researcher at the University of Auckland, New Zealand, working on his master thesis on model predictive direct torque control of medium voltage drives, which was a project supported by ABB Switzerland. In March 2011, he received the best presentation award at the 2011 Applied Power Electronics Conference (APEC), held in Fort Worth, Texas, for the work he conducted for his master thesis.

He obtained his M.Sc. degree in December 2010, and subsequently joined Delft Center of Systems and Control as a PhD candidate under the supervision of Prof. Bart De Schutter, working on the topic "Model-based Control of Large-scale Logistic System". During his first year of PhD he successfully fulfilled the requirements to be awarded the Dutch Institute of Systems and Control (DISC) certificate from the DISC graduate school. The focus of his PhD was on robust scalable model-based control solutions for large-scale baggage handling system (BHSs), where he used tailor-made methods from the control theory of linear positive systems to develop a low-complexity robust model predictive control (MPC) approaches for applications such as large-scale BHSs. In addition, he worked with Prof. Jan van Schuppen of the Mathematics Department of the Delft University of Technology on characterizing the reachable subset of discrete-time linear positive systems.

He is currently a research engineer within the thermotechnology division of Robert Bosch GmbH. His research interest are modeling and low-complexity modelbased control of networked smart energy systems, control theory of linear positive systems and its application in scalable model-based control design of complex systems.

## **TRAIL Thesis Series**

The following list contains the most recent dissertations in the TRAIL Thesis Series. For a complete overview of more than 275 titles see the TRAIL website: www.rsTRAIL.nl. The TRAIL Thesis Series is a series of the Netherlands TRAIL Research School on transport, infrastructure and logistics.

Zeinaly, Y., *Model-based Control of Large-scale Baggage Handling Systems: Leveraging the theory of linear positive systems for robust scalable control design*, T2022/8, June 2022, TRAIL Thesis Series, the Netherlands.

Fahim, P.B.M., *The Future of Ports in the Physical Internet*, T2022/7, May 2022, TRAIL Thesis Series, the Netherlands.

Huang, B., *Assessing Reference Dependence in Travel Choice Behaviour*, T2022/6, May 2022, TRAIL Thesis Series, the Netherlands.

Reggiani, G., *A Multiscale View on Bikeability of Urban Networks*, T2022/5, May 2022, TRAIL Thesis Series, the Netherlands.

Paul, J., Online Grocery Operations in Omni-channel Retailing: opportunities and challenges, T2022/4, March 2022, TRAIL Thesis Series, the Netherlands.

Liu, M., *Cooperative Urban Driving Strategies at Signalized Intersections*, T2022/3, January 2022, TRAIL Thesis Series, the Netherlands.

Feng, Y., *Pedestrian Wayfinding and Evacuation in Virtual Reality*, T2022/2, January 2022, TRAIL Thesis Series, the Netherlands.

Scheepmaker, G.M., *Energy-efficient Train Timetabling*, T2022/1, January 2022, TRAIL Thesis Series, the Netherlands.

Bhoopalam, A., *Truck Platooning: planning and behaviour*, T2021/32, December 2021, TRAIL Thesis Series, the Netherlands.

Hartleb, J., *Public Transport and Passengers: optimization models that consider travel demand*, T2021/31, TRAIL Thesis Series, the Netherlands.

Azadeh, K., *Robotized Warehouses: design and performance analysis*, T2021/30, TRAIL Thesis Series, the Netherlands.

Chen, N., *Coordination Strategies of Connected and Automated Vehicles near Onramp Bottlenecks on Motorways*, T2021/29, December 2021, TRAIL Thesis Series, the Netherlands.

Onstein, A.T.C., *Factors influencing Physical Distribution Structure Design*, T2021/28, December 2021, TRAIL Thesis Series, the Netherlands.

Olde Kalter, M.-J. T., *Dynamics in Mode Choice Behaviour*, T2021/27, November 2021, TRAIL Thesis Series, the Netherlands.

Los, J., *Solving Large-Scale Dynamic Collaborative Vehicle Routing Problems: an Auction-Based Multi-Agent Approach*, T2021/26, November 2021, TRAIL Thesis Series, the Netherlands.

Khakdaman, M., On the Demand for Flexible and Responsive Freight Transportation Services, T2021/25, September 2021, TRAIL Thesis Series, the Netherlands.

Wierbos, M.J., *Macroscopic Characteristics of Bicycle Traffic Flow: a bird's-eye view of cycling*, T2021/24, September 2021, TRAIL Thesis Series, the Netherlands.

Qu, W., *Synchronization Control of Perturbed Passenger and Freight Operations*, T2021/23, July 2021, TRAIL Thesis Series, the Netherlands.

Nguyen, T.T., *Highway Traffic Congestion Patterns: Feature Extraction and Pattern Retrieval*, T2021/22, July 2021, TRAIL Thesis Series, the Netherlands.

Pudāne, B., *Time Use and Travel Behaviour with Automated Vehicles*, T2021/21, July 2021, TRAIL Thesis Series, the Netherlands.

Gent, P. van, *Your Car Knows Best*, T2021/20, July 2021, TRAIL Thesis Series, the Netherlands.

Wang, Y., *Modeling Human Spatial Behavior through Big Mobility Data*, T2021/19, June 2021, TRAIL Thesis Series, the Netherlands.

Coevering, P. van de, *The Interplay between Land Use, Travel Behaviour and Attitudes: a quest for causality*, T2021/18, June 2021, TRAIL Thesis Series, the Netherlands.

Landman, R., *Operational Control Solutions for Traffic Management on a Network Level*, T2021/17, June 2021, TRAIL Thesis Series, the Netherlands.

Zomer, L.-B., *Unravelling Urban Wayfinding: Studies on the development of spatial knowledge*, activity patterns, and route dynamics of cyclists, T2021/16, May 2021, TRAIL Thesis Series, the Netherlands.

Núñez Velasco, J.P., Should I Stop or Should I Cross? Interactions between vulnerable road users and automated vehicles, T2021/15, May 2021, TRAIL Thesis Series, the Netherlands.

Duivenvoorden, K., *Speed Up to Safe Interactions: The effects of intersection design and road users' behaviour on the interaction between cyclists and car drivers*, T2021/14, April 2021, TRAIL Thesis Series, the Netherlands.

Nagalur Subraveti, H.H.S., *Lane-Specific Traffic Flow Control*, T2021/13, March 2021, TRAIL Thesis Series, the Netherlands.